SPECTRAL INVARIANTS FOR MONOTONE LAGRANGIANS

RÉMI LECLERCQ, FROL ZAPOLSKY

ABSTRACT. Since spectral invariants were introduced in cotangent bundles via generating functions by Viterbo in the seminal paper [Vit92], they have been defined in various contexts, mainly via Floer homology theories, and then used in a great variety of applications. In this paper we extend their definition to monotone Lagrangians, which is so far the most general case for which a "classical" Floer theory has been developed. Then, we gather and prove the properties satisfied by these invariants, and which are crucial for their applications. Finally, as a demonstration, we apply these new invariants to symplectic rigidity of some specific monotone Lagrangians.

Contents

1.	Introduction and main results	2
1.1.	Main result	3
1.2.	Overview of the construction of spectral invariants	5
1.3.	Additional properties of spectral invariants	6
1.4.	Hofer bounds	7
1.5.	An application to symplectic rigidity	8
1.6.	Relation with previous results	10
1.7.	Discussion	11
1.8.	Overview of the paper	12
1.9.	Acknowledgements	12
2.	Background: Floer homology, quantum homology, and PSS	12
2.1.	Morse theory	13
2.2.	Floer homology	13
2.3.	Quantum homology	21
2.4.	Piunikhin–Salamon–Schwarz isomorphisms	23
2.5.	Change of coefficients, twisting, quotient complexes, Novikov	
	rings	25
2.6.	Examples	30
2.7.	Duality	36
2.8.	The Lagrangian diagonal in $(M imes M, \omega \oplus (-\omega))$	38

 $\mathit{Date:}$ October 5, 2016.

2010 Mathematics Subject Classification. Primary 57R17; Secondary 53D12 53D40.

Key words and phrases. Symplectic manifolds, monotone Lagrangian submanifolds, Lagrangian Floer homology, quantum homology, spectral invariants.

2.9. Lagrangian Floer and quantum homology of products	40
2.10. Action of the symplectomorphism group	41
3. Lagrangian spectral invariants: definition	
3.1. Nondegenerate Hamiltonians	43
3.2. Continuity property of ℓ	44
3.3. Invariance of ℓ (part 1) and arbitrary Hamiltonians	46
3.4. Invariance of ℓ (part 2)	46
4. Spectral invariants: main properties	
4.1. Quantum valuation	49
4.2. Spectral invariants of continuous Hamiltonians	52
4.3. Spectral invariants of isotopies	62
5. Proof of the Hofer bounds	64
6. Proofs concerning symplectic rigidity	69
References	

1. INTRODUCTION AND MAIN RESULTS

Spectral invariants were introduced into symplectic topology by Viterbo [Vit92], and subsequently their theory was developed by Schwarz [Sch00], Oh [Oh05a] in the context of periodic orbit Floer homology, by Oh [Oh97], Leclercq [Lec08], Monzner–Vichery–Zapolsky [MVZ12] in the context of Lagrangian Floer homology, by Chaperon [Cha95], Bhupal [Bhu01], Sandon [San11], and Zapolsky [Zap13a] for Legendrian submanifolds of contact manifolds and for contactomorphisms, and by Albers–Frauenfelder [AF10] in Rabinowitz Floer homology.

They have been used in numerous deep applications, such as metrics on infinite-dimensional groups of symmetries [Vit92, Sch00, Oh05b, Kha09, Lec08, San10, MZ11, CS15, Zap13a, Zap13b, Sey14]; the symplectic camel problem [Vit92, Thé99]; quasi-morphisms on the Hamiltonian group [EP03, Ost06, Ush11, FOOO11]; quasi-states and symplectic and contact rigidity [EP06, EP09, MVZ12, Zap13a]; orderability and contact nonsqueezing [San11, AM13]; C^0 -symplectic topology [Sey13a, Sey13b, HLS16, HLS15a, HLS15b]; function theory on symplectic manifolds [EPZ07, BEP12], [PR14] and the references therein; quantum measurements and noise [Po112, Po114]; surface dynamics [HLRS16]; and contact dynamics [Zap13a]. There are other applications, and it is not feasible to list all of them here, but the above sample should give the reader a feeling of the power of this wonderful tool of symplectic topology.

Spectral invariants have been defined in various contexts, such as periodic orbit Floer homology and Lagrangian Floer homology for weakly exact Lagrangians. The present paper extends their definition to the setting of Lagrangian Floer homology for monotone Lagrangian submanifolds. This is so far the most general case for which meaningful theory can be developed, while staying in the realm of "classical" Floer theory, that is avoiding the more

 $\mathbf{2}$

advanced and complicated techniques such as virtual fundamental cycles and Kuranishi structures, used, for instance in [FOOO09a, FOOO09b, FOO011].

Future applications of spectral invariants for monotone Lagrangians include *inter alia* new results on the Lagrangian Hofer metric [KZ].

1.1. **Main result.** Let us briefly review the setting in which we will be working. Throughout this paper we fix a closed connected symplectic manifold (M, ω) of dimension 2n and a closed connected Lagrangian submanifold $L \subset M$. We have the natural homomorphisms

$$\omega: \pi_2(M, L) \to \mathbb{R}$$
, the symplectic area, and

 $\mu: \pi_2(M, L) \to \mathbb{Z}$, the **Maslov index**.

We say that L is **monotone** if there is a positive constant τ such that

 $\omega(A) = \tau \mu(A)$ for every $A \in \pi_2(M, L)$.

The **minimal Maslov number** N_L of L is defined to be the positive generator of the subgroup $\mu(\pi_2(M, L)) \subset \mathbb{Z}$ if it is nontrivial, otherwise we set $N_L = \infty$. In this paper we assume that L is monotone of minimal Maslov number at least two. Usually we will assume that N_L is finite, which then implies that the group of periods $\omega(\pi_2(M, L)) \subset \mathbb{R}$ is infinite cyclic; in this case we let $\mathsf{A} = \tau N_L \in \mathbb{R}_{>0}$ be its positive generator. Otherwise $N_L = \infty$, which is the case of L being weakly exact.

A few remarks on the setting are in order:

- (1) For the most part, we limit ourselves to closed symplectic manifolds, however, using techniques developed by Frauenfelder–Schlenk [FS07] and Lanzat [Lan13], it is straightforward to generalize to the case of manifolds which are convex at infinity [EG91], which we will assume whenever M is noncompact.
- (2) Most results here are formulated for the case of finite N_L , and we leave to the reader the straightforward task of adapting them to the case $N_L = \infty$.
- (3) The monotonicity of L implies that M is monotone, more precisely the homomorphisms $\omega: \pi_2(M) \to \mathbb{R}$ and $c_1: \pi_2(M) \to \mathbb{Z}$ either vanish¹ or satisfy $\omega = 2\tau c_1$.

We wish to emphasize that our symplectic manifold M is monotone throughout this paper, as a result of the monotonicity assumption on L.

We proceed to formulate the main result of this paper. A few preliminaries are in order. Fix a ground ring R, which we assume to be commutative throughout. If its characteristic is different from 2, we require that the following assumption be satisfied.

Assumption (O): The second Stiefel–Whitney class $w_2(TL)$ of L vanishes on the image of the boundary homomorphism $\pi_3(M, L) \to \pi_2(L)$.

¹Then M is symplectically aspherical, for example when $\pi_2(M) = 0$ or $N_L = \infty$.

REMARK 1. Assumption (O) is satisfied if L is relatively Pin^{\pm} , that is either one of the classes $w_2(TL)$ or $w_2(TL) + w_1^2(TL)$ is in the image of the restriction morphism $H^2(M;\mathbb{Z}_2) \to H^2(L;\mathbb{Z}_2)$. However it is a strictly weaker assumption, see §1.7.

In [BC09, BC12], and [Zap15, Section 4] it is shown how to define the **quantum homology** $QH_*(L)$ of L over R. This is an algebra over R, where the product operation is

$$\star: QH_j(L) \otimes QH_k(L) \to QH_{j+k-n}(L) \,,$$

with a unit which is given by the fundamental class of L, $[L] \in QH_n(L)$. It follows that $QH_*(L) \neq 0$ if and only if $[L] \neq 0$. We assume that this is the case throughout the text.

REMARK 2. The ability to work over an arbitrary ring R, developed in [BC12], [Zap15], allows for additional applications, as exemplified in Theorem 17. Also see Section 2.6 for examples of Lagrangian submanifolds and their quantum homology with various choices of coefficients.

In §3 below we define the **Lagrangian spectral invariant** associated to L, which is a function

$$\ell: QH_*(L) \times C^0(M \times [0,1]) \to \mathbb{R} \cup \{-\infty\}.$$

We use the abbreviation

(1)
$$\ell_{+} = \ell([L]; \cdot) \,.$$

For a smooth Hamiltonian H on M we let $\text{Spec}(H : L) \subset \mathbb{R}$ be its action spectrum relative to L, that is the set of critical values of the associated action functional, see the definitions in §2.

Given a continuous function $H: M \times [0,1] \to \mathbb{R}$, we denote by H the function defined by $\overline{H}_t(x) = -H_{1-t}(x)$. In case H is smooth, we have $\phi_{\overline{H}}^t = \phi_H^{1-t}\phi_H^{-1}$ for all t. If $H, K: M \times [0,1] \to \mathbb{R}$ are continuous and $H_t = K_t = 0$ for t close to 0, 1, we define $H \sharp K: M \times [0,2] \to \mathbb{R}$ to be their concatenation with respect to the time variable.²

Our first result concerns the main properties of the spectral invariant.

Theorem 3. Let L be a closed connected monotone Lagrangian submanifold of (M, ω) with minimal Maslov number $N_L \geq 2$. The spectral invariant

$$\ell: QH_*(L) \times C^0(M \times [0,1]) \to \mathbb{R} \cup \{-\infty\}$$

satisfies the following properties.

FINITENESS: $\ell(\alpha; H) = -\infty$ if and only if $\alpha = 0$.

SPECTRALITY: For $H \in C^{\infty}(M \times [0,1])$ and $\alpha \neq 0$, $\ell(\alpha; H) \in \text{Spec}(H : L)$. GROUND RING ACTION: For $r \in R$, $\ell(r \cdot \alpha; H) \leq \ell(\alpha; H)$. In particular, if r is invertible, then $\ell(r \cdot \alpha; H) = \ell(\alpha; H)$.

²The definition of spectral invariants can be generalized in a straightforward manner to Hamiltonians parametrized by a time interval which is not necessarily [0, 1].

NORMALIZATION: If c is a function of time then

$$\ell(\alpha; H+c) = \ell(\alpha; H) + \int_0^1 c(t) dt \,.$$

We have $\ell_{+}(0) = 0$.

CONTINUITY: For any Hamiltonians H, K, and $\alpha \neq 0$:

$$\int_{0}^{1} \min_{M} (K_{t} - H_{t}) \, dt \le \ell(\alpha; K) - \ell(\alpha; H) \le \int_{0}^{1} \max_{M} (K_{t} - H_{t}) \, dt$$

MONOTONICITY: If $H \leq K$, then $\ell(\alpha; H) \leq \ell(\alpha; K)$.

TRIANGLE INEQUALITY: For all α and β :

$$\ell(\alpha \star \beta; H \sharp K) \le \ell(\beta; H) + \ell(\alpha; K) \,.$$

LAGRANGIAN CONTROL: If for all t, $H_t|_L = c(t) \in \mathbb{R}$ (respectively \leq, \geq), then

$$\ell_+(H) = \int_0^1 c(t) \, dt \quad (respectively \leq \geq).$$

Thus for all H:

$$\int_0^1 \min_L H_t \, dt \le \ell_+(H) \le \int_0^1 \max_L H_t \, dt \, .$$

Non-negativity: $\ell_+(H) + \ell_+(\overline{H}) \ge 0.$

Theorem 3 is proved in §4 below as part of the more general Theorem 36.

We say that H is **normalized** if $\int_M H_t \omega^n = 0$ for all t. We have the following, by now standard, observation.

Proposition 4. If H is normalized, then for $\alpha \in QH_*(L) \setminus \{0\}$ the spectral invariant $\ell(\alpha; H)$ only depends on the homotopy class of the path $\{\phi_H^t\}_{t \in [0,1]}$ relative to the endpoints.

See §3.4 for a proof. Thus the spectral invariant descends to a function

$$\ell: QH_*(L) \times \widetilde{\operatorname{Ham}}(M, \omega) \to \mathbb{R} \cup \{-\infty\}.$$

This function satisfies a number of properties, analogous to those formulated in Theorem 3. See §4.3 for details.

1.2. Overview of the construction of spectral invariants. We present here a brief sketch of the construction of ℓ . The reader is referred to §3 for full details.

Assume that $H: M \times [0,1] \to \mathbb{R}$ is such that $\phi^1_H(L)$ intersects L transversely. It is then possible to choose a time-dependent family of ω -compatible almost complex structures J on M for which the Floer complex

$$(CF_*(H:L),\partial_{H,J})$$

is defined. This complex is filtered by the action functional of H, and we let

$$CF^a_*(H:L) \subset CF_*(H:L)$$

be the submodule generated by elements whose action is $\langle a$. From the definition of the Floer boundary operator $\partial_{H,J}$ it follows that this submodule is in fact a subcomplex. We let

$$HF_*(H, J:L)$$
 and $HF^a_*(H, J:L)$

be the homologies of $(CF_*(H:L), \partial_{H,J})$ and $(CF^a_*(H:L), \partial_{H,J})$, respectively, and we let

$$i^a_* \colon HF^a_*(H,J:L) \to HF_*(H,J:L)$$

be the morphism induced on homology by the inclusion.

There is a canonical map, the so-called PSS isomorphism

$$PSS^{H,J}: QH_*(L) \to HF_*(H,J:L).$$

For $\alpha \in QH_*(L)$ we then define

$$\ell(\alpha; H) = \inf\{a \in \mathbb{R} \mid \mathrm{PSS}^{H, J}(\alpha) \in \mathrm{im}\, i^a_*\}.$$

It is then shown that ℓ is continuous in H in the sense of Theorem 3, which implies that is possesses a unique extension to the set of continuous Hamiltonians.

1.3. Additional properties of spectral invariants. Theorem 3 collects the more basic properties of the Lagrangian spectral invariants. The latter satisfy a number of other properties, of which we mention three in this introduction. For the full list the reader is referred to Theorem 36 in §4.

The quantum homology of M acts on $QH_*(L)$ via the so-called quantum module action [BC09], [Zap15, Section 4.5]

•:
$$QH_j(M) \otimes QH_k(L) \to QH_{j+k-2n}(L)$$
.

We have the corresponding MODULE STRUCTURE property, where

$$c: QH_*(M) \times C^0(M \times S^1) \to \mathbb{R}$$

is the Hamiltonian spectral invariant [Oh05a], see also §3.1:

Proposition 5. Let H^1 , H^2 be Hamiltonians, with H^2 being time-periodic. Then for $a \in QH_*(M)$ and $\alpha \in QH_*(L)$ we have

$$\ell(a \bullet \alpha; H^1 \sharp H^2) \le c(a; H^2) + \ell(\alpha; H^1).$$

This is proved as part of the properties of spectral invariants, Theorem 36.

Next we show that Lagrangian spectral invariants generalize Hamiltonian spectral invariants. Let $\Delta \subset M \times M$ be the Lagrangian diagonal where the symplectic structure is $\omega \oplus (-\omega)$. In the following theorem $QH_*(\Delta)$ denotes the Lagrangian quantum homology of Δ .

Theorem 6. There is a canonical algebra isomorphism $QH_*(\Delta) = QH_*(M)$. For any class $\alpha \in QH_*(\Delta) = QH_*(M)$ and any time-periodic Hamiltonian H on M we have

$$c(\alpha; H) = \ell(\alpha; H \oplus 0).$$

See 4.2.2 for a proof.

The last property we wish to mention is a relation between the spectral invariants of two monotone Lagrangians $L_i \subset (M_i, \omega_i)$ such that the product $L_1 \times L_2$ is monotone of minimal Maslov at least two. In this case both L_1 , L_2 have minimal Maslov number at least 2. In §4.2.3 below we prove the following property:

$$\ell(\alpha_1 \otimes \alpha_2; H^1 \oplus H^2) \le \ell(\alpha_1; H^1) + \ell(\alpha_2; H^2)$$

where $\alpha_i \in QH_*(L_i)$, H^i is a Hamiltonian on M_i , and $\alpha_1 \otimes \alpha_2 \in QH_*(L_1 \times L_2)$ denotes the element canonically associated to α_1 , α_2 . In case the ground ring R is a field, we have an equality.

1.4. Hofer bounds. We now explain a relation between Lagrangian spectral invariants associated to two Lagrangian submanifolds $L, L' = \varphi(L)$ for some $\varphi \in \text{Ham}(M, \omega)$, and the Hofer norm of φ , as well as the Lagrangian Hofer distance between L and L'.

First, as we shall see in Theorem 36, Lagrangian spectral invariants satisfy a SYMPLECTIC INVARIANCE property, expressed as follows. Let $\psi \in$ Symp (M, ω) , and let $L' = \psi(L)$. To ψ there is associated an isomorphism

$$\psi_* \colon QH_*(L) \to QH_*(L')$$
,

see Section 2.10. Denote by ℓ' the spectral invariant associated to L'. Then for any $\alpha' \in QH_*(L')$ and any Hamiltonian H we have

$$\ell'(\alpha'; H) = \ell(\psi_*^{-1}(\alpha'); H \circ \psi).$$

For a fixed $\alpha' \in QH_*(L')$, this property allows us to compare $\ell'(\alpha'; \cdot)$ and $\ell(\psi_*^{-1}(\alpha'); \cdot)$ as functions defined on $C^0(M \times [0, 1])$ or on $\widetilde{\text{Ham}}(M, \omega)$.

The following proposition states that when ψ is Hamiltonian, the difference between these two functions is bounded from above by the **Hofer norm** of ψ , defined as

$$\|\psi\| = \inf\left\{\int_0^1 \operatorname{osc}_M(H_t) \, dt \, | \, H \colon \, M \times [0,1] \to \mathbb{R} \text{ with } \phi_H = \psi\right\}$$

Proposition 7. Let L and L' be Hamiltonian isotopic Lagrangians. Let $\alpha \in QH_*(L)$. For all $\varphi \in \text{Ham}(M, \omega)$ such that $\varphi(L) = L'$ we have

$$|\ell(\alpha; H) - \ell'(\varphi_*(\alpha); H)| \le ||\varphi|| \quad and \quad |\ell(\alpha; \widetilde{\chi}) - \ell'(\varphi_*(\alpha); \widetilde{\chi})| \le ||\varphi||,$$

where $H \in C^0(M \times [0,1])$ and $\widetilde{\chi} \in \widetilde{\text{Ham}}(M,\omega)$ are arbitrary elements.

This property is somewhat surprising since, in the monotone case, one might have expected a bound in terms of Hofer's geometry of the universal cover of the Hamiltonian diffeomorphism group and not of the group itself (compare to the weakly exact case [Lec08] and [MVZ12] where the Lagrangian spectral invariants only depend on time-1 objects).

In a similar vein we formulate a relation with the Lagrangian Hofer distance (see [Che00]). Here we use a variant of this distance defined on the universal cover of the space of Lagrangians which are Hamiltonian isotopic to a given one L. It is defined as follows. Recall that given a Hamiltonian H, the associated path of Lagrangians $L_t = \phi_H^t(L)$ is exact, meaning the tangent vector to this path at every point L_t is given by an exact 1-form on L_t . Conversely, any exact path of Lagrangians starting at L can be represented in such a way. Given a homotopy class \tilde{L} of exact paths of Lagrangians from L to L' we define the **Hofer length** of \tilde{L} to be

 $\|\widetilde{L}\| = \inf \left\{ \int_0^1 \operatorname{osc}_M(H_t) dt \,|\, [\{\phi_H^t(L)\}_{t \in [0,1]}] = \widetilde{L} \right\}.$

Such a homotopy class induces a natural isomorphism $QH_*(L) \simeq QH_*(L')$, as follows. Assume $L_t = \phi_H^t(L)$ is an exact path of Lagrangians from L to L'. The symplectomorphism ϕ_H induces an isomorphism $(\phi_H)_*: QH_*(L) \to QH_*(L')$, see §2.10. We have

Theorem 8. The induced isomorphism $QH_*(L) \to QH_*(L')$ only depends on the homotopy class of the path $\{L_t\}_{t \in [0,1]}$ relative to the endpoints.

REMARK 9. It follows that Lagrangian quantum homology forms a local system over the space of Lagrangians which are Hamiltonian isotopic to L. We then have

Proposition 10. Let \widetilde{L} be a homotopy class of exact paths of Lagrangians connecting L and L', and let $\alpha \in QH_*(L)$, $\alpha' \in QH_*(L')$ be quantum homology classes corresponding to each other by the above natural isomorphism induced by \widetilde{L} . Then

 $|\ell(\alpha; H) - \ell'(\alpha'; H)| \le \|\widetilde{L}\| \quad and \quad |\ell(\alpha; \widetilde{\chi}) - \ell'(\alpha'; \widetilde{\chi})| \le \|\widetilde{L}\|$

for any $H \in C^0(M \times [0,1])$ and $\widetilde{\chi} \in \widetilde{\operatorname{Ham}}(M,\omega)$.

Propositions 7, 10, and Theorem 8 are proved in §5.

REMARK 11. Let \mathcal{L} be the universal cover of the space of Lagrangians Hamiltonian isotopic to L, that is an element of $\widetilde{\mathcal{L}}$ is a homotopy class of exact paths of Lagrangians starting at L, relative to the endpoints. Over $\widetilde{\mathcal{L}}$ we have the natural local system of algebras $\mathcal{QH} \to \widetilde{\mathcal{L}}$ whose fiber at a homotopy class \widetilde{L} with endoint L' is given by $\mathcal{QH}_*(L')$. The above isomorphisms provide a canonical trivialization of this local system, that is $\mathcal{QH} = \mathcal{QH}_*(L) \times \widetilde{\mathcal{L}}$. The above discussion allows us, given $\alpha \in \mathcal{QH}_*(L)$, to view the associated spectral invariant as a function on the space

$$\widetilde{\mathcal{L}} \times \widetilde{\operatorname{Ham}}(M, \omega)$$
.

Proposition 10, together with the CONTINUITY property, implies that every such function is Lipschitz with respect to the natural Hofer (pseudo-)metric on this space.

1.5. An application to symplectic rigidity. We now present a sample application of this theory to symplectic rigidity. If $e \in QH_*(M)$ is any nonzero idempotent, Entov–Polterovich [EP09] defined two classes of rigid subsets of M, namely heavy and superheavy subsets with respect to e:

Definition 12. A closed subset $X \subset M$ is called *e*-heavy if for every $F \in C^{\infty}(M)$ we have

$$\lim_{k \to \infty} \frac{c(e; kF)}{k} \ge \min_X F \,,$$

and it is called *e*-superheavy if for every $F \in C^{\infty}(M)$

$$\lim_{k \to \infty} \frac{c(e; kF)}{k} \le \max_X F$$

In [EP06] they also introduced the notion of a symplectic quasi-state. In the following definition $\{\cdot, \cdot\}$ stands for the Poisson bracket and $\|\cdot\|_{C^0}$ for the supremum norm.

Definition 13. A quasi-state on M is a functional $\zeta: C^0(M) \to \mathbb{R}$ satisfying

NORMALIZATION: $\zeta(1) = 1$.

QUASI-LINEARITY: For $F, G \in C^{\infty}(M)$ with $\{F, G\} = 0$ we have $\zeta(F+G) = \zeta(F) + \zeta(G)$.

MONOTONICITY: For $F, G \in C^0(M)$ with $F \leq G$ we have $\zeta(F) \leq \zeta(G)$.

They developed a construction of (nonlinear) symplectic quasi-states on M using idempotents in $QH_*(M)$. We refer the reader to [EP08] for details. Briefly, if $QH_*(M) \simeq \mathcal{F} \oplus \mathcal{Q}$ as an algebra where \mathcal{F} is a field, then the spectral invariant $c(e; \cdot)$, where $e \in \mathcal{F}$ is the unit, has the property that the functional

$$F \mapsto \zeta_e(F) = \lim_{k \to \infty} \frac{c(e; kF)}{k}$$

is a symplectic quasi-state.

REMARK 14. If *e* gives rise to a symplectic quasi-state as above, the notions of *e*-heavy and *e*-superheavy coincide [EP09].

In this paper we use Lagrangian spectral invariants to prove the superheaviness of certain Lagrangian submanifolds. Namely, we have the following result, proved in §6 below:

Proposition 15. Let $e \in QH_*(M)$ be an idempotent. If $e \bullet [L] \neq 0$, then L is e-heavy. Furthermore, if e gives rise to a symplectic quasi-state as above, thanks to Remark 14, L is e-superheavy.

REMARK 16. This proposition is analogous to, and in certain cases follows from, Proposition 8.1 of [EP09].³ This is essentially because the quantum inclusion i_L of Biran–Cornea [BC09], appearing in the formulation of the cited proposition in the guise of the map j^q , is dual to the quantum action $QH(M) \ni e \mapsto e \bullet [L] \in QH(L)$.

We apply this proposition to two examples of monotone Lagrangians. First we consider the monotone product $S^2 \times S^2$ in which there is a monotone

³We thank Leonid Polterovich for pointing this out to us.

Lagrangian torus $L_{S^2 \times S^2}$ defined as follows. View $S^2 \subset \mathbb{R}^3$ as the set of unit vectors. Then

$$L_{S^2 \times S^2} = \left\{ (x,y) \in S^2 \times S^2 \, | \, x \cdot y = -\frac{1}{2} \, , \, \, x_3 + y_3 = 0 \right\},$$

where $x \cdot y$ is the Euclidean scalar product. See [EP10] and references therein.

The other example is the Chekanov monotone torus $L_{\mathbb{C}P^2} \subset \mathbb{C}P^2$ [CS10]. We refer the reader to the paper by Oakley–Usher [OU16], in which they review various constructions of monotone tori in $S^2 \times S^2$ and in $\mathbb{C}P^2$, and prove that they are all Hamiltonian equivalent. Also see the paper by Gadbled [Gad13]. One possible description of $L_{\mathbb{C}P^2}$ is as follows. Consider the degree 2 polarization of $\mathbb{C}P^2$ by a conic. This conic is a complex (and thus symplectic) sphere, in which there is a monotone Lagrangian circle, the equator. The Lagrangian circle bundle construction [BC01, Bir06] in this situation yields $L_{\mathbb{C}P^2}$.

Theorem 17. • The Chekanov torus $L_{\mathbb{C}P^2}$ is superheavy with respect to the fundamental class $[\mathbb{C}P^2]$ taken over the ground ring $R = \mathbb{C}$.

• Over the ground ring $R = \mathbb{C}$, the quantum homology $QH_*(S^2 \times S^2)$ contains two idempotents e_{\pm} . The torus $L_{S^2 \times S^2}$ is e_{\pm} -superheavy.

We prove this in $\S6$.

1.6. Relation with previous results. The properties of the Lagrangian spectral invariants listed in Theorem 3 have previously appeared in other contexts, and consequently they are more or less standard, see [Vit92, Oh05a, MVZ12]. The major exception here is the MODULE PROPERTY, see Proposition 5. An early version of this result was establised in [MVZ12] for the zero section of a cotangent bundle, in case a = [M] and $\alpha = [L]$. An inequality identical to ours in the setting of weakly exact Lagrangians was proved in [DKM14]. TRIANGLE INEQUALITY for Floer-homological spectral invariants was first proved by Oh [Oh99], although the inequality proved there included an error term. In the context of Hamiltonian Floer homology the sharp inequality was proved by Schwarz [Sch00]. A sharp triangle inequality for Lagrangian spectral invariants was proved in [MVZ12] based on the trick due to Abbondandolo–Schwarz [AS10].

Theorem 6 was proved by Leclercq [Lec08] for weakly exact M. The canonical isomorphism $QH_*(\Delta) = QH_*(M)$ appears, for example, in [BPS03].

Various versions of Proposition 15 have appeared in the literature, see for example [EP09], as well as [BC09]; these results use early ideas appearing in [Alb05]; see also [FOOO11]. Let us point out that the estimates on spectral invariants in these papers use quantities such as depth and height of a Hamiltonian relative to L, which in our opinion are somewhat artificial, and they do not have a clear "symplectic" meaning. In contrast, Lagrangian spectral invariants do possess such a meaning, and therefore the comparison of Hamiltonian and Lagrangian spectral invariants, used in the proof of Proposition 15 seems to be a more natural approach.

The superheaviness of the Chekanov torus $L_{\mathbb{C}P^2} \subset \mathbb{C}P^2$ was proved by Wu [Wu12]. The proof there relies on the fact that $HF_*(L_{\mathbb{C}P^2}) \neq 0$, which in turn uses the machinery of toric degenerations, while our proof does not rely on it. The superheaviness of the torus $L_{S^2 \times S^2}$ was established by Eliashberg–Polterovich [EP10]. Again, our proof is based on a more natural approach to estimating Hamiltonian spectral invariants via Lagrangian ones.

1.7. **Discussion.** We use quantum homology over arbitrary rings, therefore the issue of orientations plays a crucial role. This was resolved in the context of periodic orbit Floer homology in [FH93], which gave rise to the notion of coherent orientations, which has been used ever since. Coherent orientations for Lagrangian Floer homology are constructed in [FOOO09b], and for Lagrangian quantum homology in [BC12]; see also [HL10, WW15]. We use here the notion of canonical orientations, which also appears in [Sei08, Wel08, Abo15]. This has the advantage that the structures inherent to the theory are more transparent. The drawback, of course, is a higher level of abstraction, and that computations are less straightforward.

Assumption (O) is the minimal assumption under which the so-called canonical quantum homology can be defined [Zap15]. The corresponding chain complex distinguishes homotopy classes of disks, rather than their homology classes or symplectic area, which are the more usual equivalence relations appearing in Floer theories. The advantage of the canonical complex is that there are no further assumptions or choices required for its definition, for instance L is not required to be orientable or relatively (S)Pin.

The drawback of the canonical complex is that the resulting homology is oftentimes too small and does not contain the desired classes, for example, it may happen that not all singular homology classes of L appear in it. To remedy this, there is the possibility of forming quotient complexes, where cappings having the same area or the same homology class are identified, in which case the homology of the complex is better behaved. In order to form such quotient complexes, further assumptions and choices need to be made. Even though precise assumptions can be formulated in terms of various obstruction classes, we point out that any sort of quotient complex can be formed once L is assumed to be relatively Pin^{\pm} and a relative Pin^{\pm} structure has been chosen.

Definition 18. We say that L is **relatively** Pin^{\pm} if the class $w^{\pm}(L)$ is in the image of the restriction morphism $H^2(M; \mathbb{Z}_2) \to H^2(L; \mathbb{Z}_2)$, where $w^+(L) = w_2(TL)$ and $w^-(L) = w_2(TL) + w_1^2(TL)$.

In [Zap15, Section 7] the notion of a relative Pin^{\pm} structure is defined, following the category-theoretic approach appearing in [WW15], and it is shown how to produce coherent orientations for disks in M with boundary on Land how to use them to define the aforementioned quotient complexes. **Example 19.** Here we point out that, as we mentioned above, assumption **(O)** is strictly weaker than L being relatively Pin^{\pm} . An example is furnished by the Lagrangian $\mathbb{R}P^5 \subset \mathbb{C}P^5$. Indeed, assumption **(O)** is satisfied since $\pi_2(\mathbb{R}P^5) = 0$. On the other hand $\mathbb{R}P^5$ is not relatively Pin^{\pm} since the restriction map $H^2(\mathbb{C}P^5;\mathbb{Z}_2) \to H^2(\mathbb{R}P^5;\mathbb{Z}_2)$ vanishes while the classes $w_2(\mathbb{R}P^5) = w_2(\mathbb{R}P^5) + w_1^2(\mathbb{R}P^5)$ do not.⁴

1.8. Overview of the paper. In §2 we review the construction of Lagrangian and symplectic Floer and quantum homology with canonical orientations, as well as the relevant Piunikhin–Salamon–Schwarz isomorphisms. Various algebraic structures and operations appearing in these theories are also described, and examples of quantum complexes for various Lagrangians with different coefficients are given. In §3 we define the Lagrangian spectral invariants and prove basic properties thereof. In §4 we prove the main properties of spectral invariants. In §5 we prove the Hofer bounds on spectral invariants formulated in §1.4. Finally §6 contains the proofs regarding superheavy Lagrangians.

1.9. Acknowledgements. RL is partially supported by ANR Grant ANR-13-JS01-0008-01. FZ is partially supported by grant number 1281 from the GIF, the German–Israeli Foundation for Scientific Research and Development, by grant number 1825/14 from the Israel Science Foundation, and wishes to thank RL and Université Paris–Sud for the opportunity to visit the maths department as a professeur invité, during which substantial portions of the present paper, as well as of [Zap15], were written; the research and personal atmosphere there were excellent, and the cafeteria was nice. We are grateful to Leonid Polterovich whose suggestion lead us to Propositions 7 and 10. Finally, we thank the anonymous referee for a thorough and detailed report on the previous version of this paper.

2. Background: Floer homology, quantum homology, and PSS

Here we collect the necessary preliminaries concerning the Floer theories that we need in order to define and study Lagrangian spectral invariants, and establish notation. Floer homology was defined by Floer [Flo89a, Flo88a, Flo88b, Flo89b], and extended by Hofer–Salamon [HS95]. The Lagrangian case was handled, for instance, by Floer (*loc. cit.*), Oh [Oh93], Biran–Cornea [BC09, BC12], Seidel [Sei08], Fukaya–Ohta–Oh–Ono [FOOO09a, FOOO09b].

The principal reference for all the material in this section is [Zap15], where the reader will find a detailed description of all the concepts not defined here; the style of [Sei08] will undoubtedly be felt here, and indeed this book served as inspiration for many of the constructions presented herein.

⁴We wish to thank Jean-Yves Welschinger for this example.

2.1. Morse theory. We briefly recall the basic notions of Morse theory, mostly in order to establish notation. For a Morse function f on a manifold Q we denote by Crit f its set of critical points. The index of $q \in$ Crit f is denoted by $\inf_f q$, or just $\inf_q q$ if f is clear from the context. For a Riemannian metric ρ on Q we let $S_f(q)$, $\mathcal{U}_f(q)$ be the stable and the unstable manifolds of f at q with respect to the negative gradient flow of f relative to ρ ; mostly we omit the subscript f. We say that the pair (f, ρ) is **Morse–Smale** if every unstable manifold intersects every stable manifold transversely.

2.2. Floer homology. This section is concerned with Lagrangian and periodic orbit Floer homology.

2.2.1. Lagrangian Floer homology. We let

$$\Omega_L = \{ \gamma \colon [0,1] \to M \, | \, \gamma(0), \gamma(1) \in L \, , [\gamma] = 0 \in \pi_1(M,L) \} \, .$$

Let $D^2 \subset \mathbb{C}$ be the closed unit disk with the standard conformal structure. We consider the punctured disk $\dot{D}^2 = D^2 \setminus \{1\}$ with the induced conformal structure and endow it with the standard positive end ϵ : $[0, \infty) \times [0, 1] \rightarrow \dot{D}^2$ defined by

(2)
$$\epsilon(z) = \frac{e^{\pi z} - i}{e^{\pi z} + i}$$

A capping of an arc $\gamma \in \Omega_L$ is a smooth map $\widehat{\gamma}: (\dot{D}^2, \partial \dot{D}^2) \to (M, L)$ which is b-smooth⁵ in the sense of [Zap15, Section 3.3], and satisfies

$$\widehat{\gamma}(\epsilon(s,t)) \xrightarrow[s \to \infty]{} \gamma(t).$$

This condition in particular means that if we compactify \dot{D}^2 by adding an interval at infinity according to the end ϵ to obtain a surface with corners, then $\hat{\gamma}$ extends to a smooth map defined on this surface. Sometimes we will also use this extension, and by abuse of notation we will denote it by $\hat{\gamma}$ as well.

We let $\dot{D}_{-}^2 = D^2 \setminus \{-1\}$ with the induced conformal structure. For a capping $\hat{\gamma}$ of γ we define its opposite as the map

$$-\widehat{\gamma}: \dot{D}_{-}^{2} \to M \quad \text{given by} \quad -\widehat{\gamma}(\sigma, \tau) = \gamma(-\sigma, \tau) \,.$$

We call two cappings $\hat{\gamma}, \hat{\gamma}'$ equivalent, and denote $\hat{\gamma} \sim \hat{\gamma}'$, if the preglued map $\hat{\gamma} \sharp (-\hat{\gamma}')$ defines the trivial element in $\pi_2(M, L)$ (see [Zap15, Section 3.4] for details on pregluing). Two pairs $(\gamma, \hat{\gamma}), (\gamma', \hat{\gamma}')$ are equivalent if $\gamma = \gamma'$ and $\hat{\gamma}, \hat{\gamma}'$ are equivalent cappings. We let the class of $(\gamma, \hat{\gamma})$ be denoted by $[\gamma, \hat{\gamma}]$ and define

$$\hat{\Omega}_{L} = \left\{ \left[\gamma, \hat{\gamma} \right] \, \middle| \, \gamma \in \Omega_{L} \,, \hat{\gamma} \text{ is a capping of } \gamma \right\},\,$$

⁵This notion is borrowed from Schwarz's thesis [Sch] and his book on Morse homology [Sch93], where it appears under a different name. Roughly speaking, being b-smooth means that $\hat{\gamma} \circ \epsilon$ converges to γ sufficiently fast together with all derivatives as $s \to \infty$.

which together with the obvious projection $p: \Omega_L \to \Omega_L$ forms a covering space of Ω_L . We usually denote elements of $\widetilde{\Omega}_L$ by $\widetilde{\gamma}$ and it is understood that it is the class $[\gamma, \widehat{\gamma}]$.

For a time-dependent Hamiltonian H on M we have the action functional

$$\mathcal{A}_{H:L}: \widetilde{\Omega}_L \to \mathbb{R}, \quad \mathcal{A}_{H:L}(\widetilde{\gamma} = [\gamma, \widehat{\gamma}]) = \int_0^1 H_t(\gamma(t)) dt - \int_{\dot{D}^2} \widehat{\gamma}^* \omega.$$

Its critical point set Crit $\mathcal{A}_{H:L}$ consists of the classes $[\gamma, \hat{\gamma}]$ for which γ is a Hamiltonian arc of H, that is solves the ODE $\dot{\gamma} = X_H(\gamma)$.

We call the Hamiltonian H nondegenerate if for every critical point $\tilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}$ the linearized map $\phi_{H,*}: T_{\gamma(0)}M \to T_{\gamma(1)}M$ maps $T_{\gamma(0)}L$ to a subspace transverse to $T_{\gamma(1)}L$. For such a Hamiltonian there is a well-defined function

$$m_{H:L}$$
: Crit $\mathcal{A}_{H:L} \to \mathbb{Z}$

called the Conley–Zehnder index.⁶

Fix now a time-dependent ω -compatible almost complex structure J on M. In [Zap15, Section 3.4] it is explained how to construct, for p > 2, a Fredholm operator

$$D_{\widehat{\gamma}} \colon W^{1,p} \left(\dot{D}^2, \partial \dot{D}^2; \widehat{\gamma}^* TM, (\widehat{\gamma}|_{\partial \dot{D}^2})^* TL \right) \to L^p (\dot{D}^2; \Omega^{0,1} \otimes \widehat{\gamma}^* TM)$$

given a capping $\widehat{\gamma}$. This operator has index ind $D_{\widehat{\gamma}} = n - m_{H:L}(\widetilde{\gamma})$. *Ibid.*, it is shown how to assemble these into a Fredholm bundle $D_{\widetilde{\gamma}}$ over the space $\widetilde{\gamma}$ of cappings⁷ in a given equivalence class.⁸

Recall [Zin13] that the determinant line of a Fredholm operator D is defined to be

$$d(D) = d(\ker D) \otimes d(\operatorname{coker} D)^{\vee}$$

where for a finite-dimensional real vector space V we denote by d(V) its top exterior power.

The determinant line bundle $d(D_{\tilde{\gamma}})$ is well-defined and we have the following foundational lemma, proved in [Zap15, Section 3.7].

Lemma 20. The line bundle $d(D_{\tilde{\gamma}})$ is orientable if and only assumption (O) holds.

This allows us to define the rank 1 free \mathbb{Z} -module $C(\tilde{\gamma})$ whose two generators are the two possible orientations⁹ of the line bundle $d(D_{\tilde{\gamma}})$. The Lagrangian

⁶We normalize it so that if H is the pullback of a C^2 -small Morse function $f: L \to \mathbb{R}$ to a Weinstein neighborhood of L, then for $q \in \operatorname{Crit} f$ and \hat{q} being the constant capping we have $m_{H:L}([q, \hat{q}]) = \operatorname{ind}_f q$, and in general $m_{H:L}([\gamma, A \sharp \hat{\gamma}]) = m_{H:L}([\gamma, \hat{\gamma}]) - \mu(A)$ for $A \in \pi_2(M, L, \gamma(0)).$

⁷The set $\tilde{\gamma} = \{(\gamma, \hat{\gamma}) | \hat{\gamma} \text{ is a capping of } \gamma\}$ is of course a topological space.

⁸Strictly speaking, the family $D_{\tilde{\gamma}}$ is parametrized by a space which forms a fibration over $\tilde{\gamma}$ with contractible fibers corresponding to auxiliary choices such as connections and extensions of almost complex structures, but such details are irrelevant here.

⁹Note that $\tilde{\gamma}$ is a connected topological space.

Floer complex as a \mathbb{Z} -module is

$$CF_*(H:L) = \bigoplus_{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}} C(\widetilde{\gamma})$$

It is graded by $m_{H:L}$ and its generators are assigned actions using $\mathcal{A}_{H:L}$. For $\tilde{\gamma}_{\pm} \in \operatorname{Crit} \mathcal{A}_{H:L}$ we let

$$\widetilde{\mathcal{M}}(H,J;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+}) = \left\{ u: \mathbb{R} \times [0,1] \to M \,|\, \overline{\partial}_{H,J}u = 0 \,, u\big(\mathbb{R} \times \{0,1\}\big) \subset L \,, \\ u(\pm\infty,\cdot) = \gamma_{\pm} \,, \widehat{\gamma}_{-} \sharp u \sim \widehat{\gamma}_{+} \right\},$$

where

$$\overline{\partial}_{H,J}u = \partial_s u + J_t(u) \big(\partial_t u - X_H(u)\big) \,.$$

REMARK 21. It is well-known that the Floer equation $\overline{\partial}_{H,J}u = 0$ can be considered as the negative gradient flow equation for the action functional. To see this, define a scalar product on

$$T_{\widetilde{\gamma}}\widetilde{\Omega}_L = C^{\infty}\big([0,1],\{0,1\};\gamma^*TM,(\gamma|_{\{0,1\}})^*TL\big)$$

via

$$\langle \xi, \eta \rangle = \int_0^1 \omega \big(\xi(t), J_t \eta(t) \big) dt$$

Then the gradient of $\mathcal{A}_{H:L}$ at $\widetilde{\gamma}$ writes

$$\nabla_{\widetilde{\gamma}} \mathcal{A}_{H:L} = J(\gamma) \left(\dot{\gamma} - X_H(\gamma) \right).$$

For a fixed nondegenerate H there is a residual subset of almost complex structures J for which the linearized operator D_u is surjective for any element $u \in \widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_-, \widetilde{\gamma}_+)$. We call the pair (H, J) a **regular Floer datum** for Lif this is the case. Then it follows that $\widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_-, \widetilde{\gamma}_+)$ is a smooth manifold of dimension $m_{H:L}(\widetilde{\gamma}_-) - m_{H:L}(\widetilde{\gamma}_+)$.

The group \mathbb{R} acts on $\widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+})$ by shifts in the \mathbb{R} variable; we let $\mathcal{M}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+})$ be the quotient. This is a smooth manifold of dimension $m_{H:L}(\widetilde{\gamma}_{-}) - m_{H:L}(\widetilde{\gamma}_{+}) - 1$. If this dimension is 0, then it is a compact manifold, meaning it is a finite set of points.

When $m_{H:L}(\tilde{\gamma}_{-}) = m_{H:L}(\tilde{\gamma}_{+}) + 1$ and $u \in \mathcal{M}(H, J; \tilde{\gamma}_{-}, \tilde{\gamma}_{+})$, the operator D_u has index 1, is surjective, and its kernel is canonically isomorphic to \mathbb{R} via the \mathbb{R} -action by translations. In particular, since $d(D_u) = d(\ker D_u) \otimes d(\operatorname{coker} D_u)^{\vee} = d(\mathbb{R})$, we see that D_u has a canonical orientation, corresponding to the positive orientation of \mathbb{R} ; we denote it by $\partial_u \in d(D_u)$.

We define the boundary operator

$$\partial_{H,J}: CF_k(H:L) \to CF_{k-1}(H:L)$$

as the unique Z-linear map whose matrix element $C(\tilde{\gamma}_{-}) \to C(\tilde{\gamma}_{+})$ is the map

$$\sum_{[u]\in\mathcal{M}(H,J;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+})}C(u)$$

where $C(u): C(\tilde{\gamma}_{-}) \to C(\tilde{\gamma}_{+})$ is an isomorphism of \mathbb{Z} -modules, defined as follows. Recall that $C(\tilde{\gamma}_{\pm})$ are generated by orientations of the line bundles $d(D_{\tilde{\gamma}_{\pm}})$. Consider the following string of isomorphisms

$$\mathrm{d}(D_{\widehat{\gamma}_{-}}) \simeq \mathrm{d}(D_{u}) \otimes \mathrm{d}(D_{\widehat{\gamma}_{-}}) \simeq \mathrm{d}(D_{\widehat{\gamma}_{-}} \oplus D_{u}) \simeq \mathrm{d}(D_{\widehat{\gamma}_{+}}),$$

where the first isomorphism maps $\mathbf{o}_{-} \mapsto \partial_{u} \otimes \mathbf{o}_{-}$, the second one is the direct sum isomorphism (defined in [Zap15, Section 2]), while the third one is the composition of linear gluing (defined in *loc. cit.*, Section 3.2) and deformation isomorphisms (see *loc. cit.*, Section 3.4). The resulting isomorphism of lines $d(D_{\widehat{\gamma}_{-}}) \simeq d(D_{\widehat{\gamma}_{+}})$ gives rise to the desired isomorphism of \mathbb{Z} -modules $C(\widetilde{\gamma}_{-}) \simeq C(\widetilde{\gamma}_{+})$, which is C(u). It can be seen that this isomorphism only depends on the class $[u] \in \mathcal{M}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+})$. We have therefore defined the boundary operator $\partial_{H,J}$.

Theorem 22 ([Zap15, Section 3.8]).
$$\partial_{H,J}^2 = 0.$$

This allows us to define the **Lagrangian Floer homology** $HF_*(H, J : L)$ as the homology of the complex $(CF_*(H : L), \partial_{H,J})$.

There are **continuation maps** in Floer homology, defined as follows (see [Zap15, Section. 3.8]). Let (H^i, J^i) , i = 0, 1 be regular Floer data for L. Consider a homotopy of Floer data $(H^s, J^s)_{s \in \mathbb{R}}$, that is just a smooth s-dependent family of Floer data, which satisfies $(H^s, J^s) = (H^0, J^0)$ for $s \leq 0$, $(H^s, J^s) = (H^1, J^1)$ for $s \geq 1$. Given $\tilde{\gamma}_i \in \operatorname{Crit} \mathcal{A}_{H^i:L}$, i = 0, 1, we define the solution space

(3)
$$\mathcal{M}((H^s, J^s)_s; \widetilde{\gamma}_0, \widetilde{\gamma}_1) = \left\{ u: \mathbb{R} \times [0, 1] \to M \mid \overline{\partial}_{(H^s, J^s)_s} u = 0, u(\mathbb{R} \times \{0, 1\}) \subset L, u(-\infty, \cdot) = \gamma_0, u(\infty, \cdot) = \gamma_1, \widehat{\gamma}_0 \sharp u \sim \widehat{\gamma}_1 \right\},$$

where

(4)
$$\overline{\partial}_{(H^s, J^s)_s} u = \partial_s u + J^s(u) \big(\partial_t u - X_{H^s}(u) \big) \,.$$

Solutions of $\overline{\partial}_{(H^s,J^s)_s} u = 0$ can alternatively be considered as the negative gradient flow lines of the s-dependent functional $\mathcal{A}_{H^s:L}$ (see Remark 21).

We call the homotopy $(H^s, J^s)_s$ regular if for all $\tilde{\gamma}_i \in \operatorname{Crit} \mathcal{A}_{H^i:L}$ and $u \in \mathcal{M}((H^s, J^s)_s; \tilde{\gamma}_0, \tilde{\gamma}_1)$, the linearized operator D_u is onto. In this case the solution space (3) is a smooth manifold of dimension $m_{H^0:L}(\tilde{\gamma}_0) - m_{H^1:L}(\tilde{\gamma}_1)$. When this dimension vanishes, the solution space is compact, that is a finite set of points.

The continuation map

$$\Phi_{(H^s, J^s)_s} : CF_*(H^0 : L) \to CF_*(H^1 : L)$$

is defined as the linear map with matrix coefficients $C(\tilde{\gamma}_0) \to C(\tilde{\gamma}_1)$, where $m_{H^0:L}(\tilde{\gamma}_0) - m_{H^1:L}(\tilde{\gamma}_1) = 0$. This matrix coefficient equals

$$\sum_{u \in \mathcal{M}((H^s, J^s)_s; \tilde{\gamma}_0, \tilde{\gamma}_1)} C(u) \,,$$

where $C(u): C(\tilde{\gamma}_0) \to C(\tilde{\gamma}_1)$ is an isomorphism, defined as follows. The linearized operator D_u is onto and has index 0, therefore it is an isomorphism; we let $\mathfrak{o}_u \in \mathrm{d}(D_u)$ be the canonical positive orientation. The isomorphism C(u) is then defined via the composition of the following string of isomorphisms

$$d(D_{\widetilde{\gamma}_0}) \simeq d(D_u) \otimes d(D_{\widetilde{\gamma}_0}) \simeq d(D_{\widetilde{\gamma}_0} \oplus D_u) \simeq d(D_{\widetilde{\gamma}_1}),$$

the first isomorphism being $\mathfrak{o}_0 \mapsto \mathfrak{o}_u \otimes \mathfrak{o}_0$, the second being the direct sum, and the third the composition of the linear gluing and deformation isomorphisms, similarly to the definition of the boundary operator above.

It is then proved that $\Phi_{(H^s,J^s)_s}$ is a chain map, and, considering homotopies of homotopies, that the resulting map on homology is independent of the chosen homotopy of Floer data, thereby giving rise to continuation morphisms

$$\Phi_{H^0,J^0}^{H^1,J^1}: HF_*(H^0,J^0:L) \to HF_*(H^1,J^1:L) \,,$$

which satisfy

$$\Phi_{H^1,J^1}^{H^2,J^2} \Phi_{H^0,J^0}^{H^1,J^1} = \Phi_{H^0,J^0}^{H^2,J^2} \quad \text{and} \quad \Phi_{H,J}^{H,J} = \text{id} \ .$$

It follows that the continuation morphisms are isomorphisms and we thus have a direct system of graded modules $(HF_*(H, J : L))_{(H,J)}$ indexed by regular Floer data, connected by the continuation isomorphisms. The direct limit of this system is the **abstract Lagrangian Floer homology** $HF_*(L)$.

This carries a product. In order to define it, we will review the Floer PDE defined on punctured Riemann surfaces.

Recall [Sei08] that a punctured Riemann surface Σ is defined as the complement, in a compact Riemann surface $\hat{\Sigma}$ with boundary, of a finite set $\Theta \subset \hat{\Sigma}$. Elements of Θ are called the punctures. There is the notion of an end associated to a puncture. If θ denotes a boundary puncture of Σ , an end associated to it is a proper conformal embedding

$$\epsilon_{\theta} \colon [0,\infty) \times [0,1] \to \Sigma \quad \text{or} \quad \epsilon_{\theta} \colon (-\infty,0] \times [0,1] \to \Sigma$$

such that $\epsilon_{\theta}^{-1}(\partial \Sigma)$ equals $\mathbb{R}_{\pm} \times \{0, 1\}$. The punctures of Σ are given a sign and the end is defined on the positive or on the negative half-strip according to the sign. A **perturbation datum** on Σ is by definition a pair (K, I), where K is a 1-form on Σ with values in $C^{\infty}(M)$, for which $K|_{\partial\Sigma}$ vanishes along L, and I is a family of compatible almost complex structures on M parametrized by points of Σ . Assume we have chosen a regular Floer datum (H^{θ}, J^{θ}) associated to every puncture θ of Σ . Then (K, I) is said to be compatible with the Floer data if for every θ

$$\epsilon_{\theta}^* K = H_t^{\theta} dt$$
 and $I(\epsilon_{\theta}(s,t)) = J_t^{\theta}$.

Given a perturbation datum (K, I) on Σ , we can define the corresponding Floer PDE for smooth maps $u: (\Sigma, \partial \Sigma) \to (M, L)$:

(5)
$$\overline{\partial}_{K,I}u := (du - X_K)^{0,1} = 0,$$

where X_K is the 1-form on Σ with values in Hamiltonian vector fields on M defined via $\omega(X_K(\xi), \cdot) = -dK(\xi)$ for a tangent vector $\xi \in T\Sigma$. The symbol ^{0,1} denotes the complex antilinear part of a 1-form on Σ with values in a complex vector bundle over it, in this case u^*TM .

Assume now Σ has a unique positive puncture θ , negative punctures θ_i , and genus 0. Pick regular Floer data (H, J) and (H^i, J^i) associated to θ, θ_i , respectively, and choose $\{\widetilde{\gamma}_i \in \operatorname{Crit} \mathcal{A}_{H^i:L}\}_i$ and $\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}$. We can define

$$\mathcal{M}(K, I; \{\widetilde{\gamma}_i\}_i, \widetilde{\gamma}) = \left\{ u: (\Sigma, \partial \Sigma) \to (M, L) \, | \, \overline{\partial}_{K, I} u = 0, \\ u \text{ asymptotic to } \gamma_i, \gamma \text{ and } \widehat{\gamma}_1 \sharp \dots \sharp u \sim \widehat{\gamma} \right\}.$$

We call the perturbation datum (K, I) regular if for every choice of $\tilde{\gamma}_i, \tilde{\gamma}$ and every solution $u \in \mathcal{M}(K, I; \{\tilde{\gamma}_i\}_i, \tilde{\gamma})$ the linearized operator D_u is surjective. This is the case for a residual set of compatible perturbation data.

With these preliminaries in place we can define the product on $HF_*(L)$. Let Σ_* be a punctured Riemann surface obtained from D^2 by removing three boundary points θ^i , i = 0, 1, 2, where the punctures are ordered in accordance with the boundary orientation of ∂D^2 . We assign θ^0 , θ^1 the negative sign and θ^2 the positive sign. We let (H^i, J^i) be regular Floer data associated to θ^i . We pick a compatible regular perturbation datum (K, I). For $\tilde{\gamma}_i \in \operatorname{Crit} \mathcal{A}_{H^i:L}$ we consider the space $\mathcal{M}(K, I; \{\tilde{\gamma}_i\}_i)$. By assumption, this is a smooth manifold, and its dimension equals $m_{H^0:L}(\tilde{\gamma}_0) + m_{H^1:L}(\tilde{\gamma}_1) - m_{H^2:L}(\tilde{\gamma}_2) - n$. If this dimension is zero, this manifold is compact and therefore consists of a finite number of points.

Assume now that the $\tilde{\gamma}_i$ satisfy $m_{H^0:L}(\tilde{\gamma}_0) + m_{H^1:L}(\tilde{\gamma}_1) - m_{H^2:L}(\tilde{\gamma}_2) = n$. For $u \in \mathcal{M}(K, I; \{\tilde{\gamma}_i\}_i)$ the operator D_u is surjective and has index 0, therefore it is an isomorphism, and as such $d(D_u)$ possesses the canonical positive orientation, which we denote \mathfrak{o}_u .

Similarly to the definition of the matrix elements of the boundary operator above, we now define the matrix element

$$\sum_{u \in \mathcal{M}(K,I; \{\widetilde{\gamma}_i\}_i)} C(u) \colon C(\widetilde{\gamma}_0) \otimes C(\widetilde{\gamma}_1) \to C(\widetilde{\gamma}_2) \,,$$

where $C(u): C(\tilde{\gamma}_0) \otimes C(\tilde{\gamma}_1) \to C(\tilde{\gamma}_2)$ is an isomorphism defined as follows. Consider the string of isomorphisms

$$\mathrm{d}(D_{\widetilde{\gamma}_0}) \otimes \mathrm{d}(D_{\widetilde{\gamma}_1}) \simeq \mathrm{d}(D_u) \otimes \mathrm{d}(D_{\widetilde{\gamma}_0}) \otimes \mathrm{d}(D_{\widetilde{\gamma}_1}) \simeq \mathrm{d}(D_{\widetilde{\gamma}_0} \oplus D_{\widetilde{\gamma}_1} \oplus D_u) \simeq \mathrm{d}(D_{\widetilde{\gamma}_2})$$

where the first one maps $\mathfrak{o}_0 \otimes \mathfrak{o}_1 \mapsto \mathfrak{o}_u \otimes \mathfrak{o}_0 \otimes \mathfrak{o}_1$, the second one is the direct sum isomorphism, and the third one is the composition of linear gluing and deformation isomorphisms. The resulting isomorphism $d(D_{\tilde{\gamma}_0}) \otimes d(D_{\tilde{\gamma}_1}) \simeq$ $d(D_{\tilde{\gamma}_2})$ yields the isomorphism C(u).

We have therefore defined a bilinear map

$$\star_{K,I} : CF_k(H^0:L) \otimes CF_l(H^1:L) \to CF_{k+l-n}(H^2:L).$$

18

In [Zap15, Section 3.8] it is shown that this map is a chain map, therefore it induces a map on homology. Using cobordism arguments, it is shown that this induced map on homology is independent of the conformal structure on Σ_{\star} and of the choice of perturbation datum (K, I), which means that we have a correctly defined bilinear map on homology

★:
$$HF_k(H^0, J^0: L) \otimes HF_l(H^1, J^1: L) \to HF_{k+l-n}(H^2, J^2: L)$$
.

Furthermore, this map respects continuation maps, in the sense that

$$\star \circ \Phi \otimes \Phi = \Phi \circ \star,$$

therefore it induces a product \star on the abstract Floer homology $HF_*(L)$.

It is further shown *ibid.* that this product operation is associative and has a unit, thus $HF_*(L)$ is an associative unital algebra.

2.2.2. Periodic orbit Floer homology. Let

$$\Omega = \left\{ x: S^1 \to M \,|\, [x] = 0 \in \pi_1(M) \right\}$$

be the contractible loop space of M. We view S^2 as the Riemann sphere $\mathbb{C} \cup \{\infty\}$, and let $\dot{S}^2 = S^2 \setminus \{1\}$ be the sphere with one puncture. We endow it with the standard positive end ϵ : $[0, \infty) \times S^1 \to \dot{S}^2$ defined by

$$\epsilon(z) = \frac{e^{2\pi z} - i}{e^{2\pi z} + i},$$

where we identify $S^1 = \mathbb{R}/\mathbb{Z}$. A capping of $x \in \Omega$ is a b-smooth map $\hat{x}: \dot{S}^2 \to M$ with $\hat{x}(\epsilon(s,t)) \to x(t)$ as $s \to \infty$. Similarly to §2.2.1 we have the notion of equivalence of two cappings \hat{x}, \hat{x}' of x, meaning that their pregluing $\hat{x}\sharp(-\hat{x}')$ is a contractible sphere; we denote this by $\hat{x} \sim \hat{x}'$. We call two pairs $(x, \hat{x}), (x', \hat{x}')$ equivalent if x = x' and $\hat{x} \sim \hat{x}'$. The class of (x, \hat{x}) is denoted by $[x, \hat{x}]$. Let

$$\Omega = \left\{ [x, \hat{x}] \, | \, x \in \Omega, \hat{x} \text{ a capping for } x \right\}.$$

This comes with the obvious projection $\widetilde{\Omega} \to \Omega$ making it into a covering space. We denote elements of $\widetilde{\Omega}$ via \widetilde{x} and usually it will stand for $[x, \widehat{x}]$.

For a Hamiltonian $H: S^1 \times M \to \mathbb{R}$ we have the action functional

$$\mathcal{A}_H: \widetilde{\Omega} \to \mathbb{R}, \quad \mathcal{A}_H(\widetilde{x} = [x, \widehat{x}]) = \int_{S^1} H_t(x(t)) dt - \int_{\dot{S}^2} \widehat{x}^* \omega$$

Its critical point set Crit \mathcal{A}_H is the set of classes $[x, \hat{x}]$ for which x is a periodic orbit of $H, \dot{x} = X_H(x)$.

We call H nondegenerate if for each $\tilde{x} \in \operatorname{Crit} \mathcal{A}_H$ the linearized flow $\phi_{H*}: T_{x(0)}M \to T_{x(0)}M$ does not have 1 as an eigenvalue. In this case we have a well-defined Conley–Zehnder index¹⁰

$$m_H$$
: Crit $\mathcal{A}_H \to \mathbb{Z}$.

¹⁰Similarly to the Lagrangian case we normalize it to equal the Morse index of a critical point of H in case it is a C^2 -small Morse function, and to satisfy $m_H([x, A \sharp \hat{x}]) = m_H([x, \hat{x}]) - 2c_1(A)$ for $A \in \pi_2(M, x(0))$.

Given a time-periodic compatible almost complex structure J on M, we can define the operator

$$D_{\widehat{x}}: W^{1,p}(\dot{S}^2; \widehat{x}^*TM) \to L^p(\dot{S}^2; \Omega^{0,1} \otimes \widehat{x}^*TM),$$

and assemble these into a bundle $D_{\tilde{x}}$ of Fredholm operators over the space of cappings in class \tilde{x} , similarly to the Lagrangian case. We note that ind $D_{\tilde{x}} = 2n - m_H(\tilde{x})$. We have

Lemma 23 ([Zap15, Section 3.7]). The determinant line bundle $d(D_{\tilde{x}})$ is orientable.

We let $C(\tilde{x})$ be the free \mathbb{Z} -module of rank 1 whose two generators are the two possible orientations of the line bundle $d(D_{\tilde{x}})$. The Floer complex of H as a \mathbb{Z} -module is

$$CF_*(H) = \bigoplus_{\widetilde{x} \in \operatorname{Crit} \mathcal{A}_H} C(\widetilde{x}).$$

This is graded by m_H and its generators are assigned actions given by \mathcal{A}_H .

We can similarly define the space of solutions of the Floer PDE,

$$\mathcal{M}(H,J;\widetilde{x}_{-},\widetilde{x}_{+})$$

for $\tilde{x}_{\pm} \in \operatorname{Crit} \mathcal{A}_H$; we put $\mathcal{M}(H, J; \tilde{x}_-, \tilde{x}_+) = \widetilde{\mathcal{M}}(H, J; \tilde{x}_-, \tilde{x}_+)/\mathbb{R}$. We call (H, J) a regular Floer datum if for all \tilde{x}_{\pm} and every solution u in this space the linearized operator D_u is onto. Then this solution space is a smooth manifold of dimension $m_H(\tilde{x}_-) - m_H(\tilde{x}_+)$. The quotient $\mathcal{M}(H, J; \tilde{x}_-, \tilde{x}_+)$ is a compact manifold whenever it is zero-dimensional, and therefore it is a finite set of points.

Assume that $m_H(\tilde{x}_-) - m_H(\tilde{x}_+) = 1$ and that $u \in \mathcal{M}(H, J; \tilde{x}_-, \tilde{x}_+)$. Then D_u is surjective and has index 1. Its kernel carries a canonical orientation and therefore D_u is canonically oriented; we let this orientation be denoted via ∂_u .

Similarly to the Lagrangian case, we have the isomorphism

$$C(u): C(\widetilde{x}_{-}) \to C(\widetilde{x}_{+})$$

for such u given by applying the direct sum, linear gluing and deformation isomorphisms. The matrix element $C(\tilde{x}_{-}) \to C(\tilde{x}_{+})$ of the boundary operator $\partial_{H,J}: CF_k(H) \to CF_{k-1}(H)$ is the sum

$$\sum_{[u]\in\mathcal{M}(H,J;\widetilde{x}_{-},\widetilde{x}_{+})} C(u): C(\widetilde{x}_{-}) \to C(\widetilde{x}_{+}) \,.$$

It is proved in [Zap15, Section 3.8] that $\partial_{H,J}^2 = 0$ and we let $HF_*(H,J)$ be the homology of the resulting chain complex $(CF_*(H), \partial_{H,J})$.

Continuation maps

$$\Phi^{H^1,J^1}_{H^0,J^0} \colon HF_*(H^0,J^0) \to HF_*(H^1,J^1)$$

are defined similarly to the Lagrangian case, and it is shown that they behave well with respect to composition and that $\Phi_{H,J}^{H,J} = id$, implying that they are

isomorphisms. We let $HF_*(M)$ be the abstract Floer homology, which is the limit of the direct system of graded modules $(HF_*(H,J))_{(H,J)}$ indexed by regular Floer data, and connected by the continuation isomorphisms.

The product

*:
$$HF_k(H^0, J^0) \otimes HF_l(H^1, J^1) \to HF_{k+l-2n}(H^2, J^2)$$

is defined using the moduli space of solutions of the Floer PDE (5) on the surface Σ_* obtained from S^2 by removing three points. It is shown to be independent of the conformal structure on Σ_* and the perturbation datum used to define it, and moreover to behave well with respect to continuation isomorphisms, that is $* \circ \Phi \otimes \Phi = \Phi \circ *$. This means that we have a well-defined product * on the abstract Floer homology $HF_*(M)$. It is further shown to be supercommutative, associative, and unital.

There is an additional algebraic structure, namely the quantum module action of $HF_*(M)$ on $HF_*(L)$. To describe it, consider the surface Σ_{\bullet} obtained from D^2 by removing two boundary and one interior point, where the resulting interior puncture and one of the boundary ones are given the negative sign, while the remaining boundary puncture is given the positive sign. Using solutions of the Floer PDE for Σ_{\bullet} , we can define a bilinear map

•: $HF_k(H, J) \otimes HF_l(H^0, J^0: L) \to HF_{k+l-2n}(H^1, J^1: L)$,

which behaves well with respect to continuation isomorphisms, thereby defining a bilinear operation $\bullet: HF_*(M) \otimes HF_*(L) \to HF_*(L)$. This is shown to make $HF_*(L)$ into an algebra over the algebra $HF_*(M)$, meaning we have the structure of an $HF_*(M)$ -module on $HF_*(L)$ and in addition the latter is a superalgebra over the former.¹¹

2.3. Quantum homology. Here we briefly describe Lagrangian and symplectic quantum homology. It is advantageous to think of quantum homology as Morse–Bott Floer homology for the zero Hamiltonian and a time-independent almost complex structure. Lagrangian quantum homology was defined by Biran–Cornea [BC09, BC12]. We give here a different formulation.

2.3.1. Lagrangian quantum homology. A quantum datum for L is a triple $\mathcal{D} = (f, \rho, J)$ where $f: L \to \mathbb{R}$ is a Morse function, ρ is a Riemannian metric on L such that the pair (f, ρ) is Morse–Smale, and J is an ω -compatible almost complex structure. The corresponding quantum complex is the \mathbb{Z} -module

$$QC_*(\mathcal{D}:L) = \bigoplus_{q \in \operatorname{Crit} f} \bigoplus_{A \in \pi_2(M,L,q)} C(q,A).$$

Here C(q, A) is the rank 1 free \mathbb{Z} -module generated by the orientations of the determinant line bundle d $(D_A \sharp T \mathcal{S}(q))$, where the operator family $D_A \sharp T \mathcal{S}(q)$

¹¹This means that we have $a \bullet (\alpha \star \beta) = (a \bullet \alpha) \star \beta = (-1)^{\text{sign}} \alpha \star (a \bullet \beta)$ with a suitable Koszul sign.

is defined as follows. Denote

$$C_A^{\infty} = \left\{ u \in C^{\infty}(D^2, \partial D^2, 1; M, L, q) \, | \, [u] = A \right\}.$$

Let $u \in C^{\infty}_A$ and let

$$D_u: W^{1,p}(D^2, S^1; u^*TM, (u|_{S^1})^*TL) \to L^p(D^2; \Omega^{0,1} \otimes u^*TM)$$

be the formal linearization of the Cauchy–Riemann operator at u. See [Zap15, Section 4.2] for a more precise definition. The operators D_u for different u assemble into a Fredholm family D_A over C_A^{∞} . We let

 $D_u \sharp T \mathcal{S}(q)$

be the restriction of D_u to the subspace

$$\{\xi \in W^{1,p}(D^2, S^1; u^*TM, (u|_{S^1})^*TL) \,|\, \xi(1) \in T_q \mathcal{S}(q)\}.$$

Similarly the operators $D_u \sharp T S(q)$ assemble into a family $D_A \sharp T S(q)$. We have

Lemma 24 ([Zap15, Section 4.2]). The line bundle $d(D_A \sharp TS(q))$ is orientable if and only if assumption (O) holds.

Thus we have defined the module C(q, A) and therefore also the module $QC_*(\mathcal{D}: L)$. The grading on the latter module is determined by requiring the elements of C(q, A) to have degree $\operatorname{ind}_f q - \mu(A)$.

The boundary operator $\partial_{\mathcal{D}}: QC_*(\mathcal{D}:L) \to QC_{*-1}(\mathcal{D}:L)$ is defined using the moduli spaces of pearls defined by Biran–Cornea [BC09]. When suitable genericity conditions are satisfies (see *ibid*. for their precise formulation), we call the datum \mathcal{D} regular. See [Zap15, Section 4.2] for a detailed description of $\partial_{\mathcal{D}}$ and for a proof that $\partial_{\mathcal{D}}^2 = 0$ in the present context. We denote by $QH_*(\mathcal{D}:L)$ the homology of $(QC_*(\mathcal{D}:L), \partial_{\mathcal{D}})$. It is the quantum homology associated to the datum \mathcal{D} .

There exist continuation isomorphisms relating the different $QH_*(\mathcal{D}:L)$, which we define below in §2.4.1 and we let $QH_*(L)$ be the limit of the corresponding direct system of graded modules. This is the **quantum homology** of L. We can also define an associative product \star on $QH_*(L)$. It has degree -n.

This product admits a unit, constructed as follows. Assume f has a unique maximum q. The module C(q, 0) in this case is canonically isomorphic to \mathbb{Z} . We denote by $1_q \in C(q, 0)$ the element corresponding under this isomorphism to $1 \in \mathbb{Z}$. It can be shown that this is a cycle and that it acts as the unit for \star on the chain level. It follows that its class $[L] \in QH_*(L)$ is independent of \mathcal{D} and that it is the unit for the product.

2.3.2. Quantum homology of M. This is defined similarly to the Lagrangian case. A quantum datum on M is a triple $\mathcal{D} = (f, \rho, J)$ where (f, ρ) is a Morse–Smale pair on M and J is a compatible almost complex structure.

The quantum complex as a module is

$$QC_*(\mathcal{D}) = \bigoplus_{q \in \operatorname{Crit} f} \bigoplus_{A \in \pi_2(M,q)} C(q,A)$$

where C(q, A) is the rank 1 free \mathbb{Z} -module generated by orientations of the operator family $D_A \sharp T \mathcal{S}(q)$. Here D_A is the Fredholm family composed of the formal linearizations

$$D_u: W^{1,p}(S^2; u^*TM) \to L^p(S^2; \Omega^{0,1} \otimes u^*TM)$$

for smooth maps $u: S^2 \to M$ in class A.

REMARK 25. This family is always orientable. In fact, it possesses a canonical orientation, which can be seen as follows. Each D_u is a reallinear Cauchy–Riemann operator. On a closed Riemann surface the set of complex-linear Cauchy–Riemann operators is a deformation retract of the set of real-linear operators. The (real) determinant line bundle of the set of complex-linear operators has a canonical orientation since for each such operator its kernel and cokernel are complex vector spaces, which thus are canonically oriented. Now just pull this orientation on the set of complex-linear operators to the set of all real-linear operators by the retraction map.

The family $D_A \sharp T S(q)$ has as representatives the operators $D_u \sharp T S(q)$, where $D_u \sharp T S(q)$ is the restriction of D_u to the subspace

$$\{\xi \in W^{1,p}(S^2, u^*TM) | \xi(1) \in T_q \mathcal{S}_f(q) \}.$$

The family $D_A \sharp TS(q)$ is orientable as well and therefore we have defined the modules C(q, A). Its elements by definition have degree $\operatorname{ind}_f q - 2c_1(A)$.

The boundary operator $\partial_{\mathcal{D}}$ on $QC_*(\mathcal{D})$ is in fact just the Morse boundary operator, enriched by the local system $\pi_2(M)$. See [Zap15, Section 4.5] for more details. We thus have the quantum homology $QH_*(\mathcal{D})$ as the homology of the resulting chain complex. This is shown to be independent of \mathcal{D} using continuation isomorphisms in Morse homology; thus we obtain the abstract quantum homology $QH_*(M)$.

It carries a supercommutative associative product * with unit. There is also the bilinear map

•:
$$QH_*(M) \otimes QH_*(L) \to QH_*(L)$$

of degree -2n, which makes $QH_*(L)$ into a superalgebra over $QH_*(M)$ similarly to the case of Floer homology, see §2.2.2.

2.4. Piunikhin–Salamon–Schwarz isomorphisms. These were first defined in [PSS96], and consequently extended to various other settings [KM05, Alb08, BC09], [Zap15, Section 5]. They are commonly abbreviated to "PSS", and they relate Floer and quantum homologies.

2.4.1. Lagrangian PSS. Fix a regular Floer datum (H, J) and a regular quantum datum \mathcal{D} for L. There are well-defined mutually inverse isomorphisms

$$\operatorname{PSS}_{\mathcal{D}}^{H,J} : QH_*(\mathcal{D}:L) \to HF_*(H,J:L),$$

$$\operatorname{PSS}_{H,J}^{\mathcal{D}} : HF_*(H,J:L) \to QH_*(\mathcal{D}:L),$$

which commute with continuation morphisms in Floer homology. For instance, they allow us to define continuation isomorphisms in quantum homology, as follows. If \mathcal{D}' is another regular quantum datum for L, we define the continuation morphism

$$\Phi_{\mathcal{D}}^{\mathcal{D}'} := \mathrm{PSS}_{H,J}^{\mathcal{D}'} \mathrm{PSS}_{\mathcal{D}}^{H,J} : QH_*(\mathcal{D}:L) \to QH_*(\mathcal{D}':L) \,.$$

Since the PSS maps commute with Floer continuation maps, this is independent of the Floer datum (H, J), and since the two PSS morphisms above are mutually inverse, these satisfy

$$\Phi_{\mathcal{D}'}^{\mathcal{D}'} \Phi_{\mathcal{D}}^{\mathcal{D}'} = \Phi_{\mathcal{D}}^{\mathcal{D}''} \text{ and } \Phi_{\mathcal{D}}^{\mathcal{D}} = \mathrm{id}$$

Continuation morphisms in quantum homology can also be defined directly, as in [BC09]. The properties of the continuation maps allow us to define the abstract quantum homology $QH_*(L)$ as the direct limit of the system $(QH_*(\mathcal{D}:L))_{\mathcal{D}}$. We then have the following naturally induced isomorphism:

(6)
$$\operatorname{PSS}^{H,J}: QH_*(L) \to HF_*(H,J:L)$$

We can also show that the PSS maps are unital algebra isomorphisms. In particular, if (H^i, J^i) , i = 0, 1, 2, are regular Floer data, then the following diagram commutes:

$$QH_*(L) \otimes QH_*(L) \xrightarrow{\star} QH_*(L)$$

$$\downarrow^{\mathrm{PSS}^{H^0,J^0} \otimes \mathrm{PSS}^{H^1,J^1}} \qquad \qquad \downarrow^{\mathrm{PSS}^{H^2,J^2}}$$

$$HF_*(H^0,J^0:L) \otimes HF_*(H^1,J^1:L) \xrightarrow{\star} HF_*(H^2,J^2:L)$$

2.4.2. PSS isomorphism between periodic orbit Floer homology and quantum homology. Analogously we can define the PSS isomorphisms between the periodic orbit Floer homology $HF_*(H, J)$ for a regular Floer datum (H, J) and the quantum homology $QH_*(\mathcal{D})$ for a regular quantum datum \mathcal{D} :

 $PSS: QH_*(\mathcal{D}) \to HF_*(H,J) \text{ and } PSS: HF_*(H,J) \to QH_*(\mathcal{D}).$

These similarly respect the Floer continuation maps, and therefore induce the isomorphisms

$$\mathrm{PSS}^{H,J}: QH_*(M) \to HF_*(H,J) \quad \text{and} \quad \mathrm{PSS}_{H,J}: HF_*(H,J) \to QH_*(M) \,.$$

The induced isomorphism $QH_*(M) \simeq HF_*(M)$ is a graded algebra isomorphism. Moreover, the superalgebra structures of $QH_*(L)$ over $QH_*(M)$ and of $HF_*(L)$ over $HF_*(M)$ are intertwined by the PSS isomorphisms.

2.5. Change of coefficients, twisting, quotient complexes, Novikov rings. Here we describe various algebraic operations on the aforementioned complexes.

2.5.1. Coefficients in a ring of characteristic 2. Regardless of whether the assumption (O) is satisfied, we can define the Lagrangian Floer and quantum homology of L with coefficients in an arbitrary (commutative) ring R of characteristic 2. If (H, J) is a regular Floer datum for L, we put

$$CF_*(H:L;R) = \bigoplus_{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H \cdot L}} R \cdot \widetilde{\gamma},$$

and the boundary operator has matrix elements

$$#_2\mathcal{M}(H,J;\widetilde{\gamma}_-,\widetilde{\gamma}_+): R\cdot\widetilde{\gamma}_- \to R\cdot\widetilde{\gamma}_+$$

for $\tilde{\gamma}_{\pm} \in \operatorname{Crit} \mathcal{A}_{H:L}$ of index difference 1. Quantum homology is similarly defined, and all of the above constructions go through.

2.5.2. Change of coefficients and twisting by a local system. The above complexes are all defined over \mathbb{Z} . One can tensor them with an arbitrary ring Rto obtain complexes over it. Another possibility is to use twisted coefficients. Recall that if X is a topological space, a **local system** of groups on X is a functor from the fundamental groupoid of X to the category of groups. More concretely, it means that we are given a group G_x for every $x \in X$, and for every homotopy class of paths from x to y an isomorphism $G_x \simeq G_y$, such that the constant path at x induces the identity on G_x and concatenation commutes with composition of isomorphisms. Similarly we have local systems of modules and so on. Borrowing terminology from geometry, we call the isomorphisms corresponding to paths **parallel transport operators**.

Assume \mathcal{L} is a local system of R-modules over Ω_L and for a path u in Ω_L from $\widetilde{\gamma}_-$ to $\widetilde{\gamma}_+$ let $\mathcal{P}_u: \mathcal{L}_{\widetilde{\gamma}_-} \to \mathcal{L}_{\widetilde{\gamma}_+}$ be the parallel transport map. Let

$$CF_*(H, J: L; \mathcal{L}) = \bigoplus_{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}} C(\widetilde{\gamma}) \otimes_{\mathbb{Z}} \mathcal{L}_{\widetilde{\gamma}}.$$

For $\tilde{\gamma}_{\pm} \in \operatorname{Crit} \mathcal{A}_{H:L}$ with $|\tilde{\gamma}_{-}| = |\tilde{\gamma}_{+}| - 1$ we define the homomorphism

$$\sum_{[u]\in\mathcal{M}(H,J;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+})} C(u)\otimes\mathcal{P}_{u}: C(\widetilde{\gamma}_{-})\otimes\mathcal{L}_{\widetilde{\gamma}_{-}}\to C(\widetilde{\gamma}_{+})\otimes\mathcal{L}_{\widetilde{\gamma}_{+}}$$

where we use the lift of u to Ω_L in the parallel transport map \mathcal{P}_u . We let $\partial_{H,J} \otimes \mathcal{P}$ be the operator whose matrix elements are these homomorphisms. It is straightforward to show that $(\partial_{H,J} \otimes \mathcal{P})^2 = 0$ and we let $HF_*(H, J : L; \mathcal{L})$ be the resulting homology. Continuation maps carry over to yield the abstract twisted Lagrangian Floer homology $HF_*(L; \mathcal{L})$.

Similarly one can define twisted quantum homology $QH_*(L; \mathcal{L})$. For the reader familiar with the construction of the pearly complex we mention that a pearly trajectory can be viewed as a path in $\widetilde{\Omega}_L$, and the above construction goes through. PSS isomorphisms can also be adapted to the twisted case.

REMARK 26. There is one particularly important case of twisted coefficients, which corresponds to what it usually meant by a local system on L. Namely, let \mathcal{B} be a local system of R-modules on L. There are natural evaluation maps

$$\operatorname{ev}_i \colon \Omega_L \to L$$
, $\operatorname{ev}_i(\widetilde{\gamma} = [\gamma, \widehat{\gamma}]) = \gamma(i)$ for $i = 0, 1$.

Consider the following local system on $\widetilde{\Omega}_L$:

$$\mathcal{E}_{\widetilde{\gamma}} = \operatorname{Hom}_{R}\left((\operatorname{ev}_{0}^{*}\mathcal{B})_{\widetilde{\gamma}}, (\operatorname{ev}_{1}^{*}\mathcal{B})_{\widetilde{\gamma}}\right) = \operatorname{Hom}_{R}(\mathcal{B}_{\gamma(0)}, \mathcal{B}_{\gamma(1)}).$$

Let \mathcal{P} be the parallel transport operator of \mathcal{B} . A path in $\widetilde{\Omega}_L$ can be viewed as a map $u: \mathbb{R} \times [0,1] \to M$ mapping the boundary to L. Assume $u(\pm \infty, \cdot) = \gamma_{\pm}$. The parallel transport operator of \mathcal{E} along u,

$$\mathcal{P}_{u}^{\mathcal{E}} \colon \operatorname{Hom}_{R}(\mathcal{B}_{\gamma_{-}(0)}, \mathcal{B}_{\gamma_{-}(1)}) \to \operatorname{Hom}_{R}(\mathcal{B}_{\gamma_{+}(0)}, \mathcal{B}_{\gamma_{+}(1)}),$$

is given by

$$\mathcal{P}_u^{\mathcal{E}}(f) = \mathcal{P}_{u(\cdot,1)} \circ f \circ \mathcal{P}_{u(\cdot,0)}^{-1}.$$

2.5.3. Quotient complexes. The usual approach used to define Floer homology with arbitrary coefficients is to use the so-called coherent orientations, introduced in [FH93]. The idea is to orient the various moduli spaces of Floer curves so that gluing maps are, in an appropriate sense, orientationpreserving. The approach chosen here, which follows that of [Zap15], bypasses the need to orient moduli spaces by introducing canonical \mathbb{Z} -modules such as $C(\tilde{\gamma})$ above, and by using the canonical orientations on the moduli spaces used in the definition of the boundary operator (namely the one given by the action of \mathbb{R}) and other algebraic operations. This allows us to define the above canonical complex $(CF_*(H:L), \partial_{H,J})$ without any additional choices beyond the Floer datum (H, J).

The drawback of our canonical Floer and quantum complexes is that they all distinguish between homotopy classes of cappings rather than the more usual equivalence relations in Floer theory, such as equivalence in homology or having equal area, which are often needed in applications. This leads to the need of forming quotient complexes. This process is summarized in the current subsection. Even though we use \mathbb{Z} coefficients below, everything goes through over any ring.

In order to form these quotient complexes one needs to impose additional assumptions on L and to choose further structures, such as a relative Pin^{\pm} -structure. It is at this point that our approach and the more conventional one using coherent orientations converge. However we wish to emphasize that we do not pick coherent orientations on moduli spaces, but rather use the additional structures in order to form quotient complexes as described below, which in our opinion is a more natural procedure, albeit it is less traditional.

We first describe the case of periodic orbit Floer homology and the quantum homology of M. Quotient complexes are formed using local systems. Given a local system G (of groups, modules, *etc.*) on a topological space X, a subsystem G' < G is a collection of subgroups $G'_x < G_x$ for every x, which are preserved by the parallel transport maps. In particular this means that the action of $\pi_1(X, x)$ on G_x preserves G'_x .

The fundamental local system we need to use is $\pi_2(M)$, whose value at $q \in M$ is $\pi_2(M,q)$ and where for $q, q' \in M$ and a homotopy class of paths from q to q' the isomorphism $\pi_2(M,q) \simeq \pi_2(M,q')$ is given by the standard action of paths on homotopy groups [Hat02]. It acts on $\tilde{\Omega}$, in the following sense. Fix $q \in M$ and let $\tilde{\Omega}_q = \{\tilde{x} \mid x(0) = q\}$. Then $\pi_2(M,q)$ acts on $\tilde{\Omega}_q$ by appending spheres to cappings.

In order to form a quotient complex for which spectral invariants still make sense, we need to use the local system $\pi_2^0(M)$, which is the subsystem of $\pi_2(M)$ whose value at q is the group $\pi_2^0(M, q) := \ker \omega$. Fix a subsystem $G < \pi_2^0(M)$. The action of $\pi_2(M)$ on $\tilde{\Omega}$ restricts to an action of G. We let $\tilde{\Omega}/G$ be the quotient space. It forms a covering space $\tilde{\Omega}/G \to \Omega$. Note that the fiber of this projection over $x \in \Omega$ consists of the set of equivalence classes of cappings for x, where two cappings are equivalent if and only if their difference defines an element of $G_{x(0)}$.

The action functional of a Hamiltonian H on M, $\mathcal{A}_H: \widetilde{\Omega} \to \mathbb{R}$, descends to $\mathcal{A}_H^G: \widetilde{\Omega}/G \to \mathbb{R}$. We see that $\operatorname{Crit} \mathcal{A}_H^G = \operatorname{Crit} \mathcal{A}_H/G$ in an obvious sense. In [Zap15, Section 7] it is described how the quotient map $\operatorname{Crit} \mathcal{A}_H \to \operatorname{Crit} \mathcal{A}_H^G$ leads to a chain map

$$\left(CF_*(H), \partial_{H,J}\right) \to \left(CF^G_*(H), \partial^G_{H,J}\right),$$

where

$$CF^G_*(H) = \bigoplus_{\widetilde{x}^G \in \operatorname{Crit} \mathcal{A}^G_H} C\left(\widetilde{x}^G\right),$$

with $C(\tilde{x}^G)$ being the limit of the direct system of modules $C(\tilde{x})$ for \tilde{x} ranging in the class $\tilde{x}^G \in \tilde{\Omega}/G$, and connected by canonical isomorphisms ([Zap15, Section 7.2]).

REMARK 27. These canonical isomorphisms arise as follows. Recall that $C(\tilde{x})$ is generated by the orientations of the family $D_{\tilde{x}}$. For two different cappings \hat{x}, \hat{x}' of x one can construct, using the gluing, direct sum, and deformation isomorphisms, a canonical isomorphism of determinant lines

$$d(D_{\widehat{x}}) \simeq d(D_{\widehat{x}'}) \otimes d(T_{x(0)}M) \otimes d(D_{\widehat{x}\sharp - \widehat{x}'}).$$

By Remark 25 the operator $D_{\widehat{x}\sharp-\widehat{x}'}$, being a real-linear Cauchy–Riemann operator on a closed Riemann surface, possesses a canonical orientation, and the same is of course true of $T_{x(0)}M$. This yields an isomorphism $d(D_{\widehat{x}}) \simeq$ $d(D_{\widehat{x}'})$, leading to an isomorphism $C(\widetilde{x}) \simeq C(\widetilde{x}')$, where $\widetilde{x} = [x, \widehat{x}], \ \widetilde{x}' =$ $[x, \widehat{x}']$. It can be checked that all these isomorphisms satisfy the cocycle condition and therefore lead to a uniform identification of all the modules $C(\widetilde{x})$ for \widetilde{x} ranging over \widetilde{x}^G . The boundary operator $\partial_{H,J}^G$ is well-defined by the requirement that the above quotient map be a chain map. Thus there is an induced graded morphism $HF_*(H,J) \to HF_*^G(H,J)$. These respect continuation morphisms and thus lead to a canonical morphism $HF_*(M) \to HF_*^G(M)$.

It is also shown *ibid*. that the product structure on the Floer complexes descends to a product structure on the quotient modules. The morphism $HF_*(M) \to HF_*^G(M)$ is a morphism of algebras.

We next discuss how the Novikov ring appears in this formulation. In general the local system $\pi_2(M)$ is not constant, meaning the fundamental group $\pi_1(M,q)$ acts on $\pi_2(M,q)$ nontrivially, and this prohibits the existence of a module structure over "the group ring" $\mathbb{Z}[\pi_2(M)]$, since $\pi_2(M)$ is not a group, but rather just a groupoid. However in certain cases there is such a structure. Namely, in case F is a constant local system over M and $\phi: \pi_2(M) \to F$ is a surjective morphism of local systems such that $G = \ker \phi$, the quotient complex $CF^G_*(H)$ inherits the structure of a module over the group ring $\mathbb{Z}[F]$, and the corresponding homology $HF^G_*(M)$ the structure of an algebra over $\mathbb{Z}[F]$.

Two basic examples of this are as follows. If $F = \mathbb{Z}$ and $\phi(A) = c_1(A)/N_M$, where N_M is the minimal Chern number of M, then $G = \pi_2^0(M)$ and the complex $CF_*^{\pi_2^0(M)}(H)$ inherits the structure of a module over the group ring $\mathbb{Z}[\mathbb{Z}] = \mathbb{Z}[t, t^{-1}]$, where t is a formal variable of degree $-2N_M$. The reader will identify this as the Novikov ring, familiar in Floer homology.

Another example is $F = H_2^S(M) = \operatorname{im}(\pi_2(M) \to H_2(M;\mathbb{Z}))$ with ϕ being the Hurewicz morphism. In this case G is its kernel, and the complex $CF_*^G(H)$ carries the structure of a module over the group ring $\mathbb{Z}[H_2^S(M)]$.

In [Zap15, Section 7] there is a description of a completely analogous construction of quotient complexes in quantum homology. The result is the canonical quantum homology $QH^G_*(M)$. Taking quotient complexes respects the PSS isomorphisms, therefore we have canonically $QH^G_*(M) = HF^G_*(M)$, which is an algebra isomorphism; in case F is a constant local system and G is the kernel of a surjective morphism of local systems $\phi: \pi_2(M) \to F$, it intertwines the resulting structures of algebras over $\mathbb{Z}[F]$.

We have a similar situation concerning the Lagrangian case, but there are necessary additional structures and assumptions due to the lack of canonical orientations. We briefly describe this, referring the reader to [Zap15, Section 7] for details.

Analogously to the absolute case, we need to use local systems on L. We have the local system $\pi_2(M, L)$ and its subsystem $\pi_2^0(M, L) = \ker \omega$. There is an action of $\pi_2(M, L)$ on $\widetilde{\Omega}_L$. Given a subsystem $G < \pi_2^0(M, L)$, we let $\widetilde{\Omega}_L/G$ be the quotient of the induced action of G.

Assume (H, J) is a regular Floer datum for L. The action functional $\mathcal{A}_{H:L}: \widetilde{\Omega}_L \to \mathbb{R}$ descends to $\mathcal{A}_{H:L}^G: \widetilde{\Omega}_L/G \to \mathbb{R}$ and we have the obvious quotient map $\operatorname{Crit} \mathcal{A}_{H:L} \to \operatorname{Crit} \mathcal{A}_{H:L}^G$. However, this does not immediately lead to the formation of a quotient complex, as we have to identify the

modules $C(\tilde{\gamma})$ for $\tilde{\gamma}$ ranging in a class $\tilde{\gamma}^G \in \operatorname{Crit} \mathcal{A}^G_{H:L}$, which in general is impossible if we wish the boundary operator $\partial_{H,J}$ to induce a boundary operator on the quotient complex.¹²

A sufficient condition for this to be possible is to assume that L is relatively Pin^{\pm} (Definition 18). If this holds, one can define the notion of a relative Pin^{\pm} structure on L [Zap15, Section 7]. A choice of such a relative Pin^{\pm} structure on L determines a system of orientations on the determinant line bundles¹³ d($D_A \sharp 0$), which is sufficient to make the necessary identifications of the modules $C(\tilde{\gamma})$ for $\tilde{\gamma}$ running over a class $\tilde{\gamma}^G \in \tilde{\Omega}_L^G$, see *ibid*.

Fixing a relative Pin^{\pm} structure on L we obtain the quotient complex $\left(CF^G_*(H:L), \partial^G_{H,J}\right)$ and the quotient chain map

$$(CF_*(H:L),\partial_{H,J}) \to (CF^G_*(H:L),\partial^G_{H,J}).$$

This quotient map determines a unique bilinear map giving the abstract homology $HF^G_*(L)$ the structure of an associative unital algebra for which the natural map $HF_*(L) \to HF^G_*(L)$ is an algebra morphism. The choice of a relative Pin^{\pm} structure yields canonical isomorphisms

$$d(D_u \sharp 0) \simeq d \left(T_{u(1)}L\right)^{\otimes \mu(u)}$$

for smooth maps $u: (D^2, S^1) \to (M, L)$. Therefore in order to be able to orient the whole collection of operators $D_u \sharp 0$ with varying u, we need to restrict ourselves to the space of smooth maps u with $w_1(u) = 0$, where $w_1: \pi_2(M, L) \to \mathbb{Z}_2$ is the natural morphism of local systems defined by pulling back the first Stiefel–Whitney class $w_1(TL)$ by the boundary homomorphism $\pi_2(M, L) \to \pi_1(L)$.

A module structure over a Novikov ring is similar to the absolute case. If Fis a constant local system on L and ϕ : ker $w_1 \to F$ is a surjective morphism of local systems with $G = \ker \phi$, using the relative Pin^{\pm} structure one can endow $CF^G_*(H,L)$ with the structure of a module over the group ring $\mathbb{Z}[F]$. Again, there are two basic examples, ϕ : ker $w_1 \to \mathbb{Z}$, $\phi = \mu/\operatorname{lcm}(2, N_L)$, in which case one recovers the structure of a module over the Novikov ring $\mathbb{Z}[\mathbb{Z}] = \mathbb{Z}[t, t^{-1}].$ The other example is the relative Hurewicz morphism ϕ : ker $w_1 \to H_2^{D, \text{or}}(M, L)$, where $H_2^{D, \text{or}}(M, L) = \text{ker}\left(H_2^D(M, L) \xrightarrow{w_1} \mathbb{Z}_2\right)$ and $H_2^D(M,L) = \operatorname{im}\left(\pi_2(M,L) \to H_2(M,L;\mathbb{Z})\right)$ is the image of the relative Hurewicz morphism. In this case $CF^G_*(H:L)$ acquires the structure of a module over $\mathbb{Z}[H_2^{D,\text{or}}(M,L)]$. Note that in general only the "orientable" part of $H_2^D(M,L)$ can act on the quotient Floer complex. An example of this is described in $\S2.6.1$.

¹²Of course we can identify the modules $C(\tilde{\gamma})$ for various $\tilde{\gamma}$, since all of them are abstractly isomorphic to \mathbb{Z} , but such an identification will not in general lead to a welldefined boundary operator on the quotient complex.

¹³The operator family $D_A \sharp 0$ is defined as follows. Its representative over $u \in C_A^{\infty}$ is the restriction of D_u to the subspace of $W^{1,p}(D^2, S^1; u^*TM, (u|_{S^1})^*TL)$ consisting of sections which vanish at $1 \in S^1$.

We also note that the quantum module structure

$$QH_*(M) \otimes QH_*(L) \to QH_*(L)$$

descends to a well-defined module operation

$$QH^{G_M}_*(M) \otimes QH^{G_L}_*(L) \to QH^{G_L}_*(L)$$

provided $G_M < \pi_2^0(M)$ and $G_L < \pi_2^0(M, L)$ are local systems such that $QH_*^{G_L}(L)$ is well-defined (by the above, this is the case if a relative Pin^{\pm} structure on L has been chosen), and such that for any $q \in L$ the natural morphism $\pi_2^0(M, q) \to \pi_2^0(M, L, q)$ maps $G_{M,q}$ into $G_{L,q}$. In this case the PSS morphisms intertwine this module structure with the module structure on the quotient Floer homologies. Moreover, if G_M, G_L are kernels of surjective local system morphisms onto constant systems F_M, F_L where the following diagram commutes:

$$\begin{array}{c} \pi_2^0(M) \longrightarrow \pi_2^0(M,L) \\ \downarrow & \downarrow \\ F_M \longrightarrow F_L \end{array}$$

then by the above $QH^{G_M}_*(M)$ inherits the structure of a $\mathbb{Z}[F_M]$ -module, $QH^{G_L}_*(L)$ the structure of a $\mathbb{Z}[F_L]$ -module and the induced structures of a $\mathbb{Z}[F_M]$ -module on $QH^{G_L}_*(L)$ coincide.

Finally we note that quotient complexes can likewise be endowed with twisted coefficients. In this case the twisting local system naturally lives on the quotient space $\tilde{\Omega}_L/G$. Details are left to the reader.

2.6. Examples. Here we present a few examples of the canonical complex and some quotient and twisted complexes for various Lagrangians. Details of their computations can be found in [Zap15, Section 8].

Throughout R is the ground ring, and J_0 denotes the standard integrable complex structure on the corresponding symplectic manifold.

2.6.1. $\mathbb{R}P^n$ in $\mathbb{C}P^n$ for $n \geq 2$. The Lagrangian real projective space $\mathbb{R}P^n \subset \mathbb{C}P^n$ is the fixed point set of the antisymplectic involution given by complex conjugation. It has minimal Maslov number n + 1.

We take the quantum datum $\mathcal{D} = (f, \rho, J_0)$, where ρ is the round metric on $\mathbb{R}P^n$ induced from $S^n \subset S^{2n+1} \subset \mathbb{C}^{n+1}$, f a perfect self-indexing Morse function with exactly n+1 critical points q_0, \ldots, q_n, q_i being of index i, with exactly two gradient trajectories from q_i to q_{i-1} for each i. This example is interesting because $\mathbb{R}P^n$ is not orientable for even n, and for $n \equiv 1 \mod 4$ it is not relatively Pin^{\pm} . However assumption (**O**) is satisfied. Indeed, for $n \geq 3$ we have $\pi_2(\mathbb{R}P^n) = 0$, while $\mathbb{R}P^2$ is Pin^- , since $w_2 + w_1^2 = 0$, therefore by Remark 1 it is satisfied as well. Thus the canonical quantum complex can be defined, and it is fairly easily computed. As a module we have

$$QC_*(\mathbb{R}P^n) \simeq R$$

for every $* \in \mathbb{Z}$, whereas the boundary operator is as follows:

$$..QC_{n-2j} \xrightarrow{0} QC_{n-2j-1} \xrightarrow{\pm 2} ...$$

for every $j \in \mathbb{Z}$. Here the sign depends on a basis chosen to identify QC_* with R in each degree. We see that the canonical quantum homology in this case is

$$QH_*(\mathbb{R}P^n) \simeq \begin{cases} R/2R, & n-* \text{ is even} \\ \ker 2, & n-* \text{ is odd} \end{cases}$$

where 2: $R \to R$ is multiplication by 2. For instance if 2 is invertible in R, we have $QH_*(\mathbb{R}P^n) \equiv 0$. At the other extreme, if 2 = 0 in R, we have $QH_*(\mathbb{R}P^n) = R$ for every *. If we take $R = \mathbb{Z}$, then $QH_*(\mathbb{R}P^n) \simeq \mathbb{Z}_2$ for even n - * and 0 for odd n - *. Note that this is well-defined even if $\mathbb{R}P^n$ is not orientable!

There is another interesting phenomenon for even n. Note that $\pi_2(M,L) \simeq$ \mathbb{Z} , the isomorphism being given by $\mu/(n+1)$: $\pi_2(M,L) \to \mathbb{Z}$. It follows that $\pi_2^0(M,L) = 0$ and therefore no interesting quotient complexes can be formed if we want to keep a Z-grading. Note that the usual Novikov ring used in Lagrangian Floer or quantum homology in this case is $R[t, t^{-1}]$ with t of degree -(n+1), corresponding to generator of the action of the groupoid $\pi_2(M,L)$ on $\Omega_{\mathbb{R}P^n}$ given by attaching a Maslov n+1 disk. We see from the above that the quantum homology of $\mathbb{R}P^n$, in case 2 is not invertible, has period 2 with respect to the degree. On the other hand the Novikov variable t has odddegree -(n+1). It follows that the full Novikov ring $R[t, t^{-1}]$ cannot act on $QH_*(\mathbb{R}P^n)$ in this case. This of course has to do with the fact that $\mathbb{R}P^n$ is not orientable, that is $w_1(\mathbb{R}P^n) \neq 0$, and that a generator of $H^1(\mathbb{R}P^n;\mathbb{Z}_2)$ is given by the boundary of a Maslov n + 1 disk. Compare this with the discussion of quotient complexes above in §2.5.3, where it is indicated that in the nonorientable case only the "orientable part" of the full Novikov ring can act, the orientable part in this case being $R[t^2, t^{-2}] \subset R[t, t^{-1}]$.

2.6.2. The Clifford torus $L \subset \mathbb{C}P^2$. The Clifford torus is the monotone Lagrangian torus

$$L = \{ [z_0 : z_1 : z_2] | |z_0| = |z_1| = |z_2| \} \subset \mathbb{C}P^2.$$

It has minimal Maslov $N_L = 2$. We identify L with $\mathbb{T}^2 = \mathbb{R}^2/\mathbb{Z}^2$ by means of the map

$$\mathbb{R}^2/\mathbb{Z}^2 \to L$$
, $[\theta_1, \theta_2] \mapsto [1:e^{2\pi i \theta_1}:e^{2\pi i \theta_2}]$

Abbreviate $M = \mathbb{C}P^2$. The local system $\pi_2(M, L)$ is trivial. At the point [1:1:1] it is freely generated, as an abelian group, by the classes of the disks

 $D^2 \rightarrow M\,, \quad z \mapsto \left[1:z:1\right], \quad z \mapsto \left[1:1:z\right], \quad z \mapsto \left[z:1:1\right],$

and we let A, B, C be the corresponding classes in $\pi_2(M, L)$, which we view as global trivializing sections of this local system. We pick the quantum datum (f, ρ, J_0) where ρ is the standard flat metric on $L \simeq \mathbb{R}^2/\mathbb{Z}^2$, f is a Morse function with 4 critical points q_2, x, y, q_0 , where q_i have index i while x, y have index 1, and where the gradient trajectories from q_2 to x, y and from x, y to q_0 are slight deformations of pieces of the coordinate lines.

Consider $\pi_2(M, L)$ as a group. It is the free abelian group on the generators A, B, C. Let $R[\pi_2(M, L)]$ be the corresponding graded group ring, which is the ring of Laurent polynomials in three variables of degree -2, where we here denote e^A, e^B, e^C , that is $R[\pi_2(M, L)]$ consists of finite sums of the form $\sum_{i,j,k\in\mathbb{Z}} r_{ijk}e^{iA+jB+kC}$, $r_{ijk} \in R$.

The canonical complex $QC_*(L)$ can be given the structure of a free module over this group ring $R[\pi_2(M, L)]$; we let the basis elements be denoted by q_2, x, y, q_0 by abuse of notation. We thus have

$$QC_*(L) = R[\pi_2(M,L)] \otimes_R R\langle q_2, x, y, q_0 \rangle$$

as a graded module. The boundary operator is linear over $R[\pi_2(M, L)]$ and is given on generators by

$$\partial x = (-e^B + e^C)q_2, \quad \partial y = (-e^A + e^C)q_2,$$
$$\partial q_0 = (-e^A + e^C)x + (e^B - e^C)y,$$

and of course for the maximum we have $\partial q_2 = 0$.

From this we can compute the canonical quantum homology which turns out to be R in even degrees and 0 in odd degrees. Over the ring $R[\pi_2(M, L)]$ this homology is generated by the class of q_2 , that is [L]. We see that in this case out of all the singular homology classes of L only the fundamental class appears in the quantum homology.

To obtain more interesting results we can form a quotient complex corresponding to the trivial Pin⁺-structure on L and the subsystem $\pi_2^0(M, L)$, which is a trivial local system, and whose group of global sections is the free abelian group generated by the classes A - B, A - C. We note that the quotient local system $\pi_2(M, L)/\pi_2^0(M, L)$ is also trivial and is isomorphic to the trivial system \mathbb{Z} ; we let t be the generator of the corresponding group ring $R[\mathbb{Z}] = R[t, t^{-1}]$. We have the corresponding quotient map of rings

$$R[\pi_2(M,L)] \to R[t,t^{-1}], \quad \sum_{i,j,k \in \mathbb{Z}} r_{ijk} e^{iA+jB+kC} \mapsto \sum_{i,j,k \in \mathbb{Z}} r_{ijk} t^{i+j+k}.$$

The quotient complex as a module is thus given by

$$QC_*(L) = R[t, t^{-1}] \otimes_R R\langle q_2, x, y, q_0 \rangle$$

where we omit the superscript corresponding to $\pi_2^0(M, L)$ to simplify the notation. The induced boundary operator is obtained by substituting t instead of e^A, e^B, e^C in the above, and thus vanishes. We see that in this case the homology is

$$QH_*(L) = R[t, t^{-1}] \otimes_R R\langle q_2, x, y, q_0 \rangle \simeq H_*(L; R[t, t^{-1}]),$$

that is the quantum homology is isomorphic to the singular homology with coefficients in a suitable Novikov ring, which is the answer we are used to in Floer theory. Another variation is to use twisted coefficients on this quotient complex. We take the twisted coefficients corresponding to a local system on L with R as the fiber at every point and monodromy $\sigma, \tau \in R^{\times}$ around the loops given by the coordinates θ_1, θ_2 . The complex as a module over $R[t, t^{-1}]$ is unchanged, while the boundary operator is obtained by the substitution

$$e^A \mapsto \sigma \cdot t \,, \quad e^B \mapsto \tau \cdot t \,, \quad e^C \mapsto \sigma^{-1} \tau^{-1} \cdot t \,,$$

therefore we have

$$\partial x = t(-\tau + \sigma^{-1}\tau^{-1})q_2, \quad \partial y = t(-\sigma + \sigma^{-1}\tau^{-1})q_2,$$
$$\partial q_0 = t(-\sigma + \sigma^{-1}\tau^{-1})x + t(\tau - \sigma^{-1}\tau^{-1})y.$$

Let us compute for which local systems this boundary operator vanishes. In case R is a field, this is equivalent to the homology being nonzero, since otherwise q_2 is a boundary. We have the equations

$$\tau^2 = \sigma^{-1}, \quad \sigma^2 = \tau^{-1},$$

that is $\sigma = \tau$ and $\tau^3 = 1$, that is $\sigma = \tau$ must be a cubic root of 1. These correspond to the three components of the monotone Fukaya category of $\mathbb{C}P^2$.

2.6.3. The Chekanov torus in $\mathbb{C}P^2$. The Chekanov torus $L \subset \mathbb{C}P^2$ is obtained as follows: we view $\mathbb{C}P^2$ as the symplectic cut of the unit codisk bundle of the standard round $\mathbb{R}P^2$ by the geodesic flow on the boundary, which then gives rise to a conic in $\mathbb{C}P^2$; if we take a cotangent circle over a point of $\mathbb{R}P^2$ and flow it with the geodesic flow, it traces a Lagrangian torus, and the Chekanov torus is the unique monotone one obtained in this way. We let θ_1 be the coordinate on the cotangent circle and θ_2 the coordinate given by the time of the geodesic flow.

The local system $\pi_2(M, L)$ is again trivial, and its group of global sections is the free abelian group generated by the classes h, α, β , where h is the class of the complex line, α is the class corresponding to contracting a cotangent circle within its cotangent disk, while β is the class of a disk contracting a simple closed geodesic on L, which has intersection number 2 with the conic. We have

$$\mu(h) = 6$$
, $\mu(\alpha) = 0$, $\mu(\beta) = 2$.

We pick a quantum datum (f, ρ, J_0) , where ρ is the flat metric corresponding to the coordinates θ_1, θ_2 and f is a Morse function with 4 critical points q_2, x, y, q_0 of indices 2, 1, 1, 0, respectively, such that the isolated gradient lines are slight deformations of segments of coordinate circles.

Viewing again $\pi_2(M, L)$ as a group, we have the corresponding graded group ring $R[\pi_2(M, L)]$, consisting of finite sums $\sum_{i,j,k\in\mathbb{Z}} r_{ijk}e^{ih+j\alpha+k\beta}$, and where the grading is given by the Maslov index. The canonical complex can be given the structure of a free module over this ring, and denoting a basis by q_2, x, y, q_0 by abuse of notation, we have

$$QC_*(L) = R[\pi_2(M, L)] \otimes_R R\langle q_2, x, y, q_0 \rangle.$$

The boundary operator is given in this basis by

$$\partial x = (-e^{\beta} + 2(e^{h-2\beta-\alpha} + 2e^{h-2\beta} + e^{h-2\beta+\alpha}))q_2, \ \partial y = (e^{h-2\beta-\alpha} - e^{h-2\beta+\alpha})q_2, \\ \partial q_0 = (e^{h-2\beta-\alpha} - e^{h-2\beta+\alpha})x + (e^{\beta} - 2(e^{h-2\beta-\alpha} + 2e^{h-2\beta} + e^{h-2\beta+\alpha}))y.$$

We will not compute the homology of this canonical complex. Instead we will compute it for some quotient complexes and twisted coefficients.

First note that if R has characteristic 2, then q_2 is the boundary of $-e^{-\beta}x$ and therefore the homology vanishes. This still holds for any quotient complex and any choice of twisted coefficients.

We can form the quotient complex corresponding to the trivial Pin⁺structure and the subsystem $\pi_2^0(M, L)$, which is freely generated by α and $h - 3\beta$. The quotient group ring is $R[t, t^{-1}]$ with t of degree -2 and the quotient map being given by $e^h \mapsto t^3, e^\alpha \mapsto 1, e^\beta \mapsto t$. The quotient complex as a module is

$$QC_*(L) = R[t, t^{-1}] \otimes_R R\langle q_2, x, y, q_0 \rangle$$
.

The quotient boundary operator is

$$\partial x = 7tq_2, \ \partial y = 0, \ \partial q_0 = -7ty.$$

Thus the homology vanishes if 7 is invertible in R, for instance if R is a field of characteristic $\neq 7$.

In order to obtain nonzero homology in case R is a field, we use twisted coefficients corresponding to a local system on L with R as the fiber at every point and monodromy around the coordinate loops corresponding to θ_1, θ_2 being $\sigma, \tau \in R^{\times}$, respectively. The corresponding boundary operator is obtained by substituting

$$e^h \mapsto t^3$$
, $e^\alpha \mapsto \sigma$, $e^\beta \mapsto t \cdot \tau$,

therefore

$$\partial x = t(-\tau + 2\tau^{-2}(\sigma^{-1} + 2 + \sigma))q_2, \ \partial y = t\tau^{-2}(\sigma^{-1} - \sigma)q_2,$$
$$\partial q_0 = (t\tau^{-2}(\sigma^{-1} - \sigma)x + t(\tau - 2\tau^{-2}(\sigma^{-1} + 2 + \sigma))y.$$

Let us determine for which values of σ, τ this boundary operator vanishes. We have the equations

$$\sigma^2 = 1$$
, $\tau^3 = 2(\sigma + \sigma^{-1})^2$,

which have solutions $\sigma = 1$ and τ is a cubic root of 8.

2.6.4. The torus $L_{S^2 \times S^2}$ in $S^2 \times S^2$. The torus $L = L_{S^2 \times S^2}$ can be viewed in an analogous way to the Chekanov torus, namely, $M = S^2 \times S^2$ can be viewed as the symplectic cut of the cotangent disk bundle of S^2 by the geodesic flow on the boundary, where the zero section becomes the antidiagonal $\overline{\Delta} \subset$ $S^2 \times S^2$ while the boundary maps to the symplectic diagonal $\Delta \subset S^2 \times S^2$. Taking a cotangent circle over a point in S^2 and flowing it with the geodesic flow yields a Lagrangian torus, and there is a unique radius of the initial cotangent circle for which the resulting torus is monotone in M — this is L. We put coordinates θ_1, θ_2 on L, where θ_1 is the coordinate of the cotangent circle while θ_2 is the coordinate corresponding to the time of the geodesic flow.

The local system $\pi_2(M, L)$ is trivial and is freely generated as an abelian group by the classes $A = [S^2 \times \text{pt}], B = [\text{pt} \times S^2], \alpha$ which is the class of the cotangent disk contracting the initial cotangent circle, and finally β which is the class of a disk contracting a closed geodesic lying on the torus L and whose intersection number with Δ is 2. We have

$$\mu(A) = \mu(B) = 4$$
, $\mu(\alpha) = 0$, $\mu(\beta) = 2$.

We pick a quantum datum (f, ρ, J_0) where ρ is the flat metric and f is a Morse function with 4 critical points q_2, x, y, q_0 of indices 2, 1, 1, 0, respectively, and where the gradient trajectories are slight deformations of segments of coordinate circles.

Viewing $\pi_2(M, L)$ as a group, its group ring is $R[\pi_2(M, L)]$ with the same notation as above. It is graded by the Maslov index. The canonical complex has the structure of a free module over this ring with generators denoted by abuse of notation q_2, x, y, q_0 , thus

$$QC_*(L) = R[\pi_2(M, L)] \otimes_R R\langle q_2, x, y, q_0 \rangle.$$

The boundary operator relative to this basis is

$$\partial x = (-e^{\beta} + e^{A-\beta} + e^{B-\beta} + e^{A-\beta-\alpha} + e^{B-\beta+\alpha})q_2, \ \partial y = (e^{A-\beta-\alpha} - e^{B-\beta+\alpha})q_2$$
$$\partial q_0 = (e^{A-\beta-\alpha} - e^{B-\beta+\alpha})x + (e^{\beta} - (e^{A-\beta} + e^{B-\beta} + e^{A-\beta-\alpha} + e^{B-\beta+\alpha}))y.$$

Again, we will not compute the homology of this canonical complex, but rather we will pass to the quotient complex corresponding to the subsystem $\pi_2^0(M, L)$ and the trivial Pin⁺-structure on L. This quotient complex is

$$QC_*(L) = R[t, t^{-1}] \otimes_R R\langle q_2, x, y, q_0 \rangle$$

with the boundary operator obtained from the substitution $e^A, e^B \mapsto t^2$, $e^{\alpha} \mapsto 1, e^{\beta} \mapsto t$:

$$\partial x = 3tq_2, \ \partial y = 0, \ \partial q_0 = -3ty.$$

Thus the homology is nonzero whenever 3 is not invertible.

We can also twist this quotient complex by a local system on L with R as the fiber at every point and monodromy around the basic loops corresponding to θ_1, θ_2 being $\sigma, \tau \in R^{\times}$. The boundary operator is obtained by substituting $e^A, e^B \mapsto t^2, e^{\alpha} \mapsto \sigma, e^{\beta} \mapsto t \cdot \tau$:

$$\partial x = t(-\tau + \tau^{-1}(\sigma^{-1} + 2 + \sigma))q_2, \ \partial y = t\tau(\sigma^{-1} - \sigma)q_2,$$
$$\partial q_0 = t\tau(\sigma^{-1} - \sigma)x + t(\tau - \tau^{-1}(\sigma^{-1} + 2 + \sigma))y.$$

If we wish the boundary operator to vanish, we obtain the equations

$$\sigma^2 = 1, \quad \tau^2 = (\sigma + \sigma^{-1})^2,$$

whose solutions are $\sigma = 1$ and $\tau = \pm 2$.

2.7. **Duality.** We first describe duality in Lagrangian Floer homology. To this end we need to introduce Lagrangian Floer cohomology. We construct everything over \mathbb{Z} , but it is straightforward to adapt this to any coefficient ring. Let (H, J) be a regular Floer datum for L. We let the corresponding Lagrangian Floer cochain complex be

$$CF^*(H, J:L) = \bigoplus_{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}} C(\widetilde{\gamma})^{\vee}$$

where for a \mathbb{Z} -module Q we let $Q^{\vee} = \operatorname{Hom}_{\mathbb{Z}}(Q, \mathbb{Z})$ be the dual module. Note that if Q is free and of rank 1, there is a natural isomorphism $Q^{\vee} = Q$ given by sending a generator q^{\vee} of Q^{\vee} to the generator q of Q with $q^{\vee}(q) = 1$.

The differential $\partial_{H,J}^{\vee}$ is defined as the operator whose matrix element

$$C(\widetilde{\gamma}_+)^{\vee} \to C(\widetilde{\gamma}_-)^{\vee}$$

is dual to the matrix element $C(\tilde{\gamma}_{-}) \to C(\tilde{\gamma}_{+})$ of $\partial_{H,J}$. We let the differential

$$\delta_{H,J}: CF^k(H,J:L) \to CF^{k+1}(H,J:L)$$

be defined as $\delta_{H,J} = (-1)^{k-1} \partial_{H,J}^{\vee}$.

Clearly we have $\delta_{H,J}^2 = 0$ and we let $HF^*(H, J : L)$ be the cohomology of the resulting cochain complex $(CF^*(H, J : L), \delta_{H,J})$.

Similarly to the above, we can define the twisted Floer cochain complex $CF^*(H:L;\mathcal{L})$ and the corresponding twisted cohomology $HF^*(H,J:L;\mathcal{L})$.

For the remainder of this section we let \mathcal{L} be the \mathbb{Z} -bundle over Ω_L with fiber $|\operatorname{d}(T_{\gamma(1)}L)|$ over $\widetilde{\gamma} \in \widetilde{\Omega}_L$, where for a real line V we let |V| be the free \mathbb{Z} -module of rank 1 whose two generators are the two orientations of V.

We now define the dual datum $(\overline{H}, \overline{J})$ as follows:

$$\overline{H}_t(x) = -H_{1-t}(x), \quad \overline{J}_t(x) = J_{1-t}(x).$$

It is also regular. There is a bijection

$$\operatorname{Crit} \mathcal{A}_{H:L} \leftrightarrow \operatorname{Crit} \mathcal{A}_{\overline{H}:L}, \quad \widetilde{\gamma} = [\gamma, \widehat{\gamma}] \mapsto \overline{\widetilde{\gamma}} = [\overline{\gamma}, \overline{\widetilde{\gamma}}],$$

where $\overline{\gamma}(t) = \gamma(1-t)$ and $\overline{\widehat{\gamma}}(\sigma,\tau) = \widehat{\gamma}(\sigma,-\tau)$. We have

$$\mathcal{A}_{H:L}(\widetilde{\gamma}) = -\mathcal{A}_{\overline{H}:L}(\overline{\widetilde{\gamma}}) \text{ and } m_{H:L}(\widetilde{\gamma}) = n - m_{\overline{H}:L}(\overline{\widetilde{\gamma}}).$$

Likewise there are natural diffeomorphisms

$$\widetilde{\mathcal{M}}(H,J;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+})\leftrightarrow\widetilde{\mathcal{M}}(\overline{H},\overline{J};\overline{\widetilde{\gamma}}_{+},\overline{\widetilde{\gamma}}_{-})\,,\quad u\mapsto\overline{u}(s,t)=u(-s,1-t)\,.$$

It is proved in [Zap15, Section 3.11] that for any $\tilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}$ there is a natural isomorphism

(7)
$$C(\overline{\widetilde{\gamma}}) = [C(\widetilde{\gamma}) \otimes \mathcal{L}_{\widetilde{\gamma}}]^{\vee},$$

leading to graded module isomorphisms

(8)
$$CF_*(\overline{H}:L) = CF^{n-*}(H,L;\mathcal{L}).$$
Ibid., it is proved that the following diagram commutes for every $\widetilde{\gamma}_{\pm} \in$ Crit $\mathcal{A}_{H:L}$ of index difference 1 and any $u \in \widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+})$:

$$C(\overline{\widetilde{\gamma}}_{+}) = [C(\widetilde{\gamma}_{+}) \otimes \mathcal{L}_{\widetilde{\gamma}_{+}}]^{\vee}$$

$$\downarrow^{C(\overline{u})} \qquad \qquad \downarrow^{(-1)^{n-m_{H:L}(\widetilde{\gamma}_{-})}(C(u) \otimes \mathcal{P}_{u})^{\vee}}$$

$$C(\overline{\widetilde{\gamma}}_{-}) = [C(\widetilde{\gamma}_{-}) \otimes \mathcal{L}_{\widetilde{\gamma}_{-}}]^{\vee}$$

It follows that the identification of modules (8) is in fact an identification of complexes, leading to a canonical isomorphism

$$HF_*(\overline{H}, \overline{J}:L) = HF^{n-*}(H, J:L;\mathcal{L}),$$

which we refer to as **the duality isomorphism**. Continuation isomorphisms are intertwined by this identification and therefore we have a canonical isomorphism of abstract homologies $HF_*(L) = HF^{n-*}(L; \mathcal{L})$.

REMARK 28. Roughly, the reason for twisting by the determinant bundle of L in the formulation of duality is as follows: if $[\gamma, \hat{\gamma}] \in \operatorname{Crit} \mathcal{A}_{H:L}$ and $[\overline{\gamma}, \overline{\hat{\gamma}}] \in \operatorname{Crit} \mathcal{A}_{\overline{H}:L}$ are critical points corresponding to one another by the above bijection, gluing the linearized operators $D_{\widehat{\gamma}}$ and $D_{\overline{\widehat{\gamma}}}$ (composed with the conformal isomorphism $z \mapsto -z$) yields an operator on the disk, which can be deformed into the Dolbeault operator $\overline{\partial}$ on the trivial bundle pair $(T_{\gamma(1)}M, T_{\gamma(1)}L) \to (D^2, S^1)$. Therefore we have a canonical isomorphism

$$d(D_{\widetilde{\gamma}}) \otimes d(D_{\overline{\widetilde{\gamma}}}) \simeq d(\partial) \equiv d(T_{\gamma(1)}L),$$

leading to the identification (7). A similar argument shows the relation with the (co)boundary operators. This appearance of the determinant bundle is analogous to Poincaré duality on (not necessarily orientable) manifolds. An orientation of L gives a trivialization of the \mathbb{Z} -bundle \mathcal{L} , which consequently can be omitted.

Duality in periodic orbit Floer homology is a similarly defined canonical identification $HF_*(M) = HF^{2n-*}(M)$. Here there is no need to twist by the orientation bundle of M because it carries the canonical orientation coming from the symplectic form.

We also have duality isomorphisms in quantum homologies of L and M, which take the form of canonical identifications

(9) $QH_*(L) = QH^{n-*}(L; \mathcal{L})$ and $QH_*(M) = QH^{2n-*}(M)$.

The PSS maps can be defined between quantum and Floer cohomologies and homologies with twisted coefficients, and they intertwine these canonical identifications.

Finally we note that there are obvious degree-preserving homomorphisms

$$HF^*(H, J:L) \to HF_*(H, J:L)^{\vee} \quad \text{and} \quad QH^*(L) \to QH_*(L)^{\vee}$$

coming from the duality pairing; similar homomorphisms exist for the twisted versions. This observation, together with the duality isomorphism above

leads to the following Poincaré duality pairing:

(10)
$$\langle \cdot, \cdot \rangle \colon QH_*(L) \otimes QH_{n-*}(L; \mathcal{L}) \to \mathbb{Z}.$$

2.8. The Lagrangian diagonal in $(M \times M, \omega \oplus (-\omega))$. Recall that the diagonal $\Delta \subset M \times M$ is a Lagrangian submanifold. It is monotone if and only if M is a monotone symplectic manifold and, in this case, the minimal Maslov number of Δ coincides with twice the minimal Chern number of M. In this subsection we establish a relation between Lagrangian quantum and Floer homology of Δ and the quantum and Floer homology of M.

Let (H, J) be a time-periodic regular Floer datum on M. We define a Floer datum (\hat{H}, \hat{J}) for $\Delta \subset M \times M$ as follows. First define

$$H_t^1 = H_t$$
, $H_t^2 = H_{1-t}$, $J_t^1 = J_t$, $J_t^2 = -J_{1-t}$ for $t \in [0, \frac{1}{2}]$,

and put

$$\widehat{H}_t(x,y) = H_t^1(x) + H_t^2(y), \quad \widehat{J}_t(x,y) = J_t^1(x) \oplus J_t^2(y).$$

Then $(\widehat{H}, \widehat{J})$ is a regular Floer datum¹⁴ for the diagonal Δ .

We claim that there is a canonical isomorphism of chain complexes

(11)
$$(CF_*(H), \partial_{H,J}) = (CF_*(\widehat{H} : \Delta), \partial_{\widehat{H},\widehat{J}}),$$

preserving the grading and the actions of the generators.

We start with the Hamiltonian orbits. A periodic orbit γ of H on M gives rise to a Hamiltonian arc $\Gamma: [0, \frac{1}{2}] \to M \times M$ of \widehat{H} with endpoints on Δ , namely

$$\Gamma(t) = (\gamma_1(t), \gamma_2(t))$$
 where $\gamma_1(t) = \gamma(t), \quad \gamma_2(t) = \gamma(1-t).$

This defines a bijection between the periodic orbits of H and arcs of \hat{H} with endpoints on Δ .

If $\hat{\gamma}$ is a capping of an orbit γ of H, we can define a capping $\hat{\Gamma}$ of Γ , as follows. The capping $\hat{\gamma}: \dot{S}^2 \to M$ is defined on the punctured Riemann sphere $\dot{S}^2 = \hat{\mathbb{C}} \setminus \{1\}$ where $\hat{\mathbb{C}} = \mathbb{C} \cup \{\infty\}$. There is a conformal embedding $\mathbb{R} \times \mathbb{R}/\mathbb{Z} \to \dot{S}^2$ given by $z = s + it \mapsto \frac{e^{2\pi z} - i}{e^{2\pi z} + i}$; it is onto $S^2 \setminus \{\pm 1\}$. Relative to the coordinates (s, t) induced by this embedding we can define the maps

$$\widehat{\gamma}_1(s,t) = \widehat{\gamma}(s,t), \quad \widehat{\gamma}_2(s,t) = \widehat{\gamma}(s,1-t) \text{ for } t \in [0,\frac{1}{2}].$$

The same conformal embedding restricted to $\mathbb{R} \times [0, \frac{1}{2}]$ is onto the punctured disk $D^2 \setminus \{\pm 1\}$. Composing the inverse of this embedding with the maps $\widehat{\gamma}_{1,2}$ we get maps $\widehat{\gamma}_{1,2}$: $\dot{D}^2 \to M$, where we abuse notation slightly. Putting

$$\widehat{\Gamma}(z) = \left(\widehat{\gamma}_1(z), \widehat{\gamma}_2(z)\right)$$

¹⁴Note that this datum is defined for $t \in [0, \frac{1}{2}]$. We leave it to the reader to make the necessary adjustments of the Floer theories.

we see that $\widehat{\Gamma}$ is asymptotic to Γ at the puncture and that it maps the boundary to Δ . The map thus defined

$$\operatorname{Crit} \mathcal{A}_H \to \operatorname{Crit} \mathcal{A}_{\widehat{H}:\Delta}, \quad \widetilde{\gamma} = [\gamma, \widehat{\gamma}] \mapsto \widetilde{\Gamma} = [\Gamma, \widehat{\Gamma}]$$

is a bijection, which moreover preserves the action values and the grading. Recall that

$$CF_*(H) = \bigoplus_{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_H} C(\widetilde{\gamma}) \text{ and } CF_*(\widehat{H} : \Delta) = \bigoplus_{\widetilde{\Gamma} \in \operatorname{Crit} \mathcal{A}_{\widehat{H} \cdot \Delta}} C(\widetilde{\Gamma}).$$

We will now establish a canonical isomorphism $C(\tilde{\gamma}) = C(\tilde{\Gamma})$ where $\tilde{\gamma}, \tilde{\Gamma}$ correspond to each other by the above bijection. Consider a capping $\hat{\gamma}$ of γ and the corresponding Fredholm operator

$$D_{\widehat{\gamma}} \colon W^{1,p}(\dot{S}^2, \widehat{\gamma}^*TM) \to L^p(\dot{S}^2, \Omega^{0,1} \otimes \widehat{\gamma}^*TM).$$

We have seen that $\hat{\gamma}$ gives rise to two maps $\hat{\gamma}_{1,2}$: $\dot{D}^2 \to M$. Similarly, restricting sections of the bundles $\hat{\gamma}^*TM$ and $\Omega^{0,1} \otimes \hat{\gamma}^*TM$ to the disks produces pairs of sections which agree on the boundary. This gives rise to natural isomorphisms

$$W^{1,p}(\dot{S}^2, \hat{\gamma}^*TM) \simeq W^{1,p}(\dot{D}^2, \partial \dot{D}^2; \hat{\Gamma}^*T(M \times M), (\hat{\Gamma}|_{\partial \dot{D}^2})^*T\Delta)$$
$$L^p(\dot{S}^2, \Omega^{0,1} \otimes \hat{\gamma}^*TM) \simeq L^p(\dot{D}^2, \Omega^{0,1} \otimes \hat{\Gamma}^*T(M \times M)).$$

These isomorphisms intertwine the operators $D_{\widehat{\gamma}}$ and $D_{\widehat{\Gamma}}$, thereby inducing a Fredholm isomorphism between them (see [Zap15, Section 2]). It follows that the determinant lines of these two operators are canonically isomorphic, which means that we have obtained the desired canonical isomorphism $C(\widetilde{\gamma}) = C(\widetilde{\Gamma})$, and consequently the isomorphism (11) as modules.

We now must show that this module isomorphism is a chain isomorphism. Given a Floer cylinder $u: \mathbb{R} \times S^1 \to M$ of (H, J) asymptotic to orbits $\tilde{\gamma}_{\pm} = \lim_{s \to \pm \infty} u(s, \cdot)$ we can define two maps $u_i: \mathbb{R} \times [0, \frac{1}{2}] \to M$, i = 1, 2, via

$$u_1(s,t) = u(s,t), \quad u_2(s,t) = u(s,1-t).$$

Putting $U(z) = (u_1(z), u_2(z))$, we obtain a map $U: \mathbb{R} \times [0, \frac{1}{2}] \to M \times M$ sending the boundary to Δ . It can be checked that $\partial_{\widehat{H},\widehat{J}}U = 0$ and that Uis asymptotic to Γ_{\pm} . This defines a map

$$\widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+}) \to \widetilde{\mathcal{M}}(\widehat{H}, \widehat{J}; \widetilde{\Gamma}_{-}, \widetilde{\Gamma}_{+}).$$

It can be seen that this map is a bijection. Indeed, given a Floer strip U for (\hat{H}, \hat{J}) , we can define a continuous map $u: \mathbb{R} \times [0, 1] \to M$ by gluing the two components of U; the resulting map solves the Floer PDE for (H, J) away from the seams. It can also be seen that u is in fact C^1 along the seams. It then follows that it is smooth by elliptic regularity.

It is then straightforward to show that the following diagram commutes:

$$\begin{array}{c} C(\widetilde{\gamma}_{-}) \xrightarrow{C(u)} C(\widetilde{\gamma}_{+}) \\ \| & \| \\ C(\widetilde{\Gamma}_{-}) \xrightarrow{C(U)} C(\widetilde{\Gamma}_{+}) \end{array}$$

using the gluing and deformation isomorphisms. This shows that the isomorphism (11) is indeed an isomorphism of chain complexes.

It can also be shown that the above procedure can be adapted to produce a canonical isomorphism $QH_*(M) = QH_*(\Delta)$. Moreover, the PSS maps on both sides are respected by these isomorphisms, in particular we have the following commutative diagram:

$$QH_*(M) = QH_*(\Delta)$$

$$\downarrow PSS \qquad \qquad \downarrow PSS$$

$$HF_*(H,J) = HF_*(\widehat{H}, \widehat{J} : \Delta)$$

2.9. Lagrangian Floer and quantum homology of products. Here we establish a relation between the Floer and quantum homologies of a pair of Lagrangians L_1, L_2 with those of their product.

Let (M_i, ω_i) be symplectic manifolds and $L_i \subset M_i$ Lagrangian submanifolds, i = 1, 2, so that the product $L_1 \times L_2$ is a monotone Lagrangian of minimal Maslov number at least 2. This implies that both L_1 and L_2 are monotone and have minimal Maslov numbers at least 2. Pick regular Floer data $(H^i, J^i)_i$ for L_i . We claim that there is a canonical chain isomorphism

(12)
$$CF_*(H^1:L_1) \otimes CF_*(H^2:L_2) = CF_*(H^1 \oplus H^2:L_1 \times L_2).$$

It is clear that there is a natural bijection

$$\operatorname{Crit} \mathcal{A}_{H^1:L_1} \times \operatorname{Crit} \mathcal{A}_{H^2:L_2} = \operatorname{Crit} \mathcal{A}_{H^1 \oplus H^2:L_1 \times L_2}$$

preserving actions and the grading. We first need to establish a canonical isomorphism $C(\tilde{\gamma}_1) \otimes C(\tilde{\gamma}_2) = C(\tilde{\gamma})$ when $\tilde{\gamma}_i \in \operatorname{Crit} \mathcal{A}_{H^i:L_i}$ and $\tilde{\gamma}$ corresponds to $\tilde{\gamma}_1 \times \tilde{\gamma}_2$ via the bijection. This is done as follows. There is an obvious isomorphism of vector bundles

$$\widehat{\gamma}^*T(M_1 \times M_2) = \widehat{\gamma}_1^*TM_1 \oplus \widehat{\gamma}_2^*TM_2$$

which induces an isomorphism of the corresponding Sobolev spaces of sections, thereby producing a Fredholm isomorphism

$$D_{\widehat{\gamma}} = D_{\widehat{\gamma}_1} \oplus D_{\widehat{\gamma}_2}$$

which implies, using the direct sum isomorphism, that

$$\mathrm{d}(D_{\widehat{\gamma}}) = \mathrm{d}(D_{\widehat{\gamma}_1}) \otimes \mathrm{d}(D_{\widehat{\gamma}_2}),$$

therefore that we have the desired isomorphism $C(\tilde{\gamma}) = C(\tilde{\gamma}_1) \otimes C(\tilde{\gamma}_2)$.

We also claim that $(H^1 \oplus H^2, J^1 \oplus J^2)$ is a regular Floer datum for $L_1 \times L_2$. Moreover, if $u: \mathbb{R} \times [0, 1] \to M_1 \times M_2$ is a solution of the Floer PDE for this datum, and it has index 1, and we let $u_i: \mathbb{R} \times [0, 1] \to M_i$ be its components, then since $D_u = D_{u_1} \oplus D_{u_2}$ and all the operators are surjective, it follows that precisely one of the operators D_{u_i} is an isomorphism, while the other has index 1. Therefore Floer trajectories of $(H^1 \oplus H^2, J^1 \oplus J^2)$ of index 1 have the form $u = (u_1, u_2)$ where one of the u_i is *s*-independent, while the other is an index 1 Floer trajectory for its datum. This, together with the Koszul signs arising from the direct sum isomorphisms as in [Zap15, Section 3.7], implies that the boundary operator $\partial_{H^1 \oplus H^2, J^1 \oplus J^2}$ is indeed the tensor product of the operators ∂_{H^i, J^i} , in the graded sense. This shows that the above isomorphism of modules (12) is a chain isomorphism, and so we have a canonical map of Floer homologies

$$HF_*(H^1, J^1: L_1) \otimes HF_*(H^2, J^2: L_2) \to HF_*(H^1 \oplus H^2, J^1 \oplus J^2: L_1 \times L_2),$$

which is injective in case the ground ring R is a PID, and an isomorphism if it is a field.

We similarly have a map of quantum homologies

$$QH_*(L_1) \otimes QH_*(L_2) \rightarrow QH_*(L_1 \times L_2)$$
,

and the PSS isomorphisms are respected by these, so that we have a commutative diagram

$$\begin{array}{c|c} QH_*(L_1) \otimes QH_*(L_2) & \longrightarrow QH_*(L_1 \times L_2) \\ & & & & \downarrow \text{PSS} \\ & & & & \downarrow \text{PSS} \\ HF_*(H^1, J^1:L_1) \otimes HF_*(H^2, J^2:L_2) & \longrightarrow HF_*(H^1 \oplus H^2, J^1 \oplus J^2:L_1 \times L_2) \end{array}$$

2.10. Action of the symplectomorphism group. If $\psi \in \text{Symp}(M, \omega)$, let $L' = \psi(L)$. In this subsection we construct a canonical chain isomorphism

$$\psi_*: \left(CF_*(H:L), \partial_{H,J} \right) \to \left(CF_*(H^{\psi}:L'), \partial_{H^{\psi},J^{\psi}} \right),$$

where (H, J) is a regular Floer datum for L, (H^{ψ}, J^{ψ}) is the Floer datum $H^{\psi} = H \circ \psi^{-1}, J^{\psi} = \psi_* J$, which is regular for L'. Likewise, we construct a chain isomorphism

$$\psi \colon QC_*(\mathcal{D}:L) \to QC_*(\mathcal{D}^{\psi}:L')$$

of quantum complexes where $\mathcal{D} = (f, \rho, I)$ is a regular quantum datum for Land $\mathcal{D}^{\psi} = (f^{\psi}, \rho^{\psi}, I^{\psi})$ is the quantum datum with $f^{\psi} = f \circ \psi^{-1}, \ \rho^{\psi} = \psi_* \rho, I^{\psi} = \psi_* I$, which is regular for L'.

First, note that there is an obvious bijection

$$\psi_*\colon \operatorname{Crit} \mathcal{A}_{H:L} \to \operatorname{Crit} \mathcal{A}_{H^{\psi}:L'}$$

given by $\tilde{\gamma} = [\gamma, \hat{\gamma}] \mapsto \tilde{\gamma}^{\psi} = [\gamma^{\psi} = \psi \circ \gamma, \hat{\gamma}^{\psi} = \psi \circ \hat{\gamma}]$. It clearly preserves the actions and Conley–Zehnder indices. Then there is a canonical isomorphism

of Z-modules, $C(\tilde{\gamma}) \simeq C(\tilde{\gamma}^{\psi})$, constructed in the following manner. The symplectomorphism ψ induces a bundle pair isomorphism

$$\psi_*: \left(\widehat{\gamma}^*TM, (\widehat{\gamma}|_{\partial \dot{D}^2})^*TL\right) \to \left((\widehat{\gamma}^{\psi})^*TM, (\widehat{\gamma}^{\psi}|_{\partial \dot{D}^2})^*TL\right)$$

which in turn induces the commutative diagram of Fredholm operators and Banach isomorphisms

It follows that the operators $D_{\widehat{\gamma}}$, $D_{\widehat{\gamma}^{\psi}}$ are canonically isomorphic and therefore so are their determinant bundles, whose orientations are the generators of $C(\widehat{\gamma})$, $C(\widehat{\gamma}^{\psi})$. The ensuing isomorphism of \mathbb{Z} -modules

$$CF_*(H:L) \simeq CF_*(H^{\psi}:L')$$

is a chain isomorphism. Indeed, let $\widetilde{\gamma}_{\pm} \in \operatorname{Crit} \mathcal{A}_{H:L}$ and define

$$\widetilde{\mathcal{M}}(H,J;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+})\to\widetilde{\mathcal{M}}(H^{\psi},J^{\psi};\widetilde{\gamma}_{-}^{\psi},\widetilde{\gamma}_{+}^{\psi})\,,\quad u\mapsto u^{\psi}=\psi\circ u\,.$$

This map is easily shown to be a diffeomorphism, which also preserves the canonical orientations in case $\tilde{\gamma}_{\pm}$ have index difference 1. It is clear that in the latter case the following diagram commutes:

which implies our claim.

The isomorphism in quantum homology is similarly constructed; the details are left to the reader. It is also obvious that ψ_* intertwines the PSS isomorphisms on homology, giving the commutative diagram

$$HF_*(H, J:L) \xrightarrow{\psi_*} HF_*(H^{\psi}, J^{\psi}:L')$$

$$\downarrow^{\text{PSS}} \qquad \qquad \qquad \downarrow^{\text{PSS}}$$

$$QH_*(L) \xrightarrow{\psi_*} QH_*(L')$$

3. LAGRANGIAN SPECTRAL INVARIANTS: DEFINITION

Recall that in order to be able to define Lagrangian quantum and Floer homology of L, we need to fix a ground ring R, a local subsystem $G < \pi_2^0(M, L)$, and in case G is nontrivial and R has characteristic different from 2, a relative Pin[±] structure, in order to be able to form quotient complexes as described in §2.5.3. We fix these choices once and for all and omit them from notation throughout §§3-4.

In this section, we define Lagrangian spectral invariants for continuous time-dependent Hamiltonians on M, as well as for elements of $\operatorname{Ham}(M, \omega)$. This is done using the following steps, each one performed in its own subsection.

§3.1: To any regular Floer datum (H, J) for L we associate a function

 $\ell(\cdot; H, J): QH_*(L) \to \mathbb{R} \cup \{-\infty\}.$

§3.2: We show that for any non-zero $\alpha \in QH_*(L)$ and any regular Floer data (H^i, J^i) for L, i = 0, 1 we have

$$\int_{0}^{1} \min_{M} (H^{1} - H^{0}) dt \le \ell(\alpha; H^{1}, J^{1}) - \ell(\alpha; H^{0}, J^{0}) \le \int_{0}^{1} \max_{M} (H^{1} - H^{0}) dt$$

§3.3: It follows that $\ell(\alpha; H, J)$ only depends on H, and we let the common value be denoted by $\ell(\alpha; H)$. Also it follows that ℓ can be uniquely extended to a function

$$\ell: QH_*(L) \times C^0(M \times [0,1]) \to \mathbb{R} \cup \{-\infty\}.$$

§3.4: Finally we show that ℓ only depends on the homotopy class of the path $\{\phi_H^t\}_{t\in[0,1]}$ with fixed endpoints, provided H is normalized, so that ℓ descends to a function

$$\ell: QH_*(L) \times \widetilde{\operatorname{Ham}}(M, \omega) \to \mathbb{R} \cup \{-\infty\}.$$

3.1. Nondegenerate Hamiltonians. Pick a regular Floer datum (H, J) for L and recall that in particular H is supposed to be nondegenerate. There exists a canonical PSS isomorphism (6)

$$PSS^{H,J}: QH_*(L) \to HF_*(H,J:L).$$

The Floer complex can be naturally filtered by the action. Namely, for $a \in \mathbb{R}$ define

$$CF^a_*(H:L) = \bigoplus_{\substack{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L} \\ \mathcal{A}_{H:L}(\widetilde{\gamma}) < a}} C(\widetilde{\gamma}) \,.$$

If the matrix element $C(\widetilde{\gamma}_{-}) \to C(\widetilde{\gamma}_{+})$ of the boundary operator $\partial_{H,J}$ is non-zero, then there exists a Floer trajectory $u \in \widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+})$. Therefore, since the Floer equation is the negative gradient equation for $\mathcal{A}_{H:L}$ (see Remark 21), we obtain

$$\mathcal{A}_{H:L}(\widetilde{\gamma}_{-}) - \mathcal{A}_{H:L}(\widetilde{\gamma}_{+}) > 0,$$

because u, which connects critical points of $\mathcal{A}_{H:L}$ of index difference one, necessarily depends on s. Thus the action decreases along Floer trajectories, the boundary operator preserves the above filtration, and $CF^a_*(H:L) \subset$ $CF_*(H:L)$ is a subcomplex. We denote by $HF^a_*(H,J:L)$ the homology of $(CF^a_*(H:L), \partial_{H,J})$ and let

$$i^a_*: HF^a_*(H, J:L) \to HF_*(H, J:L)$$

be the map induced on homology by the inclusion. The Lagrangian spectral invariant corresponding to a non-zero class $\alpha \in QH_*(L)$ is then defined as

(14)
$$\ell(\alpha; H, J) = \inf \left\{ a \in \mathbb{R} : \mathrm{PSS}^{H, J}(\alpha) \in \mathrm{im}(i^a_*) \right\}.$$

Finally, recall from §2.3.1 that there exists a canonical class $[L] \neq 0$ in $QH_*(L)$. The spectral invariant associated to this specific class will be of particular interest to us and we will denote it

(15)
$$\ell_+(H,J) = \ell([L];H,J).$$

We will also need spectral invariants coming from cohomology. We refer the reader to §2.7 for relevant definitions. We note that

$$CF_a^*(H:L) := CF^*(H:L) / \bigoplus_{\substack{\widetilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L} \\ \mathcal{A}_{\widetilde{\gamma}} > a}} C(\widetilde{\gamma})^{\vee}$$

is a quotient cochain complex. We let $HF_a^*(H, J : L)$ be the corresponding cohomology and we denote by i_a^* : $HF^*(H, J : L) \to HF_a^*(H, J : L)$ the map induced on cohomology by the quotient map $CF^*(H : L) \to CF_a^*(H : L)$. Then for $\alpha^{\vee} \in QH^*(L)$ we can define

(16)
$$\ell(\alpha^{\vee}; H) = \sup \left\{ a \in \mathbb{R} \mid \mathrm{PSS}^{H, J}(\alpha^{\vee}) \in \ker i_a^* \right\},$$

where $\text{PSS}^{H,J}: QH^*(L) \to HF^*(H,J:L)$ is the PSS isomorphism on cohomology.

Finally we will use Hamiltonian spectral invariants. If (H, J) is a timeperiodic regular Floer datum on M, the Floer complex $CF_*(H)$ can similarly be filtered by action to yield a subcomplex $CF_*^a(H)$ generated by critical points of action $\langle a$. Letting i_*^a be the map induced on homology by the inclusion $CF_*^a(H) \hookrightarrow CF_*(H)$, we define for any non-zero class $b \in QH_*(M)$

$$c(b; H) = \inf \left\{ a \in \mathbb{R} \mid \mathsf{PSS}^{H, J}(b) \in \operatorname{im} i^a_* \right\}.$$

3.2. Continuity property of ℓ . We consider regular Floer data (H^i, J^i) , i = 0, 1. In this section we prove the property (13) of the invariants defined by (14). In order to do so, it is enough to prove that, given $\epsilon > 0$, we can pick a regular homotopy of Floer data $(H^s, J^s)_{s \in \mathbb{R}}$ between (H^0, J^0) and (H^1, J^1) , which is stationary for $s \notin (0, 1)$, such that for any $a \in \mathbb{R}$, the corresponding continuation morphism

$$\Phi_{(H^s, J^s)_s} : CF_*(H^0 : L) \to CF_*(H^1 : L)$$

maps $CF^a_*(H^0:L)$ into $CF^{a+b}_*(H^1:L)$ where $b = \int_0^1 \max_M(H^1_t - H^0_t) dt + \epsilon$. Indeed, in this case we obtain the following commutative diagram

$$\begin{split} HF^a_*(H^0,J^0:L) & \xrightarrow{i^a_*} HF_*(H^0,J^0:L) < \xrightarrow{\mathrm{PSS}^{H^0,J^0}} QH_*(L) \\ & \downarrow^{\Phi} & \downarrow^{\Phi} \\ HF^{a+b}_*(H^1,J^1:L) \xrightarrow{i^{a+b}_*} HF_*(H^1,J^1:L) \end{split}$$

which implies

$$\ell(\alpha; H^1, J^1) - \ell(\alpha; H^0, J^0) \le \int_0^1 \max_M (H_t^1 - H_t^0) \, dt + \epsilon \, .$$

Since this is true for any $\epsilon > 0$, we obtain the right inequality in (13). The other inequality is obtained by exchanging the roles of H^0 and H^1 .

To prove that there is such a regular homotopy of Floer data, consider the special homotopy of Hamiltonians given by

$$K_t^s(x) = H_t^0(x) + \beta(s) \left(H_t^1(x) - H_t^0(x) \right),$$

where $\beta \colon \mathbb{R} \to [0, 1]$ is a smooth nondecreasing function which satisfies $\beta(s) = 0$ for $s \leq 0$ and $\beta(s) = 1$ for $s \geq 1$. There is a regular homotopy of Floer data $(H^s, J^s)_s$, stationary for $s \notin (0, 1)$, such that

$$\max_{(x,t)\in M\times[0,1]}\left(\frac{\partial H^s_t}{\partial s}(x)-\frac{\partial K^s_t}{\partial s}(x)\right)\leq\epsilon\,.$$

Let $\tilde{\gamma}_i \in \operatorname{Crit} \mathcal{A}_{H^i:L}$ have the same index and let $v \in \mathcal{M}((H^s, J^s)_s; \tilde{\gamma}_0, \tilde{\gamma}_1)$. Recall that v satisfies the parametrized Floer equation $\overline{\partial}_{(H^s, J^s)_s} u = 0$, see (4). There is a natural lift of v to a path in $\tilde{\Omega}_L$ running from $\tilde{\gamma}_0$ to $\tilde{\gamma}_1$; by abuse of notation we denote this path by $v: \mathbb{R} \to \tilde{\Omega}_L$. The parametrized Floer equation can equivalently be written as $v'(s) = -\nabla \mathcal{A}_{H^s:L}(v(s))$. We therefore have

$$\begin{split} \mathcal{A}_{H^{1}:L}(\widetilde{\gamma}_{1}) - \mathcal{A}_{H^{0}:L}(\widetilde{\gamma}_{0}) &= \int_{-\infty}^{\infty} \frac{d}{ds} \mathcal{A}_{H^{s}:L}\big(v(s)\big) \, ds \\ &= \int_{-\infty}^{\infty} \left(d_{v(s)} \mathcal{A}_{H^{s}:L}\big(v'(s)\big) + \frac{\partial \mathcal{A}_{H^{s}:L}}{\partial s}\big(v(s)\big) \right) \, ds \\ &= -\int_{-\infty}^{\infty} \|\nabla_{v(s)} \mathcal{A}_{H^{s}:L}\|^{2} \, ds \\ &\qquad + \int_{\mathbb{R} \times [0,1]} \frac{\partial H_{t}^{s}}{\partial s}\big(v(s,t)\big) \, ds \, dt \\ &\leq \int_{\mathbb{R} \times [0,1]} \frac{\partial H_{t}^{s}}{\partial s}\big(v(s,t)\big) \, ds \, dt \, . \end{split}$$

The last term equals

$$\int_{\mathbb{R}\times[0,1]} \frac{\partial K_t^s}{\partial s} \big(v(s,t) \big) \, ds \, dt + \int_{\mathbb{R}\times[0,1]} \left(\frac{\partial H_t^s}{\partial s} \big(v(s,t) \big) - \frac{\partial K_t^s}{\partial s} \big(v(s,t) \big) \right) \, ds \, dt \, .$$

Here in the first integral we have

$$\int_{\mathbb{R}\times[0,1]} \frac{\partial K_t^s}{\partial s} (v(s,t)) \, ds \, dt = \int_{\mathbb{R}\times[0,1]} \beta'(s) (H_t^1 - H_t^0) \circ v \, ds \, dt$$
$$\leq \int_0^1 \max_M (H_t^1 - H_t^0) \, dt \int_{-\infty}^\infty \beta'(s) \, ds$$
$$= \int_0^1 \max_M (H_t^1 - H_t^0) \, dt \, .$$

By the choice of H^s we have

$$\int_{\mathbb{R}\times[0,1]} \left(\frac{\partial H^s_t}{\partial s} - \frac{\partial K^s_t}{\partial s}\right) \circ v \, ds \, dt = \int_{[0,1]\times[0,1]} \left(\frac{\partial H^s_t}{\partial s} - \frac{\partial K^s_t}{\partial s}\right) \circ v \, ds \, dt \le \epsilon \, .$$

All these estimates together imply

$$\mathcal{A}_{H^{1}:L}(\widetilde{\gamma}_{1}) - \mathcal{A}_{H^{0}:L}(\widetilde{\gamma}_{0}) \leq \int_{0}^{1} \max_{M} (H_{t}^{1} - H_{t}^{0}) dt + \epsilon \,,$$

which means that the homotopy of Floer data $(H^s, J^s)_s$ has the property claimed above.

3.3. Invariance of ℓ (part 1) and arbitrary Hamiltonians. First, by putting $H^0 = H^1$ in (13), we immediately deduce that $\ell(\alpha; H, J)$ does not depend on the choice of the almost complex structure J. We will therefore denote the resulting common value by $\ell(\alpha; H)$ from now on. Similarly, we use the notation $\ell_+(H)$ for $\ell_+(H, J)$.

Again, by (13), we can define the spectral numbers for an arbitrary continuous function $H: M \times [0,1] \to \mathbb{R}$, as follows. Pick a sequence of smooth nondegenerate Hamiltonians $\{H_n\}_n$ satisfying

$$\lim_{n \to \infty} \|H_n - H\|_{C^0(M \times [0,1])} = 0$$

and define $\ell(\alpha; H)$ as the limit $\lim_{n\to\infty} \ell(\alpha; H_n)$. Equation (13) ensures that this limit exists and does not depend on the choice of the sequence $\{H_n\}_n$.

3.4. Invariance of ℓ (part 2). We now show that ℓ only depends on homotopy classes of Hamiltonian isotopies with fixed endpoints, thereby proving Proposition 4. Recall that we call a Hamiltonian H normalized if for all $t \in [0, 1], \int_M H_t \omega^n = 0.$

Definition 29. Two normalized Hamiltonians H^0 and H^1 are called equivalent if there is a homotopy $(H^s)_{s \in [0,1]}$ between them so that for all s, H_s is normalized and $\phi_{H_s}^1 = \phi$.

The universal cover of $\operatorname{Ham}(M, \omega)$ coincides with the set of equivalence classes of normalized Hamiltonians.

REMARK 30. Before proving the invariance property of ℓ , recall that when dealing with spectral invariants of the Hamiltonian H_t in question, we may assume that it vanishes for t near 0, 1, see for instance [Pol01, MVZ12]. From now on all Hamiltonians are assumed to have this property.

For a smooth Hamiltonian H we let $\text{Spec}(H : L) = \mathcal{A}_{H:L}(\text{Crit }\mathcal{A}_{H:L})$ be its **action spectrum**. The following can be proved similarly to the Hamiltonian case (see [Oh05a]):

Lemma 31. The action spectrum $\text{Spec}(H : L) \subset \mathbb{R}$ is a closed nowhere dense subset.

The proof of the fact that ℓ only depends on the equivalence class of a normalized Hamiltonian boils down to the following lemma.

Lemma 32. Let H be normalized. Then Spec(H : L) only depends on the equivalence class of H.

We denote $\operatorname{Spec}(\widetilde{\phi}:L) = \operatorname{Spec}(H:L)$ for any normalized representative H of $\widetilde{\phi} \in \operatorname{Ham}(M, \omega)$ and call it the action spectrum of $\widetilde{\phi}$.

First let us show that the lemma implies the desired invariance property of ℓ . Indeed, if $(H^s)_s$ is a normalized homotopy so that for all $s \in [0, 1]$, $\phi_{H^s}^1 = \phi$, by Lemma 32 the spectrum of $\mathcal{A}_{H^s:L}$ does not depend on s. This, combined with the fact that the spectrum is nowhere dense (Lemma 31) and with the SPECTRALITY and CONTINUITY properties of spectral invariants (see Theorem 36 below), shows that for every non-zero $\alpha \in QH_*(L)$, $\ell(\alpha; H^0) = \ell(\alpha; H^1)$.

Proof of Lemma 32. Let $(H^s)_{s \in [0,1]}$ be a normalized homotopy so that $\phi_{H^s}^1 = \phi$ for all s. We need to show that $\operatorname{Spec}(H^0 : L) = \operatorname{Spec}(H^1 : L)$. Let us define a map

$$\operatorname{Crit} \mathcal{A}_{H^0:L} \to \operatorname{Crit} \mathcal{A}_{H^1:L}$$

as follows. Let $\tilde{\gamma}_0 = [\gamma_0, \hat{\gamma}_0] \in \operatorname{Crit} \mathcal{A}_{H^0:L}$ and let $u(s,t) = \phi_{H^s}^t(\gamma_0(0))$ for $(s,t) \in [0,1]^2$. Define the glued map $\hat{\gamma}_1 = \hat{\gamma}_0 \sharp u$, which we view as a capping of the Hamiltonian arc $\gamma_1 = u(1, \cdot)$ of H^1 . The above bijection then maps $\tilde{\gamma}_0$ to $\tilde{\gamma}_1 = [\gamma_1, \hat{\gamma}_1]$ by definition.

The proof will be complete once we show that $\mathcal{A}_{H^0:L}(\widetilde{\gamma}_0) = \mathcal{A}_{H^1:L}(\widetilde{\gamma}_1)$. This is done in two steps.

Step 1. First we show that

(17)
$$\mathcal{A}_{H^1:L}(\widetilde{\gamma}_1) - \mathcal{A}_{H^0:L}(\widetilde{\gamma}_0) = \int_{I^2} (\partial_s H^s_t) \circ u \, ds \, dt$$

with I = [0, 1]. Let $\gamma_s = u(s, \cdot)$, $\widehat{\gamma}_s = \widehat{\gamma}_0 \sharp u|_{[0,s] \times [0,1]}$, and $\widetilde{\gamma}_s = [\gamma_s, \widehat{\gamma}_s]$. We have

$$\begin{aligned} \mathcal{A}_{H^{1}:L}(\widetilde{\gamma}_{1}) - \mathcal{A}_{H^{0}:L}(\widetilde{\gamma}_{0}) &= \int_{0}^{1} \frac{d}{ds} \mathcal{A}_{H^{s}:L}(\widetilde{\gamma}_{s}) \, ds \\ &= \int_{0}^{1} d_{\widetilde{\gamma}_{s}} \mathcal{A}_{H^{s}:L}(\partial_{s}\widetilde{\gamma}_{s}) \, ds + \int_{I^{2}} (\partial_{s}H^{s}_{t}) \circ u \, ds \, dt \end{aligned}$$

Since by construction $\tilde{\gamma}_s$ is a critical point of $\mathcal{A}_{H^s:L}$, the first summand in the last expression vanishes, and we have proved (17).

Step 2. We prove that

(18)
$$\int_{I^2} (\partial_s H^s_t) \left(\phi^t_{H^s}(p) \right) ds \, dt$$

does not depend on p.

Consider the time-periodic Hamiltonian $K^s = H^s \not \exists \overline{H^0}$. For $p \in M$ let $u_p(s,t) = \phi_{H^s}^t(p)$. Define $\delta_s^p = u_p(s,\cdot) \not \exists \overline{u_p(0,\cdot)}$. Then δ_s^p is a periodic orbit of K^s and the map $\widehat{\delta}_s^p = u_p|_{[0,s]\times[0,1]}$ can be viewed as a capping of δ_s^p . We let $\widetilde{\delta}_s^p = [\delta_s^p, \widehat{\delta}_s^p]$ and observe that this is a critical point of \mathcal{A}_{K^s} . Noting that $\mathcal{A}_{K^0}(\widetilde{\delta}_0^p) = 0$, we have:

$$\begin{aligned} \mathcal{A}_{K^1}(\widetilde{\delta}_1^p) &= \int_0^1 \frac{d}{ds} \mathcal{A}_{K^s}(\widetilde{\delta}_s^p) \, ds \\ &= \int_0^1 d_{\widetilde{\delta}_s^p} \mathcal{A}_{K^s}(\partial_s \widetilde{\delta}_s^p) \, ds + \int_{I^2} (\partial_s H_t^s) \circ u_p \, ds \, dt \end{aligned}$$

Since $\tilde{\delta}_s^p$ is a critical point of \mathcal{A}_{K^s} , this first summand in the last expression vanishes, and so we obtain that the sought-for integral (18) in fact equals the value of the action functional \mathcal{A}_{K^1} at the critical point $\tilde{\delta}_1^p$.

Note that $M \ni p \mapsto \widetilde{\delta}_1^p$ is a smooth embedding of M into the set of critical points of \mathcal{A}_{K^1} . Since a functional is constant on a connected submanifold of the set of its critical points, we conclude that our integral (18) is indeed independent of p.

End of the proof. Using the normalization condition, the fact that $\phi_{H^s}^t$ is a symplectomorphism for all s, t, and Steps 1, 2, we obtain finally:

$$\begin{split} 0 &= \int_0^1 dt \int_M \left(H_t^1(p) - H_t^0(p) \right) \omega_p^n \\ &= \int_{I^2} \left(\int_M \partial_s H_t^s(p) \, \omega_p^n \right) ds \, dt \\ &= \int_{I^2} \left(\int_M \partial_s H_t^s(\phi_{H^s}^t(p)) \, \omega_p^n \right) ds \, dt \\ &= \int_{I^2} \partial_s H_t^s(\phi_{H^s}^t(p)) \, ds \, dt \cdot \int_M \omega^n \\ &= \left(\mathcal{A}_{H^1:L}(\widetilde{\gamma}_1) - \mathcal{A}_{H^0:L}(\widetilde{\gamma}_0) \right) \int_M \omega^n \,, \end{split}$$

thereby finishing the proof of Lemma 32.

This concludes the proof of invariance. In view of this property, we have actually defined

$$\ell: QH_*(L) \times \widetilde{\operatorname{Ham}}(M, \omega) \to \mathbb{R} \cup \{-\infty\}$$

with $\ell(\alpha; \tilde{\phi}) = \ell(\alpha; H)$ for any normalized Hamiltonian H representing $\tilde{\phi}$. We denote $\ell_+(\tilde{\phi}) = \ell([L]; \tilde{\phi})$.

4. Spectral invariants: main properties

We maintain the choices made at the beginning of §3.

4.1. Quantum valuation. Here we introduce the valuation in Lagrangian quantum homology. It is analogous to the valuation introduced in [EP03], and serves a similar purpose.

Fix a regular quantum datum \mathcal{D} for L. The quantum valuation

$$\nu: QH_*(\mathcal{D}:L) \to \mathbb{Z} \cup \{-\infty\}$$

is defined as follows. If $\alpha \in QH_*(\mathcal{D}:L)$ is non-zero, let $C \in QC_*(\mathcal{D}:L)$ be a chain representing it. We define the valuation of C to be

 $\nu(C) = \tfrac{1}{\mathsf{A}} \max\left\{ -\omega(A) \, | \, \text{the component of } C \text{ in } C(q,A) \text{ is not } 0 \right\}.$

Then we define

$$\nu(\alpha) = \inf\{\nu(C) \mid [C] = \alpha\}$$

and we put $\nu(0) = -\infty$. Proposition 35 below will show that $\nu \cdot A$ equals the spectral invariant of the zero Hamiltonian, which implies that ν in fact descends to a well-defined function

$$\nu: QH_*(L) \to \mathbb{Z} \cup \{-\infty\}.$$

REMARK 33. When $N_L = \infty$, it follows from the definitions that $\nu(\alpha) = 0$ for any nonzero α .

Lemma 34. If $\alpha \neq 0$, then $\nu(\alpha) \in \mathbb{Z}$.

Proof. Just like in [EP03], we can show that $\nu(\alpha)$ is the maximum of $\nu(\alpha_k)$ where $\alpha_k \in QH_k(\mathcal{D}:L)$ is the degree-k component of α . Therefore it suffices to show that $\nu(\alpha)$ is finite for homogeneous α . In this case, due to the monotonicity of L, the direct summands C(q, A) of $QC_k(\mathcal{D}:L)$ have the property that there are only finitely many values that $-\omega(A)/A$ can attain. Therefore any nonzero cycle $C \in QC_k(\mathcal{D}:L)$ has valuation $\nu(C)$ which is at least the minimal such value. It follows that $\nu(\alpha)$ is also at least this minimal value.

There is a more natural way to define ν in the context of this paper. Define the function

$$\mathcal{A}_0$$
: $\{(q, A) | q \in \operatorname{Crit} f, A \in \pi_2(M, L, q)\} \to \mathbb{R} \quad \text{by} \quad \mathcal{A}_0(q, A) = -\omega(A).$

Morally speaking, since quantum homology is the Morse–Bott Floer homology of the zero Hamiltonian, this is the corresponding action functional, evaluated on the set of critical points of the auxiliary Morse function on the critical manifold.

Using \mathcal{A}_0 we can now define a subcomplex $QC^a_*(\mathcal{D}:L)$ spanned by those generators of $QC_*(\mathcal{D}:L)$ with $\mathcal{A}_0 < a$. This is indeed a subcomplex since it follows from the definition of the quantum boundary operator that if its matrix element $C(q, A) \to C(q', A')$ is non-zero, then necessarily $\omega(A') \geq \omega(A)$. We let $QH^a_*(\mathcal{D}:L)$ be the homology of this subcomplex and let

$$i_{\Omega}^{a}: QH^{a}_{*}(\mathcal{D}:L) \to QH_{*}(\mathcal{D}:L)$$

be the map induced by the inclusion $QC^a_*(\mathcal{D}:L) \hookrightarrow QC_*(\mathcal{D}:L)$. It is then obvious from the definitions that

$$\nu(\alpha) = \frac{1}{\mathsf{A}} \inf \left\{ a \in \mathbb{R} \, | \, \alpha \in \operatorname{im}(i_Q^a) \right\}.$$

Intuitively, this means that $\nu(\alpha)A$ is the spectral invariant of the zero Hamiltonian. We now show that this analogy is in fact precise.

Proposition 35. We have

$$\int_0^1 \min_M H_t \, dt \le \ell(\alpha; H) - \nu(\alpha) \mathsf{A} \le \int_0^1 \max_M H_t \, dt \, .$$

In particular $\nu(\alpha)A = \ell(\alpha; 0)$ and therefore ν is independent of the quantum datum \mathcal{D} used to define it.

Proof. Fix a regular Floer datum (H, J). We claim that for any given $\epsilon > 0$ we can choose a suitable perturbation datum needed in the definition of the PSS morphism

$$QC_*(\mathcal{D}:L) \to CF_*(H:L)$$

so that for any $a \in \mathbb{R}$ it restricts to a map

$$QC^a_*(\mathcal{D}:L) \to CF^{a+b}_*(H:L)$$

where $b = \int_0^1 \max_M H_t dt + \epsilon$. Let us first show how this implies the right inequality of the proposition. It follows that there is a commutative diagram

$$QH^{a}_{*}(\mathcal{D}:L) \longrightarrow HF^{a+b}_{*}(H,J:L)$$

$$\downarrow^{i^{a}_{Q}} \qquad \qquad \downarrow^{i^{a+b}_{*}}$$

$$QH_{*}(\mathcal{D}:L) \xrightarrow{\mathrm{PSS}^{H,J}_{\mathcal{D}}} HF_{*}(H,J:L)$$

The definition of spectral invariants and the valuation then imply that

$$\ell(\alpha; H) \leq \mathsf{A} \cdot \nu(\alpha) + \int_0^1 \max_M H_t \, dt + \epsilon$$

Since this holds for any ϵ , we obtain the right inequality above. The other inequality is similarly proved using the inverse PSS morphism.

To prove the assertion, we need to use the definition of the PSS isomorphism described in [Zap15, Section 5]. We do not need the full-detail version of the definition, but rather enough of it in order to prove action estimates. We fix $\epsilon > 0$.

Let $q \in \operatorname{Crit} f$, $A \in \pi_2(M, L, q)$, $\tilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}$, and assume that the matrix element $C(q, A) \to C(\tilde{\gamma})$ of the PSS morphism does not vanish. This means that there are maps

$$u_i: (D^2, S^1) \to (M, L), \quad i = 1, \dots, k, \text{ and } u_0: (\dot{D}^2, \partial \dot{D}^2) \to (M, L),$$

with the following properties. Each u_i for i > 0 is *I*-holomorphic, there is a gradient trajectory of f running from q to $u_1(-1)$, and gradient trajectories of f running from $u_i(1)$ to $u_{i+1}(-1)$ for i < k, and from $u_k(1)$ to $u_0(-1)$. The map u_0 satisfies the Floer PDE on \dot{D}^2 where the perturbation datum is chosen as follows. There is a conformal identification $\mathbb{R} \times [0,1] \simeq \dot{D}^2 \setminus \{-1\}$ given by the same formula as (2) and with respect to the coordinates (s,t) induced on $\dot{D}^2 \setminus \{-1\}$ via this identification, the perturbation datum takes the form of a homotopy of Floer data $(H^s, J^s)_{s \in \mathbb{R}}$, which is stationary for $s \notin (0,1)$, and such that $(H^s, J^s) = (0,I)$ for $s \leq 0$, $(H^s, J^s) = (H,J)$ for $s \geq 1$, and where moreover there exists a nondecreasing smooth function $\beta \colon \mathbb{R} \to [0,1]$ with $\beta(s) = 0$ for $s \leq 0$ and $\beta(s) = 1$ for $s \geq 1$ such that

(19)
$$\max_{(x,t)\in M\times[0,1]} \left(\frac{\partial_s H_t^s}{\partial s}(x) - \beta'(s)H_t(x)\right) \le \epsilon.$$

Moreover, by definition the capping $\widehat{\gamma}$ in $\widetilde{\gamma}$ has the property that $\widehat{\gamma} \sim A \sharp u_1 \sharp \dots \sharp u_k \sharp u_0$ (~ being the equivalence relation on the set of cappings,

see \$2.2.1), where the concatenation is performed along the pieces of gradient trajectories of f mentioned above. We have

$$\begin{aligned} \mathcal{A}_{H:L}(\widetilde{\gamma}) &= \int_0^1 H_t(\gamma(t)) \, dt - \int \widehat{\gamma}^* \omega \\ &= \int_0^1 H_t(\gamma(t)) \, dt - \omega(A) - \sum_{i=1}^k \int u_i^* \omega - \int u_0^* \omega \\ &\leq -\omega(A) + \int_0^1 H_t(\gamma(t)) \, dt - \int u_0^* \omega \,, \end{aligned}$$

since the u_i are holomorphic. We claim that

$$\int_0^1 H_t(\gamma(t)) dt - \int u_0^* \omega \leq \int_0^1 \max_M H_t dt + \epsilon.$$

This follows formally using the same argument as in §3.2, as we will now demonstrate. Observe that u_0 , written in the coordinates (s, t) on $\dot{D}^2 \setminus \{-1\}$, satisfies the parametrized Floer equation for the homotopy of Floer data $(H^s, J^s)_s$. The point $u_0(-1)$ can be considered as a Hamiltonian arc of the zero Hamiltonian, and we can cap it using the constant map at the same point. Since the homotopy $(H^s)_s$ satisfies the condition (19), we have, using the argument in §3.2:

$$\int_0^1 H_t(\gamma(t)) dt - \int u_0^* \omega = \mathcal{A}_{H:L}([\gamma, u_0]) - \mathcal{A}_{0:L}([u_0(-1), u_0(-1)])$$
$$\leq \int_0^1 \max_M (H_t - 0) + \epsilon \,,$$

which is precisely what we wanted to show.

4.2. Spectral invariants of continuous Hamiltonians. In this section we prove the following theorem, which generalizes Theorem 3 of the introduction and includes Proposition 5.

Theorem 36. Let L be a closed monotone Lagrangian of (M, ω) with minimal Maslov number $N_L \geq 2$. The function

$$\ell: QH_*(L) \times C^0(M \times [0,1]) \to \mathbb{R} \cup \{-\infty\}$$

constructed in §3 satisfies the following properties.

FINITENESS: $\ell(\alpha; H) = -\infty$ if and only if $\alpha = 0$.

SPECTRALITY: For $H \in C^{\infty}(M \times [0,1])$ and $\alpha \neq 0$, $\ell(\alpha; H) \in \text{Spec}(H:L)$.

GROUND RING ACTION: For $r \in R$, $\ell(r \cdot \alpha; H) \leq \ell(\alpha; H)$. In particular, if r is invertible, then $\ell(r \cdot \alpha; H) = \ell(\alpha; H)$.

SYMPLECTIC INVARIANCE: Let $\psi \in \text{Symp}(M, \omega)$ and $L' = \psi(L)$. Let

$$\ell' \colon QH_*(L') \times C^0(M \times [0,1]) \to \mathbb{R} \cup \{-\infty\}$$

be the corresponding spectral invariant. Then $\ell(\alpha; H) = \ell'(\psi_*(\alpha); H \circ \psi^{-1})$.

NORMALIZATION: If c is a function of time then

$$\ell(\alpha; H+c) = \ell(\alpha; H) + \int_0^1 c(t) dt.$$

We have $\ell(\alpha; 0) = \nu(\alpha) A$ and $\ell_+(0) = 0$.

CONTINUITY: For any H and K, and $\alpha \neq 0$:

$$\int_0^1 \min_M (K_t - H_t) \, dt \le \ell(\alpha; K) - \ell(\alpha; H) \le \int_0^1 \max_M (K_t - H_t) \, dt$$

MONOTONICITY: If $H \leq K$, then $\ell(\alpha; H) \leq \ell(\alpha; K)$.

TRIANGLE INEQUALITY: For all α and β , $\ell(\alpha \star \beta; H \sharp K) \leq \ell(\beta; H) + \ell(\alpha; K)$.

MODULE STRUCTURE: Assume that H^2 is 1-periodic. For all $a \in QH_*(M)$ and $\alpha \in QH_*(L)$, $\ell(a \bullet \alpha; H^1 \sharp H^2) \leq c(a; H^2) + \ell(\alpha; H^1)$.

DUALITY: For $\alpha \in QH_*(L)$ let $\alpha^{\vee} \in QH^{n-*}(L; \mathcal{L})$ be the element corresponding to α under the duality isomorphism (9). Then we have

$$-\ell(\alpha;\overline{H}) = \ell(\alpha^{\vee};H) \le \inf\left\{\ell(\beta;H) \,|\, \beta \in QH_{n-*}(L;\mathcal{L}) : \langle \alpha^{\vee},\beta \rangle \neq 0\right\}.$$

In case the ground ring R is a field and the Floer complexes of nondegenerate Hamiltonians are finite-dimensional in every degree, the last inequality is an equality.

Recall from §2.5.3 that if G is the kernel of a surjective local system morphism $\ker w_1 \to F$, where F is a group, then $QH^G_*(L)$ is a module over the group ring R[F]. We have

NOVIKOV ACTION: For $A \in F$, we have $\ell(A \cdot \alpha; H) = \ell(\alpha; H) - \omega(A)$. LAGRANGIAN CONTROL: If for all t, $H_t|_L = c(t) \in \mathbb{R}$ (respectively \leq, \geq), then

$$\ell(\alpha; H) = \int_0^1 c(t) \, dt + \nu(\alpha) \mathsf{A} \quad (respectively \leq, \geq) \, .$$

So that, for all H:

$$\int_0^1 \min_L H_t \, dt \le \ell(\alpha; H) - \nu(\alpha) \mathsf{A} \le \int_0^1 \max_L H_t \, dt$$

Recall that $\ell([L]; H)$ is denoted $\ell_+(H)$ and that we assume $[L] \neq 0$.

NON-NEGATIVITY: $\ell_+(H) + \ell_+(\overline{H}) \ge 0.$

MAXIMUM: $\ell(\alpha; H) \leq \ell_+(H) + \nu(\alpha) A$.

4.2.1. Proof of Theorem 36. First, notice that the fact that $\ell(\alpha; H) \in \mathbb{R}$ for any $\alpha \neq 0$ and any H easily comes from NORMALIZATION and CONTINU-ITY.¹⁵ Moreover, we set $\ell(0; H) = -\infty$ for all H so that FINITENESS is

¹⁵Alternatively, the finiteness of $\ell(\alpha; H)$ for nondegenerate H can be seen as follows. Just like in the proof of Proposition 35, it is enough to consider the case of homogeneous α . The set of actions of critical points of $\mathcal{A}_{H:L}$ of given Conley–Zehnder index is finite

proven. Notice also that CONTINUITY was proven above for generic smooth Hamiltonians, see §3.2. It obviously extends to continuous functions, and immediately implies MONOTONICITY. Finally, notice that GROUND RING ACTION is obvious by definition.

We now prove the remaining properties, where we only deal with the nontrivial case of non-zero quantum classes.

• SPECTRALITY: The proof is similar to the Hamiltonian case, for instance see [Oh05a]. Recall that $\omega(\pi_2(M, L)) = \mathbb{Z}A$ is a discrete subset of \mathbb{R} . Since we work in a slightly more general situation where we distinguish between homotopy classes of cappings, we include a proof for the sake of completeness.

When H is smooth and nondegenerate, this property follows from the fact that the complement of the spectrum of H is open and dense, since we work in monotone manifolds, and from the fact that the image of the map i_*^a is unchanged as long as a stays in the complement of the spectrum.

When H is degenerate, choose a sequence of nondegenerate Hamiltonians $\{H_n\}_n C^2$ -converging to H. By our definition of spectral invariants, $\ell(\alpha; H) = \lim_{n \to +\infty} \ell(\alpha; H_n)$. By the nondegenerate case, we can choose a sequence $\{(\gamma_n, \widehat{\gamma}_n)\}_n$ so that for all $n, \widetilde{\gamma}_n = [\gamma_n, \widehat{\gamma}_n] \in \operatorname{Crit} \mathcal{A}_{H_n:L}$ and $\ell(\alpha; H_n) = \mathcal{A}_{H_n}(\widetilde{\gamma}_n)$. By C^2 -convergence and compactness of M, the Arzela–Ascoli Theorem ensures the existence of a subsequence of $\{\gamma_n\}_n C^1$ converging to a Hamiltonian chord γ of H.

Fix a Riemannian metric on M with respect to which L is totally geodesic and choose a number ϵ less than its injectivity radius. For n large enough define a strip $u_n: [0,1]^2 \to M$ such that for fixed $t, s \mapsto u_n(s,t)$ is the geodesic arc of length at most ϵ running from $\gamma_n(t)$ to $\gamma(t)$. Similarly we define the strip $u_{n,n'}$ connecting γ_n to $\gamma_{n'}$ for large enough n, n'. Now fix n_0 large enough and define $\hat{\gamma} = \hat{\gamma}_{n_0} \sharp u_{n_0}$. We claim that

$$\mathcal{A}_{H:L}(\widetilde{\gamma} = [\gamma, \widehat{\gamma}]) = \ell(\alpha; H).$$

Note that $\int u_n^* \omega = \int u_{n,n'}^* \omega + \int u_{n,n'}^* \omega$. Indeed, by the Stokes formula and due to the fact that L is Lagrangian, we see that the difference between the two terms equals the integral of $d\omega = 0$ over the set realizing a homotopy between the three strips, and therefore it vanishes. Next, we see that $\int u_n^* \omega \to 0$ as $n \to \infty$, as well as $\int u_{n,n'}^* \omega \to 0$ when $n, n' \to \infty$. Also note that $\int \hat{\gamma}_n^* \omega + \int u_{n,n'}^* \omega = \int \hat{\gamma}_{n'}^* \omega$. Indeed, by construction the difference between the two sides must be the area of an element of $\pi_2(M, L)$, but on the other hand since $\{\int \hat{\gamma}_n^* \omega\}_n$ is Cauchy and by the previous observation, this difference tends to zero, therefore it must be identically zero, owing to the fact that the group of periods of ω is discrete.

due to the monotonicity of L, in particular $\ell(\alpha; H)$ is at least the minimum such action. The assertion follows.

We have

$$\mathcal{A}_{H:L}(\widetilde{\gamma}) - \mathcal{A}_{H_n:L}(\widetilde{\gamma}_n) = \int_0^1 \left(H_t(\gamma(t)) - H_{n,t}(\gamma_n(t)) \right) dt - \int (\widehat{\gamma}^* \omega - \widehat{\gamma}_n^* \omega) dt dt$$

By construction, the first integral tends to zero when $n \to \infty$. In the second integral we have

$$\int (\widehat{\gamma}^* \omega - \widehat{\gamma}^*_n \omega) = \int (\widehat{\gamma}^*_{n_0} \omega - \widehat{\gamma}^*_n \omega) + \int u^*_{n_0} \omega = \int u^*_{n,n_0} \omega + \int u^*_{n_0} \omega = \int u^*_n \omega$$

which tends to zero when $n \to \infty$. Thus we see that

$$\mathcal{A}_{H:L}(\widetilde{\gamma}) = \lim_{n \to \infty} \mathcal{A}_{H_n:L}(\widetilde{\gamma}_n) = \lim_{n \to \infty} \ell(\alpha; H_n) = \ell(\alpha; H).$$

• SYMPLECTIC INVARIANCE First notice that, for $\psi \in \text{Symp}(M, \omega)$, $L' = \psi(L)$ shares the same properties as L so that ℓ' is well-defined. It is enough to prove the assertion for nondegenerate H. Let therefore (H, J) be a regular Floer datum. In §2.10 it is shown that there exists a canonical chain isomorphism

$$\psi_*: \left(CF_*(H:L), \partial_{H,J} \right) \to \left(CF_*(H^{\psi}, J^{\psi}:L'), \partial_{H^{\psi}, J^{\psi}} \right),$$

where $H^{\psi} = H \circ \psi^{-1}$ and $J^{\psi} = \psi_* J$. Since ψ_* preserves the actions of the critical points of the respective action functionals, we obtain, for each $a \in \mathbb{R}$, the commutative diagram, where the vertical arrows are inclusions:

It is also shown in §2.10 that ψ_* induces an isomorphism on homology and an isomorphism $\psi_*: QH_*(L) \to QH_*(L')$. These are featured in the following diagram:

from which it is clear that $\ell(\alpha; H) = \ell'(\psi_*(\alpha); H^{\psi})$, as claimed.

• TRIANGLE INEQUALITY: The proof of this fact is rather standard and has been carried out in several context, see [AS10, MVZ12]. By CONTINUITY it suffices to prove this for smooth nondegenerate H^1 , H^2 . Recall from §2.2.1 that we denote by Σ_{\star} the closed disk with three boundary punctures. Pick almost complex structures J^1 , J^2 so that (H^i, J^i) , i = 1, 2 are regular Floer data for L, and associate them to the negative punctures of Σ_{\star} , so that H^2 comes before H^1 in the cyclic order on ∂D^2 . Pick $\varepsilon > 0$. By Remark 30 we can assume that H^1_t , H^2_t vanish for t close to 0, 1. Let H^3 be a nondegenerate Hamiltonian which satisfies $||H^3 - H^1 \sharp H^2||_{C^0} \leq \varepsilon$, and let J^3 be such that (H^3, J^3) is a regular Floer datum for L, and associate it to the positive puncture of Σ_{\star} . We will now choose a suitable perturbation datum (K, I) on Σ_{\star} . We note that the open set

$$\Upsilon_{\star} = \mathbb{R} \times (0,2) \setminus (-\infty,0] \times \{1\}$$

with the conformal structure induced from its inclusion into $\mathbb{R}^2 = \mathbb{C}$ maps biholomorphically onto the interior of Σ_{\star} , and we can choose this biholomorphism so that the negative (positive) ends of Υ_{\star} correspond to the negative (positive) punctures of Σ_{\star} (see Figure 1 below).

On Υ_{\star} we put the following perturbation datum (K, I). The form K is characterized by $K(\partial_s) = 0$ and

$$K(s,t,x)(\partial_t) = \begin{cases} H^1(t,x) & \text{if } s \le 1, \ t \in (0,1), \\ H^2(t-1,x) & \text{if } s \le 1, \ t \in (1,2), \\ H^3(t,x) & \text{if } s \ge 2, \ t \in (0,2) \end{cases}$$

and $|\partial_s K(\partial_t)| \leq \varepsilon$ for all $t \in (0,2)$ and all $s \in \mathbb{R}$. The family of almost complex structures $I_{(s,t)}$ on M for $s \in \mathbb{R}$ and $t \in (0,2)$ is chosen so that (K, I) is regular and

$$I(s,t,x) = \begin{cases} J^1(t,x) & \text{if } s \le 1, \ t \in (0,1), \\ J^2(t-1,x) & \text{if } s \le 1, \ t \in (1,2), \\ J^3(t,x) & \text{if } s \ge 2, \ t \in (0,2). \end{cases}$$

Now we push this perturbation datum on Υ_{\star} to Σ_{\star} via the aforementioned biholomorphism, and extend the resulting datum on the interior of Σ_{\star} to the whole surface by continuity. Abusing notation we denote this perturbation datum on Σ_{\star} also by (K, I).

Now assume that the matrix element $C(\tilde{\gamma}^2) \otimes C(\tilde{\gamma}^1) \to C(\tilde{\gamma}^3)$ of $\star_{K,I}$ does not vanish, where $\tilde{\gamma}^i \in \operatorname{Crit} \mathcal{A}_{H^i:L}$. Then there is $u \in \mathcal{M}(K,I;{\{\tilde{\gamma}^i\}_i})$. Computing the energy of u in the coordinates (s,t) on Υ_{\star} in a manner similar to the computation in §3.2 above leads to

$$\mathcal{A}_{H^1:L}(\widetilde{\gamma}^1) + \mathcal{A}_{H^2:L}(\widetilde{\gamma}^2) - \mathcal{A}_{H^3:L}(\widetilde{\gamma}^3) \ge -2\varepsilon$$

This indicates that the restriction of $\star_{K,I}$ to the filtered Floer complexes satisfies:

$$\star_{K,I} : CF_*^{a_1}(H^1:L) \otimes CF_*^{a_2}(H^2:L) \to CF_*^{a_1+a_2+2\varepsilon}(H^3:L)$$

for any a_1, a_2 . We can deduce the following inequality on spectral invariants:

$$\ell(\alpha \star \beta; H^3) \le \ell(\beta; H^1) + \ell(\alpha; H^2) + 2\varepsilon.$$

Since ε is arbitrary, we get the expected result

$$\ell(\alpha \star \beta; H^1 \sharp H^2) \le \ell(\beta; H^1) + \ell(\alpha; H^2).$$

• NORMALIZATION: For the first assertion, consider the homotopy H^s defined by $H^s = H + sc$ for $s \in [0, 1]$. For fixed s, $\text{Spec}(H^s : L)$ is Spec(H : L)



FIGURE 1. Main object yielding the definition of product (left) and module (right) structures

translated by $s \int_0^1 c(t) dt$. Thus the result holds for smooth H by SPEC-TRALITY and CONTINUITY. By C^0 -limit, it also holds for any continuous function. The second assertion follows from Proposition 35.

Finally, let us consider the case $\alpha = [L]$. The maximum of a Morse function on L with a single maximum is always a cycle representing [L] (see §2.3.1); this cycle has valuation 0. It follows from the definition of ν that $\nu([L]) \leq 0$, that is $\ell_+(0) = \nu([L]) A \leq 0$. On the other hand, since $[L] \star [L] = [L]$, TRIANGLE INEQUALITY yields

$$\ell([L]; 0) \le \ell([L]; 0) + \ell([L]; 0),$$

which means $\ell([L]; 0) \ge 0$, and thus $\ell([L]; 0) = 0$.

• MODULE STRUCTURE: The proof of this inequality is formally similar to that of TRIANGLE INEQUALITY. The only difference is that H^2 is timeperiodic, and that we map the surface $\Upsilon_{\bullet} = \Upsilon_{\star}$ to the surface Σ_{\bullet} so that the complement is the boundary together with a segment connecting the interior puncture to the boundary (see Figure 1).

The perturbation datum (K, I) on Υ_{\bullet} is still given by the same formulas as in the proof of TRIANGLE INEQUALITY above, and we push it to Σ_{\bullet} and extend the result by continuity. Computing the energy of a solution of the Floer PDE with respect to this perturbation datum implies that the restriction of $\bullet_{K,I}$ to the filtered Floer complexes satisfies

•_{K,I}:
$$CF^{a_2}_*(H^2) \otimes CF^{a_1}_*(H^1:L) \to CF^{a_1+a_2+\varepsilon}_*(H^3:L)$$

for arbitrary $\varepsilon > 0$, from which the desired inequality readily follows.

• DUALITY: The first equality is obvious from the definition of homological (see (14)) and cohomological (see (16)) spectral invariants, together with the fact that the duality isomorphism (8) flips the sign of the action of the generators.

To prove the inequality we need a general result on filtered based chain complexes and their duals. We present a self-contained account here for the sake of completeness. **Definition 37.** Let R be a ring. A based filtered chain complex over R is a quadruple $\mathcal{V} = (V, \mathcal{B}, \mathcal{A}, \partial)$ where V is a free R-module, \mathcal{B} is a basis for V, that is

$$V = \bigoplus_{v \in \mathcal{B}} R \cdot v \,,$$

 $\mathcal{A}: \mathcal{B} \to \mathbb{R}$ is a function, and $\partial: V \to V$ is a boundary operator, that is $\partial^2 = 0$, such that $\langle \partial u, v \rangle \neq 0$ implies $\mathcal{A}(u) > \mathcal{A}(v)$ for $u, v \in \mathcal{B}$.

Given such a based filtered chain complex, we can define the associated spectral invariants, as follows. For $a \in R$ define

$$V^a = \bigoplus_{\substack{v \in \mathcal{B} \\ \mathcal{A}(v) < a}} R \cdot v \,.$$

It follows that ∂ preserves V^a , that is $V^a \subset V$ is a subcomplex. We let $i^a: H(V^a) \to H(V)$ be the induced map on homology. The spectral invariant of $\alpha \in H(V)$ is

(20)
$$\ell(\alpha, \mathcal{V}) = \inf\{a \in \mathbb{R} \mid \alpha \in \operatorname{im} i^a\}.$$

A priori $\ell(\alpha, \mathcal{V})$ may be infinite even if $\alpha \neq 0$, therefore we make the assumption that \mathcal{V} is such that this does not happen. The above Floer complex of a regular Floer datum is an example.

We define the dual based filtered cochain complex of \mathcal{V} to be $\mathcal{V}^{\vee} = (V^{\vee}, \mathcal{B}^{\vee}, \mathcal{A}^{\vee}, \partial^{\vee})$. Here $\mathcal{B}^{\vee} = \{v^{\vee} | v \in \mathcal{B}\}$, where v^{\vee} is the functional on V having value 1 on v and 0 on the other elements of \mathcal{B}, V^{\vee} is the subspace of the dual module $\operatorname{Hom}_R(V, R)$ spanned by $\mathcal{B}^{\vee}, \mathcal{A}^{\vee}(v^{\vee}) = \mathcal{A}(v)$ for $v \in \mathcal{B}$ and ∂^{\vee} is the restriction to V^{\vee} of the coboundary operator dual to ∂ .

We can similirally define spectral invariants of \mathcal{V}^{\vee} , as follows. For $a \in \mathbb{R}$ let

(21)
$$V_a^{\vee} = V^{\vee} / \bigoplus_{\substack{v^{\vee} \in \mathcal{B}^{\vee} \\ \mathcal{A}^{\vee}(v^{\vee}) > a}} R \cdot v^{\vee}.$$

Then, since by assumption $\langle \partial^{\vee} v^{\vee}, u^{\vee} \rangle \neq 0$ implies $\mathcal{A}^{\vee}(u^{\vee}) > \mathcal{A}^{\vee}(v^{\vee})$, we see that V_a^{\vee} is a quotient cochain complex of V. We let $i_a \colon H^{\vee}(V^{\vee}) \to H^{\vee}(V_a^{\vee})$ be the induced map on cohomology. The spectral invariant of $\alpha^{\vee} \in H^{\vee}(V^{\vee})$ is

$$\ell^{\vee}(\alpha^{\vee}, \mathcal{V}^{\vee}) = \sup\{a \in \mathbb{R} \mid \alpha^{\vee} \in \ker i_a\}$$

Again, we make the assumption that this is a finite number for $\alpha^{\vee} \neq 0$. We have the duality pairings

$$\langle \cdot, \cdot \rangle {:} \; H^{\vee}(V^{\vee}) \times H(V) \to R \quad \text{and} \quad \langle \cdot, \cdot \rangle {:} \; H^{\vee}(V_a^{\vee}) \times H(V^a) \to R$$

induced from the duality pairing between V^{\vee} and V. The main result here is the following lemma.

Lemma 38. We have, for $\alpha^{\vee} \in H^{\vee}(V^{\vee})$: $\ell^{\vee}(\alpha^{\vee}, \mathcal{V}^{\vee}) \leq \inf \{\ell(\alpha, \mathcal{V}) \mid \alpha \in H(V) : \langle \alpha^{\vee}, \alpha \rangle \neq 0\}.$

Proof. We have the following commutative diagram

$$\begin{array}{c} H^{\vee}(V^{\vee}) \longrightarrow \left(H(V)\right)^{\vee} \\ \downarrow_{i_{a}} \qquad \qquad \downarrow^{(i^{a})^{\vee}} \\ H^{\vee}(V_{a}^{\vee}) \longrightarrow \left(H(V^{a})\right)^{\vee} \end{array}$$

where the horizontal arrows are induced from the duality pairings whereas $(i^a)^{\vee}$ is the dual map of i^a .

Assume $\alpha \in H(V)$ is such that $\langle \alpha^{\vee}, \alpha \rangle \neq 0$ and $\ell(\alpha, \mathcal{V}) \leq a \in \mathbb{R}$. Then there is $\alpha' \in H(V^a)$ with $i^a(\alpha') = \alpha$. We then have, using the above diagram:

$$\langle i_a(\alpha^{\vee}), \alpha' \rangle = \langle \alpha^{\vee}, i^a(\alpha') \rangle = \langle \alpha^{\vee}, \alpha \rangle \neq 0,$$

meaning $i_a(\alpha^{\vee}) \neq 0$, that is $\alpha^{\vee} \notin \ker i_a$, which implies $a \geq \ell^{\vee}(\alpha^{\vee}, \mathcal{V}^{\vee})$. This proves the desired inequality.

The inequality in DUALITY property now follows by noting that $CF_*(H : L; \mathcal{L})$ is a based filtered chain complex in the sense of Definition 37, while the cochain complex $CF^*(H : L; \mathcal{L})$ is its dual (up to the sign in the differential, which of course does not matter), and the definitions of spectral invariants (14), (16) are particular cases of (20), (21).

In order to prove the last assertion of this item, namely the one about the equality

$$\ell(\alpha^{\vee}; H) = \inf \left\{ \ell(\beta; H) \, | \, \langle \alpha^{\vee}, \beta \rangle \neq 0 \right\},\,$$

assume that the ground ring R is a field and that the Floer complex $CF_*(H : L)$ is finite-dimensional in every degree. This is then a particular case of a general statement about filtered graded chain complexes. We say that a filtered chain complex \mathcal{V} is graded if there is a function $|\cdot|: \mathcal{B} \to \mathbb{Z}$, called the grading, and we have the decomposition

$$V = \bigoplus_{k \in \mathbb{Z}} V_k$$
 where $V_k = \bigoplus_{v \in \mathcal{B}, |v|=k} R \cdot v$,

and the boundary operator ∂ maps V_k into V_{k-1} . Note that in this case the dual filtered cochain complex \mathcal{V}^{\vee} is also graded, in the sense that there is an analogous decomposition by degree and the differential ∂^{\vee} raises the degree by 1.

Lemma 39. Let $\mathcal{V} = (V, \mathcal{B}, |\cdot|, \mathcal{A}, \partial)$ be a filtered graded chain complex over a field R and assume it is finite-dimensional in every degree. Then

$$\ell^{\vee}(\alpha^{\vee}, \mathcal{V}^{\vee}) = \inf \left\{ \ell(\alpha, \mathcal{V}) \, | \, \langle \alpha^{\vee}, \alpha \rangle \neq 0 \right\}.$$

Proof. Note that it is enough to prove this for homogeneous $\alpha^{\vee} \in V_k^{\vee}$. Next, it suffices to show that for any $a \in \mathbb{R}$ with $a > \ell^{\vee}(\alpha^{\vee}, \mathcal{V}^{\vee})$ there is $\alpha \in H(V)$ with $\langle \alpha^{\vee}, \alpha \rangle \neq 0$ such that $\ell(\alpha, \mathcal{V}) \leq a$. Assume therefore $a > \ell^{\vee}(\alpha^{\vee}, \mathcal{V}^{\vee})$, that is $i_a(\alpha^{\vee}) \neq 0$. The assumptions on R and the degreewise finite-dimensionality of V, and therefore of V^{\vee} , imply that the natural map

$$H_k^{\vee}(V_a^{\vee}) \to (H_k(V^a))^{\vee}$$

is an isomorphism.¹⁶ Thus there is $\alpha' \in H_k(V^a)$ such that $\langle i_a(\alpha^{\vee}), \alpha' \rangle \neq 0$. Put $\alpha = i^a(\alpha')$ and then we see that

$$\langle \alpha^{\vee}, \alpha \rangle = \langle i_a(\alpha^{\vee}), \alpha' \rangle \neq 0$$

and by construction $\ell(\alpha, \mathcal{V}) \leq a$.

• NOVIKOV ACTION: By the definition of the action of the Novikov ring R[F] on the Lagrangian Floer complex $CF_*(H : L)$, the multiplication by the element $A \in F$ induces a chain isomorphism

$$CF_*(H:L) \to CF_{*-\mu(A)}(H:L)$$
,

which restricts to a chain isomorphism

$$CF^a_*(H:L) \to CF^{a-\omega(A)}_{*-\mu(A)}(H:L)$$

for any $a \in \mathbb{R}$. These chain maps induce the two left vertical arrows in the following commutative diagram:

$$\begin{split} HF^a_*(H,J:L) & \xrightarrow{i^a_*} HF_*(H:L) < \xrightarrow{\mathrm{PSS}^{H,J}} QH_*(L) \\ & \downarrow^A & \downarrow^A & \downarrow^A \\ HF^{a-\omega(A)}_{*-\mu(A)}(H,J:L) \xrightarrow{i^a_*-\omega(A)} HF_{*-\mu(A)}(H,J:L) \xleftarrow{\mathrm{PSS}^{H,J}} QH_{*-\mu(A)}(L) \end{split}$$

where the right arrow is the action of the Novikov ring on quantum homology. The equality now follows from the definition of spectral invariants.

• LAGRANGIAN CONTROL: Notice that it is sufficient to prove this property for smooth Hamiltonians. Assume that H restricted to L is a function of time, $H_t|_L = c(t)$. Then for any $s \in \mathbb{R}$ we see that $H^s = sH$ satisfies $H_t^s|_L = sc(t)$. Since L is Lagrangian, for any fixed s, the chords of H^s are contained in L. Each of these comes with a natural capping (itself), contained in L so that it has area 0. This shows that for all $s \in \mathbb{R}$

$$\operatorname{Spec}(H^s) = \left\{ s \cdot \int_0^1 c(t) \, dt + k \mathsf{A} \, \big| \, k \in \mathbb{Z} \right\}.$$

Now, by Spectrality and Continuity, $\ell(\alpha; H^s) = s \int_0^1 c(t) dt + k_0 \mathsf{A}$ for some $k_0 \in \mathbb{Z}$, independent of s. Since $H^0 = 0$, $k_0 = \nu(\alpha)$ by NORMALIZA-TION. We get for s = 1 that $\ell(\alpha; H) = \nu(\alpha)\mathsf{A} + \int_0^1 c(t) dt$.

60

¹⁶In fact, we only use the injectivity of this map.

In case H satisfies $H_t|_L \leq c(t)$ for all t, pick any function K such that $K \geq H$ and $K_t|_L = c(t)$. By MONOTONICITY, $\ell(\alpha; H) \leq \ell(\alpha; K) = \nu(\alpha)\mathsf{A} + \int_0^1 c(t) dt$. The last case is similar and the bounds on $\ell(\alpha; H)$ follow.

• NON-NEGATIVITY: By TRIANGLE INEQUALITY with $\alpha = \beta = [L]$, we have $\ell_+(H) + \ell_+(\overline{H}) \geq \ell_+(H \sharp \overline{H})$. By NORMALIZATION and the definition of \overline{H} and \sharp , it suffices to prove the result when H is normalized. In that case, $H \sharp \overline{H}$ is also normalized and is equivalent to 0, in the sense of Definition 29. By invariance of spectral invariants, see §3.4, $\ell_+(H \sharp \overline{H}) = \ell_+(0)$ which vanishes by NORMALIZATION.

• MAXIMUM: First, assume that H is normalized. Since $\alpha = [L] \star \alpha$ and $H \sharp 0$ is equivalent to H in the sense of Definition 29, by §3.4 and TRIANGLE IN-EQUALITY we obtain $\ell(\alpha; H) \leq \ell_+(H) + \ell(\alpha; 0)$. When H is not normalized, the result follows from the first case and NORMALIZATION.

Theorem 36, and consequently Theorem 3 and Proposition 5, are now proved.

4.2.2. The equivalence of Hamiltonian spectral invariants and Lagrangian spectral invariants of the diagonal. In this subsection we prove Theorem 6 of the introduction.

Recall from $\S2.8$ that there is a canonical isomorphism

$$QH_*(\Delta) = QH_*(M) \,.$$

It is enough to show this for a nondegenerate smooth H. Let us therefore assume that (H, J) is a regular time-periodic Floer datum on M. In view of Remark 30, we can assume that $H_t \equiv 0$ for $t \in [\frac{1}{2}, 1]$.

In §2.8 it is shown that there is a regular Floer datum $(\widehat{H}, \widehat{J})$ for Δ , for which there is a canonical chain isomorphism

$$(CF_*(H), \partial_{H,J}) = (CF_*(\widehat{H} : \Delta), \partial_{\widehat{H},\widehat{J}}),$$

preserving the grading and the action of the generators. Moreover, for any $a \in \mathbb{R}$ we have the following commutative diagram, where the left two vertical arrows are induced by this isomorphism and the right vertical arrow is an analogous isomorphism on quantum homology, described *ibid*.:

From this we have the equality of spectral invariants

$$\ell(\alpha; H) = c(\alpha; H).$$

Recall how \widehat{H} is defined: it is the direct sum $\widehat{H}_t = H_t \oplus 0$ for $t \in [0, \frac{1}{2}]$. The Hamiltonian \widehat{H} therefore differs from $H \oplus 0$ by time reparametrization, which means that $\ell(\alpha; H \oplus 0) = \ell(\alpha; \widehat{H})$, thereby completing the proof of the theorem.

4.2.3. Spectral invariants of products of Lagrangians. Here we formulate precisely and prove the property of the spectral invariants mentioned at the end of §1.1.

In §2.9 we introduced, for monotone Lagrangians $L_i \subset (M_i, \omega_i), i = 1, 2$, the canonical map

$$QH_*(L_1) \otimes QH_*(L_2) \rightarrow QH_*(L_1 \times L_2).$$

Theorem 40. For Hamiltonians H_i on M_i and classes $\alpha_i \in QH_*(L_i)$ we have

$$\ell(\alpha_1 \otimes \alpha_2; H_1 \oplus H_2) \le \ell(\alpha_1; H_1) + \ell(\alpha_2; H_2)$$

with equality if R is a field.

REMARK 41. We cannot omit the assumption that R is a field in order to obtain an equality in Theorem 40. The reason for this is that even if both classes α_1 , α_2 are nonzero, their tensor product $\alpha_1 \otimes \alpha_2$ in $QH_*(L_1) \otimes QH_*(L_2)$ may vanish.

Proof. We have the following commutative diagram

$$\begin{aligned} HF_*^{a_1}(H_1, J_1: L_1) \times HF_*^{a_2}(H_2, J_2: L_2) &\longrightarrow HF_*^{a_1+a_2}(H_1 \oplus H_2, J_1 \oplus J_2: L_1 \times L_2) \\ & \downarrow_{i_*^{a_1} \times i_*^{a_2}} & \downarrow_{i_*^{a_1+a_2}} \\ HF_*(H_1, J_1: L) \times HF_*(H_2, J_2: L) &\longrightarrow HF_*(H_1 \oplus H_2, J_1 \oplus J_2: L_1 \times L_2) \\ & \uparrow_{PSS \times PSS} & \uparrow_{PSS} \\ QH_*(L_1) \times QH_*(L_2) &\longrightarrow QH_*(L_1 \times L_2) \end{aligned}$$

which implies the inequality.

In case R is a field, the equality can be proved using the method developed in [EP09, Section 5].

4.3. Spectral invariants of isotopies.

Theorem 42. Let L be a closed monotone Lagrangian of (M, ω) with minimal Maslov number $N_L \geq 2$. The function

 $\ell \colon QH_*(L) \times \widetilde{\operatorname{Ham}}(M, \omega) \longrightarrow \mathbb{R} \cup \{-\infty\}$

constructed in §3.4 above satisfies the following properties.

FINITENESS: $\ell(\alpha; \widetilde{\phi}) = -\infty$ if and only if $\alpha = 0$.

Spectrality: For all $\alpha \neq 0$, $\ell(\alpha; \tilde{\phi}) \in \operatorname{Spec}(\tilde{\phi}: L)$.

GROUND RING ACTION: For all $r \in R$, $\ell(r \cdot \alpha; \tilde{\phi}) \leq \ell(\alpha; \tilde{\phi})$, with equality if r is a unit.

NORMALIZATION: We have $\ell(\alpha; id) = \nu(\alpha)A$ and $\ell_+(id) = 0$.

CONTINUITY: Assume H and K are normalized, then

$$\int_0^1 \min_M (H_t - K_t) dt \le \ell(\alpha; \widetilde{\phi}_H) - \ell(\alpha; \widetilde{\phi}_K) \le \int_0^1 \max_M (H_t - K_t) dt \, .$$

MONOTONICITY: If M is noncompact, H and K have compact support, and $H \leq K$, then $\ell(\alpha; \tilde{\phi}_H) \leq \ell(\alpha; \tilde{\phi}_K)$.

TRIANGLE INEQUALITY: For all α and β , ¹⁷

$$\ell(\alpha \star \beta; \widetilde{\phi}\widetilde{\psi}) \le \ell(\alpha; \widetilde{\phi}) + \ell(\beta; \widetilde{\psi})$$

MODULE STRUCTURE: For all $a \in QH_*(M)$ and $\alpha \in QH_*(L)$, $\ell(a \bullet \alpha; \widetilde{\phi}\widetilde{\psi}) \leq c(a; \widetilde{\psi}) + \ell(\alpha; \widetilde{\phi})$.

LAGRANGIAN CONTROL: If for all t, $H_t|_L = c(t) \in \mathbb{R}$ (respectively $\leq \geq$), then

$$\ell(\alpha; \widetilde{\phi}_H) = \nu(\alpha) \mathsf{A} + \int_0^1 \left(c(t) - \int_M H_t \, \omega^n \right) dt \quad (respectively \leq \geq).$$

Thus, for all H:

$$\int_0^1 \min_L H_t \, dt \le \ell(\alpha; \widetilde{\phi}_H) - \nu(\alpha) \mathsf{A} + \int_0^1 \int_M H_t \, \omega^n \, dt \le \int_0^1 \max_L H_t \, dt \, .$$

NON-NEGATIVITY: $\ell_+(\phi) + \ell_+(\phi^{-1}) \ge 0.$

MAXIMUM: $\ell(\alpha; \widetilde{\phi}) \leq \ell_+(\widetilde{\phi}) + \ell(\alpha; \mathrm{id}) = \ell_+(\widetilde{\phi}) + \nu(\alpha) \mathsf{A}.$

DUALITY: For $\alpha \in QH_*(L)$ and $\alpha^{\vee} \in QH^{n-*}(L; \mathcal{L})$ corresponding to it via the duality isomorphism (9) we have

$$-\ell(\alpha;\widetilde{\phi}^{-1}) = \ell(\alpha^{\vee};\widetilde{\phi}) \le \inf\left\{\ell(\beta;\widetilde{\phi}) \,|\, \langle \alpha^{\vee},\beta \rangle \neq 0\right\}$$

with equality if the ground ring R is a field and the Floer complexes of nondegenerate Hamiltonians are finite-dimensional in each degree.

NOVIKOV ACTION: In case there is an action of the Novikov ring R[F] as above, we have, for $A \in F$: $\ell(A \cdot \alpha; \widetilde{\phi}) = \ell(\alpha; \widetilde{\phi}) - \omega(A)$.

SYMPLECTIC INVARIANCE: Let $\psi \in \text{Symp}(M, \omega)$, $L' = \psi(L)$ and let ℓ' be the associated spectral invariant function. Then¹⁸ $\ell(\alpha; \tilde{\phi}) = \ell'(\psi_*(\alpha); \psi \tilde{\phi} \psi^{-1})$.

First notice that we assume M to be noncompact for MONOTONICITY. This comes from the facts that isotopies naturally correspond to *normalized* Hamiltonians, and that there are no normalized Hamiltonians H and K satisfying $H \leq K$ on a compact manifold except if they coincide everywhere. On noncompact manifolds, isotopies also correspond to normalized Hamiltonians and the natural normalization condition is requiring the Hamiltonian to have compact support.

¹⁷Note that the order of α and β here is different from the one in TRIANGLE INEQUALITY property in Theorem 36; this is because $\tilde{\phi}_{H\sharp K} = \tilde{\phi}_K \tilde{\phi}_H$.

 $^{^{18}\}mathrm{Here}$ we use the natural action of the group $\mathrm{Symp}(M,\omega)$ on $\widetilde{\mathrm{Ham}}(M,\omega)$ by conjugation.

This theorem follows in a straightforward manner from Theorem 36, with the exception of LAGRANGIAN CONTROL and SYMPLECTIC INVARIANCE. To prove LAGRANGIAN CONTROL as stated here it suffices to note that the Hamiltonian $H_t - \int_M H_t \omega^n$ is normalized and generates the same Hamiltonian flow as H_t , therefore

$$\ell(\alpha; \widetilde{\phi}_H) = \ell(\alpha; H_t - \int_M H_t \,\omega^n) = \ell(\alpha; H) - \int_0^1 \int_M H_t \,\omega^n \,dt \,,$$

and the property now follows from LAGRANGIAN CONTROL for Hamiltonian functions (Theorem 36).

For SYMPLECTIC INVARIANCE it suffices to use Theorem 36 together with the observation that if $\tilde{\phi} = \tilde{\phi}_H$ then $\psi \tilde{\phi}_H \psi^{-1} = \tilde{\phi}_{H \circ \psi^{-1}}$.

5. Proof of the Hofer bounds

We now prove Propositions 7, 10, assuming Theorem 8, whose proof follows immediately after.

Proof of Proposition 7. The bounds for $\tilde{\chi}$ follow from the bounds for H if we set $\tilde{\chi} = \tilde{\phi}_H$, therefore we establish the latter.

Let L and L' be Hamiltonian isotopic Lagrangians. Choose any $\varphi \in$ Ham (M, ω) such that $\varphi(L) = L'$ and fix a class $\alpha \in QH_*(L)$.

First, by SYMPLECTIC INVARIANCE, $\ell'(\alpha'; H) = \ell(\alpha; H \circ \varphi)$ with $\alpha = \varphi_*^{-1}(\alpha')$ so that we are interested in $|\ell(\alpha; H) - \ell(\alpha; H \circ \varphi)|$. Since adding the same function of time to both H and $H \circ \varphi$ does not affect their difference, by NORMALIZATION we get

$$|\ell(\alpha; H) - \ell(\alpha; H \circ \varphi)| = |\ell(\alpha; \phi_H) - \ell(\alpha; \varphi^{-1} \phi_H \varphi)|.$$

Now, it is useful to notice that for any Hamiltonian $K: M \times [0,1] \rightarrow \mathbb{R}$ such that $\phi_K = \varphi$, the Hamiltonian¹⁹ isotopies $t \mapsto \varphi^{-1} \phi_H^t \varphi$ and $t \mapsto (\phi_K^t)^{-1} \phi_H^t \phi_K^t$ are homotopic with fixed endpoints and thus define the same element in $\operatorname{Ham}(M, \omega)$, that is $\varphi^{-1} \widetilde{\phi}_H \varphi = \widetilde{\phi}_K^{-1} \widetilde{\phi}_H \widetilde{\phi}_K$.

Using this, the fact that the fundamental class [L] is the unit of the quantum homology ring, and TRIANGLE INEQUALITY, we get

$$\ell(\alpha; \varphi^{-1}\widetilde{\phi}_H \varphi) = \ell([L] \star \alpha \star [L]; \widetilde{\phi}_K^{-1}\widetilde{\phi}_H \widetilde{\phi}_K) \le \ell_+(\widetilde{\phi}_K^{-1}) + \ell(\alpha; \widetilde{\phi}_H) + \ell_+(\widetilde{\phi}_K)$$

from which we deduce that

$$\ell(\alpha; \widetilde{\phi}_H) - \ell(\alpha; \varphi^{-1}\widetilde{\phi}_H \varphi) \ge -\ell_+(\widetilde{\phi}_K^{-1}) - \ell_+(\widetilde{\phi}_K).$$

In a similar way, by writing $\ell(\alpha; \tilde{\phi}_H) = \ell(\alpha; \tilde{\phi}_K(\tilde{\phi}_K^{-1}\tilde{\phi}_H\tilde{\phi}_K)\tilde{\phi}_K^{-1})$, we get that

$$\ell(\alpha; \widetilde{\phi}_H) - \ell(\alpha; \varphi^{-1}\widetilde{\phi}_H \varphi) \le \ell_+(\widetilde{\phi}_K^{-1}) + \ell_+(\widetilde{\phi}_K)$$

and thus conclude that

$$|\ell(\alpha; H) - \ell(\alpha; H \circ \varphi)| \le \ell_+(\widetilde{\phi}_K^{-1}) + \ell_+(\widetilde{\phi}_K).$$

¹⁹It is well-known that a smooth one-parameter family of Hamiltonian diffeomorphisms is a Hamiltonian isotopy.

By CONTINUITY of spectral invariants, since $\ell_+(\widetilde{\phi}_K) \leq \int_0^1 \max_M K_t dt$ and $\ell_+(\widetilde{\phi}_K^{-1}) \leq \int_0^1 \max_M \overline{K}_t dt = -\int_0^1 \min_M K_t dt$, we deduce

$$|\ell(\alpha; H) - \ell(\alpha; H \circ \varphi)| \le \int_0^1 \operatorname{osc}_M K_t \, dt \, .$$

Taking infimum over all K with $\varphi = \phi_K$ concludes the proof.

Proposition 10 also follows from the same estimate, since in order to obtain it we need to take infimum over those K generating the given homotopy class of paths \tilde{L} . Theorem 8 implies that the left-hand side is independent of K. By definition, the infimum on the right-hand side equals $\|\tilde{L}\|$.

Before passing to the proof of Theorem 8, we formulate two lemmas. We let \mathcal{L} be the space of Lagrangians Hamiltonian isotopic to L.

Lemma 43. Endow $\operatorname{Ham}(M)$ and \mathcal{L} with the C^{∞} -topologies. Then the map

 $\Pi: \operatorname{Ham}(M) \to \mathcal{L}, \quad \phi \mapsto \phi(L)$

is a fiber bundle with fiber $\Pi^{-1}(L') = \operatorname{Ham}_{L'}(M)$. In particular it is a fibration.

Lemma 44. For $s \in [0,1]$ let $\chi^s \in \text{Ham}(M)$ be such that $\chi^0 = \text{id}_M$ and $\chi^s(L) = L$ for all s. Then $\chi^1_* = \text{id}$ on $QH_*(L)$.

We now prove the theorem, assuming the lemmas.

Proof of Theorem 8. Let $\{L_{s,t}\}_{(s,t)\in[0,1]^2}$ be a two-parameter family of Lagrangians in the space \mathcal{L} , such that for all s we have $L_{s,0} = L$ and $L_{s,1} = L'$. We wish to show that if $\phi^t, \psi^t \in \operatorname{Ham}(M)$ are such that $\phi^0 = \psi^0 = \operatorname{id}_M$ and $\phi^t(L) = L_{0,t}, \psi^t(L) = L_{1,t}$, then $\phi^1_* = \psi^1_* \colon QH_*(L) \to QH_*(L')$.

By Lemma 43, there exists a two-parameter family $\phi^{s,t} \in \text{Ham}(M)$ such that $\phi^{s,0} = \text{id}_M, \phi^{0,t} = \phi^t, \phi^{1,t} = \psi^t, \phi^{s,1}(L) = L'$, and finally $\phi^{s,t}(L) = L_{s,t}$. Define $\chi^s = (\phi^1)^{-1}\phi^{s,1}$. Then $\chi^s(L) = L, \chi^0 = \text{id}_M$, and $\chi^1 = (\phi^1)^{-1}\psi^1$. Observe that for two symplectomorphisms $\sigma_1, \sigma_2 \in \text{Symp}(M, \omega)$ we have

$$(\sigma_2 \circ \sigma_1)_* = (\sigma_2)_* \circ (\sigma_1)_* \colon QH_*(L) \to QH_*(\sigma_2(\sigma_1(L)))$$

Therefore $\chi_*^1 = (\phi_*^1)^{-1} \circ \psi_*^1$, and since by Lemma 44 we have $\chi_*^1 = id$, the proof of the theorem is complete.

Proof of Lemma 43. Given $L' \in \mathcal{L}$, we must trivialize Π over a neighborhood of L'. Note that the C^{∞} -topology is stronger than the C^1 -topology. Using the Weinstein neighborhood theorem, a tubular neighborhood of L' inside M can be identified with a tubular neighborhood of the zero section in T^*L' . Consequently a C^1 -neighborhood of L' in \mathcal{L} is given by a C^0 -neighborhood of zero in the space of exact 1-forms on L'. Let us denote the corresponding neighborhood of $L' \in \mathcal{L}$ by \mathcal{U} . Since elements of \mathcal{U} are given by graphs of exact 1-forms, we can use the corresponding functions on L', pull them back by the projection $T^*L' \to L'$ and cut off to obtain Hamiltonian functions on M such that the time-1 map of the function corresponding to $L'' \in \mathcal{U}$ maps

L' to L''. This yields a local section of Π over \mathcal{U} , and since $\operatorname{Ham}(M)$ is a group, we obtain a trivialization

$$\operatorname{Ham}(M) \supset \Pi^{-1}(\mathcal{U}) \to \mathcal{U} \times \operatorname{Ham}_{L'}(M) \,.$$

This shows that Π is indeed the projection of a fiber bundle.

Proof of Lemma 44. From the construction in Section 2.10 it follows that if (H, J) is a regular Floer datum for L, then we have the following commutative diagram

From our definition of the PSS isomorphisms we have the following commutative diagram, where Φ stands for the continuation isomorphism:

$$HF_*(H, J:L) \xrightarrow{\Phi} HF_*(H^{\chi^1}, J^{\chi^1}:L)$$

$$\xrightarrow{\text{PSS}} \qquad \qquad \uparrow \text{PSS}$$

$$QH_*(L) \xrightarrow{\text{id}} QH_*(L)$$

It follows that the lemma will be proved if we can show that

$$\chi^1_* = \Phi: HF_*(H, J: L) \to HF_*(H^{\chi^1}, J^{\chi^1}: L).$$

In fact, we will show that this holds on the chain level, provided the perturbation datum in the definition of the continuation morphism is suitably chosen.

The main point of the proof is as follows. Recall that given two regular Floer data for L, (H^i, J^i) , i = 0, 1, in the definition of the continuation morphism

$$\Phi: HF_*(H^0, J^0: L) \to HF_*(H^1, J^1: L)$$

we used a (generically chosen) homotopy of Floer data $(H^s, J^s)_{s \in \mathbb{R}}$, stationary for s outside (0, 1). This is shown to be independent of the chosen homotopy by considering homotopies of homotopies and the resulting homotopy operators. In fact, we will still obtain the same continuation morphism on homology level if we use any perturbation datum on $\mathbb{R} \times [0, 1]$, compatible with (H^i, J^i) , the reason being that the set of compatible perturbation data is contractible, and therefore one can similarly construct homotopy operators between any two continuation morphisms, including the case of more general perturbation data. The perturbation datum corresponding to an s-dependent homotopy is $(K(s,t) = H^s_t \otimes dt, I(s,t) = J^s_t)$, and the only technical novelty here is that we can use perturbation data of the form

$$(K(s,t) = F(s,t) \otimes ds + G(s,t) \otimes dt, I(s,t) = J_t^s),$$

66

where F(s, t) vanishes for s outside (0, 1), and (as follows from the definition of a perturbation datum) F(i, t) is a function on M whose restriction to L is a time-dependent constant, for i = 0, 1. The Floer equation on $\mathbb{R} \times [0, 1]$ corresponding to this more general perturbation datum is

$$\partial_s u - X_F(u) + I(\partial_t u - X_G(u)) = 0.$$

This equation is obtained from

$$(du - X_K)^{0,1} = 0$$

upon substituting the vector $\partial_s \in T(\mathbb{R} \times [0, 1])$.

Assume we have a solution $u \in \widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}_{-}, \widetilde{\gamma}_{+})$. Without loss of generality we can assume that χ^s is independent of s for s near 0, 1, and we extend the definition of χ^s to all $s \in \mathbb{R}$ in an obvious way. Let us define $v: \mathbb{R} \times [0, 1] \to M$ by $v(s, t) = \chi^s(u(s, t))$, that is $u(s, t) = (\chi^s)^{-1}(v(s, t))$. Since u solves

$$\partial_s u + J(\partial_t u - X_H) = 0 \,,$$

it follows that

$$(\chi_*^s)^{-1}\partial_s v + \frac{\partial(\chi^s)^{-1}}{\partial s}(v) + J((\chi_*^s)^{-1}\partial_t v - X_H((\chi^s)^{-1}(v)) = 0,$$

and precomposing with χ_*^s we obtain

$$\partial_s v + \chi_*^s \frac{\partial(\chi_*^s)^{-1}}{\partial s}(v) + I(\partial_t v - X_{H \circ \chi^s}(v)) = 0,$$

where $I = \chi_*^s \circ J \circ (\chi_*^s)^{-1}$. Note that the vector field

$$-\chi_*^s \frac{\partial(\chi_*^s)^{-1}}{\partial s}$$

is Hamiltonian, since χ^s is a Hamiltonian path. Therefore it is generated by a unique normalized Hamiltonian F(s,t). If we also let $G(s,t) = H_t \circ \chi^s$, we see that the equation we obtained is

$$\partial_s v - X_F(v) + I(\partial_t v - X_G(v)) = 0$$

that is v solves the Floer equation corresponding to the perturbation datum (K = F ds + G dt, I). We have therefore constructed a map

$$\widetilde{\mathcal{M}}(H,J;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+}) \to \mathcal{M}(K,I;\widetilde{\gamma}_{-},\widetilde{\gamma}_{+}^{\chi^{1}}),$$

and it is easy to see that it is a diffeomorphism.

Abbreviate $\chi \equiv \chi^1$. Recall the definition of χ_* from Section 2.10. The upshot is that it is induced by a chain isomorphism

$$\chi_*: CF_*(H:L) \to CF_*(H^{\chi}:L) ,$$

whose only nonzero matrix elements are the isomorphisms

$$C(\widetilde{\gamma}) \to C(\widetilde{\gamma}^{\chi})$$

induced from the isomorphism of Fredholm operators

$$D_{\widehat{\gamma}} \to D_{\widehat{\gamma}\chi}$$

Now, the continuation map

$$\Phi: CF_*(H:L) \to CF_*(H^{\chi}:L)$$

is defined as follows (see Section 2.2). For $\tilde{\gamma} \in \operatorname{Crit} \mathcal{A}_{H:L}$ and $\delta \in \operatorname{Crit} \mathcal{A}_{H^{\chi}:L}$ the corresponding matrix element

$$C(\widetilde{\gamma}) \to C(\widetilde{\delta})$$

is the sum of isomorphisms C(v) where v runs over the space $\mathcal{M}(K, I; \tilde{\gamma}, \tilde{\delta})$, where we use the perturbation datum defined above from χ^s . It follows from the definitions that in order for v to contribute to the matrix element the moduli space to which it belongs must be zero-dimensional. We have just seen that there is a natural diffeomorphism

$$\mathcal{M}(K, I; \widetilde{\gamma}, \widetilde{\delta}) \simeq \widetilde{\mathcal{M}}(H, J; \widetilde{\gamma}, \widetilde{\delta}^{\chi^{-1}}).$$

It follows that the latter space is also zero-dimensional, which is only possible if $\tilde{\delta}^{\chi^{-1}} = \tilde{\gamma}$, that is $\tilde{\delta} = \tilde{\gamma}^{\chi}$, in which case it consists of the map $(s, t) \mapsto \gamma(t)$. Thus the only nonzero matrix elements of Φ are the isomorphisms

$$C(v): C(\widetilde{\gamma}) \to C(\widetilde{\gamma}^{\chi}),$$

where $v(s,t) = \chi^s(\gamma(t))$. It remains to show that C(v) coincides with the above isomorphism $C(\tilde{\gamma}) \to C(\tilde{\gamma}^{\chi})$ coming from χ .

This is done as follows. Pick a smooth function $h: \mathbb{R} \times [0,1] \to \mathbb{R}$ with the following properties: $h(s,\sigma) = 0$ for $s \leq 0$, $h(s,\sigma) = \sigma$ for $s \geq 1$, and $\partial_s h \geq 0$. Define $\rho^{\sigma}: \mathbb{R} \to \operatorname{Ham}(M)$ by $\rho^{\sigma}(s) = \chi^{h(s,\sigma)}$. It then follows that $\rho^0 \equiv \operatorname{id}, \rho^{\sigma}(s) = \chi^{\sigma}$ for $s \geq 1$. That is, ρ^{σ} is a Hamiltonian path from id to χ^{σ} in $\operatorname{Ham}_L(M)$, stationary for s outside (0, 1).

Now we can run the above procedure with the path ρ^{σ} instead of the original path $(\chi^s)_s$. As a result we obtain isomorphisms

$$\Phi, \chi^{\sigma}_* \colon C(\widetilde{\gamma}) \to C(\widetilde{\gamma}^{\chi^o}),$$

and we will be done if we can show that they coincide, since the desired result is obtained for $\sigma = 1$. Note that for every σ these isomorphisms in fact come from isomorphisms of determinant bundles

$$\mathrm{d}(D_{\widetilde{\gamma}})\simeq \mathrm{d}(D_{\widetilde{\gamma}\chi^{\sigma}})\,.$$

Fix an orientation \mathfrak{o} of $d(D_{\widetilde{\gamma}})$ and let $\mathfrak{o}_{\sigma}^{\chi}$ and $\mathfrak{o}_{\sigma}^{\Phi}$ be the orientations of $d(D_{\widetilde{\gamma}\chi^{\sigma}})$ obtained using the isomorphisms χ_*^{σ} , Φ , respectively. We have $\mathfrak{o}_0^{\chi} = \mathfrak{o} = \mathfrak{o}_0^{\Phi}$, since for $\sigma = 0$ we have $\chi_*^0 = \mathrm{id} = \Phi$. On the other hand, these orientations depend continuously on σ , and therefore $\mathfrak{o}_{\sigma}^{\chi} = \mathfrak{o}_{\sigma}^{\Phi}$ for all σ . It follows that the isomorphisms $\chi_*^{\sigma}, \Phi: C(\widetilde{\gamma}) \to C(\widetilde{\gamma}^{\chi^{\sigma}})$ map the generator \mathfrak{o} to the same generator for all σ , meaning that they coincide, which concludes the proof of the lemma.

6. PROOFS CONCERNING SYMPLECTIC RIGIDITY

Proof of Proposition 15. Let $F \in C^{\infty}(M)$ and let $c = \min_{L} F$. By MODULE STRUCTURE, NORMALIZATION, and LAGRANGIAN CONTROL we have for $k \in \mathbb{N}$:

$$c(e;kF) + \ell([L];0) \ge \ell(e \bullet [L];kF) \ge kc + \nu(e \bullet [L])\mathsf{A},$$

which yields

$$\lim_{k \to \infty} c(e; kF)/k \ge c = \min_L F \,,$$

meaning L is e-heavy.

Proof of Theorem 17. The proof is an application of Proposition 15, combined with results appearing in §2.6. See also [Zap15, Section 8]. In both cases we take \mathbb{C} as the ground ring.

- Let M = CP² and L = L_{CP²}. In §2.6 we saw that there is a choice of twisted coefficients for which the homology of the quotient complex corresponding to π⁰₂(M, L) does not vanish. Since [M] = [CP²] is the unit of QH_{*}(CP²), it follows that [M] [L] = [L] ≠ 0.
 Abbreviate M = S² × S² and L = L_{S²×S²}. Again in §2.6 it is shown
- Abbreviate $M = S^2 \times S^2$ and $L = L_{S^2 \times S^2}$. Again in §2.6 it is shown that there is a choice of twisted coefficients on the quotient complex corresponding to $\pi_2^0(M, L)$, for which $QH_*(L) \neq 0$. It is known that the Lagrangian antidiagonal $\overline{\Delta} \subset M$ is e_- -superheavy. Since L is disjoint from the antidiagonal, it follows that it is not e_- -superheavy (see [EP09]), therefore Proposition 15 implies $e_- \bullet [L] = 0$. Therefore, since $[M] = e_+ + e_-$, we have

$$e_{+} \bullet [L] = e_{-} \bullet [L] + e_{+} \bullet [L] = (e_{-} + e_{+}) \bullet [L] = [M] \bullet [L] = [L],$$

and the result follows from another use of Proposition 15.

References

[Abo15] Mohammed Abouzaid. Symplectic cohomology and Viterbo's theorem. In Free loop spaces in geometry and topology, volume 24 of IRMA Lect. Math. Theor. Phys., pages 271–485. Eur. Math. Soc., Zürich, 2015. [AF10] Peter Albers and Urs Frauenfelder. Spectral invariants in Rabinowitz–Floer homology and global Hamiltonian perturbations. J. Mod. Dyn., 4(2):329-357, 2010. [Alb05] Peter Albers. On the extrinsic topology of Lagrangian submanifolds. Int. Math. Res. Not., (38):2341-2371, 2005. [Alb08] Peter Albers. A Lagrangian Piunikhin–Salamon–Schwarz morphism and two comparison homomorphisms in Floer homology. Int. Math. Res. Not., (4):Art. ID rnm134, 56, 2008. [AM13] Peter Albers and Will J. Merry. Orderability, contact non-squeezing, and Rabinowitz-Floer homology. 2013. arXiv:1302.6576. [AS10] Alberto Abbondandolo and Matthias Schwarz. Floer homology of cotangent bundles and the loop product. Geom. Topol., 14(3):1569-1722, 2010. [BC01] Paul Biran and Kai Cieliebak. Symplectic topology on subcritical manifolds. Comment. Math. Helv., 76(4):712-753, 2001.

- [BC09] Paul Biran and Octav Cornea. A Lagrangian quantum homology. In New perspectives and challenges in symplectic field theory, volume 49 of CRM Proc. Lecture Notes, pages 1–44. Amer. Math. Soc., Providence, RI, 2009.
- [BC12] Paul Biran and Octav Cornea. Lagrangian topology and enumerative geometry. Geom. Topol., 16(2):963–1052, 2012.
- [BEP12] Lev Buhovsky, Michael Entov, and Leonid Polterovich. Poisson brackets and symplectic invariants. *Selecta Math.* (N.S.), 18(1):89–157, 2012.
- [Bhu01] Mohan Bhupal. A partial order on the group of contactomorphisms of \mathbb{R}^{2n+1} via generating functions. *Turkish J. Math.*, 25(1):125–135, 2001.
- [Bir06] Paul Biran. Lagrangian non-intersections. Geom. Funct. Anal., 16(2):279– 326, 2006.
- [BPS03] Paul Biran, Leonid Polterovich, and Dietmar Salamon. Propagation in Hamiltonian dynamics and relative symplectic homology. *Duke Math. J.*, 119(1):65– 118, 2003.
- [Cha95] Marc Chaperon. On generating families. In *The Floer memorial volume*, volume 133 of *Progr. Math.*, pages 283–296. Birkhäuser, Basel, 1995.
- [Che00] Yu. V. Chekanov. Invariant Finsler metrics on the space of Lagrangian embeddings. Math. Z., 234(3):605–619, 2000.
- [CS10] Yuri Chekanov and Felix Schlenk. Notes on monotone Lagrangian twist tori. Electron. Res. Announc. Math. Sci., 17:104–121, 2010.
- [CS15] Vincent Colin and Sheila Sandon. The discriminant and oscillation lengths for contact and Legendrian isotopies. J. Eur. Math. Soc. (JEMS), 17(7):1657– 1685, 2015.
- [DKM14] Jovana Djuretić, Jelena Katić, and Darko Milinković. Comparison of spectral invariants in Lagrangian and Hamiltonian Floer theory. 2014. to appear in Filomat.
- [EG91] Yakov Eliashberg and Mikhael Gromov. Convex symplectic manifolds. In Several complex variables and complex geometry, Part 2 (Santa Cruz, CA, 1989), volume 52 of Proc. Sympos. Pure Math., pages 135–162. Amer. Math. Soc., Providence, RI, 1991.
- [EP03] Michael Entov and Leonid Polterovich. Calabi quasimorphism and quantum homology. Int. Math. Res. Not., 30:1635–1676, 2003.
- [EP06] Michael Entov and Leonid Polterovich. Quasi-states and symplectic intersections. Comm. Math. Helv., 81(1):75–99, 2006. DOI: 10.4171/CMH/43.
- [EP08] Michael Entov and Leonid Polterovich. Symplectic quasi-states and semisimplicity of quantum homology. In *Toric topology*, volume 460 of *Contemp. Math.*, pages 47–70. Amer. Math. Soc., Providence, RI, 2008.
- [EP09] Michael Entov and Leonid Polterovich. Rigid subsets of symplectic manifolds. Compos. Math., 145(3):773–826, 2009.
- [EP10] Yakov Eliashberg and Leonid Polterovich. Symplectic quasi-states on the quadric surface and Lagrangian submanifolds. 2010. arXiv:1006.2501.
- [EPZ07] Michael Entov, Leonid Polterovich, and Frol Zapolsky. Quasi-morphisms and the Poisson bracket. Pure Appl. Math. Q., 3(4, Special Issue: In honor of Grigory Margulis. Part 1):1037–1055, 2007.
- [FH93] A. Floer and H. Hofer. Coherent orientations for periodic orbit problems in symplectic geometry. Math. Z., 212(1):13–38, 1993.
- [Flo88a] Andreas Floer. Morse theory for Lagrangian intersections. J. Differential Geom., 28(3):513–547, 1988.
- [Flo88b] Andreas Floer. The unregularized gradient flow of the symplectic action. Comm. Pure Appl. Math., 41(6):775–813, 1988.
- [Flo89a] Andreas Floer. Symplectic fixed points and holomorphic spheres. Comm. Math. Phys., 120(4):575–611, 1989.

- [Flo89b] Andreas Floer. Witten's complex and infinite-dimensional Morse theory. J. Differential Geom., 30(1):207–221, 1989.
- [FOOO09a] Kenji Fukaya, Yong-Geun Oh, Hiroshi Ohta, and Kaoru Ono. Lagrangian intersection Floer theory: anomaly and obstruction. Part I, volume 46 of AMS/IP Studies in Advanced Mathematics. American Mathematical Society, Providence, RI; International Press, Somerville, MA, 2009.
- [FOOO09b] Kenji Fukaya, Yong-Geun Oh, Hiroshi Ohta, and Kaoru Ono. Lagrangian intersection Floer theory: anomaly and obstruction. Part II, volume 46 of AMS/IP Studies in Advanced Mathematics. American Mathematical Society, Providence, RI; International Press, Somerville, MA, 2009.
- [FOO011] Kenji Fukaya, Yong-Geun Oh, Hiroshi Ohta, and Kaoru Ono. Spectral invariants with bulk, quasimorphisms and Lagrangian Floer theory. 2011. arXiv:1105.5123.
- [FS07] Urs Frauenfelder and Felix Schlenk. Hamiltonian dynamics on convex symplectic manifolds. Israel J. Math., 159:1–56, 2007.
- [Gad13] Agnès Gadbled. On exotic monotone Lagrangian tori in \mathbb{CP}^2 and $\mathbb{S}^2 \times \mathbb{S}^2$. J. Symplectic Geom., 11(3):343–361, 2013.
- [Hat02] Allen Hatcher. Algebraic topology. Cambridge University Press, Cambridge, 2002.
- [HL10] Shengda Hu and François Lalonde. A relative Seidel morphism and the Albers map. Trans. Amer. Math. Soc., 362(3):1135–1168, 2010.
- [HLRS16] Vincent Humilière, Frédéric Le Roux, and Sobhan Seyfaddini. Towards a dynamical interpretation of Hamiltonian spectral invariants on surfaces. Geom. Topol., 20(4):2253–2334, 2016.
- [HLS15a] Vincent Humilière, Rémi Leclercq, and Sobhan Seyfaddini. Coisotropic rigidity and C⁰-symplectic geometry. Duke Math. J., 164(4):767–799, 2015.
- [HLS15b] Vincent Humilière, Rémi Leclercq, and Sobhan Seyfaddini. New energycapacity-type inequalities and uniqueness of continuous Hamiltonians. Comment. Math. Helv., 90:1–21, 2015.
- [HLS16] Vincent Humilière, Rémi Leclercq, and Sobhan Seyfaddini. Reduction of symplectic homeomorphisms. Ann. Sci. Éc. Norm. Supér. (4), 49(3):633–668, 2016.
- [HS95] H. Hofer and D. A. Salamon. Floer homology and Novikov rings. In *The Floer memorial volume*, volume 133 of *Progr. Math.*, pages 483–524. Birkhäuser, Basel, 1995.
- [Kha09] Michael Khanevsky. Hofer's metric on the space of diameters. J. Topol. Anal., 1(4):407–416, 2009.
- [KM05] Jelena Katić and Darko Milinković. Piunikhin–Salamon–Schwarz isomorphisms for Lagrangian intersections. Differential Geom. Appl., 22(2):215–227, 2005.
- [KZ] Michael Khanevsky and Frol Zapolsky. In preparation.
- [Lan13] Sergei Lanzat. Quasi-morphisms and symplectic quasi-states for convex symplectic manifolds. Int. Math. Res. Not., (23):5321–5365, 2013.
- [Lec08] Rémi Leclercq. Spectral invariants in Lagrangian Floer theory. J. Mod. Dyn., 2(2):249–286, 2008.
- [MVZ12] Alexandra Monzner, Nicolas Vichery, and Frol Zapolsky. Partial quasimorphisms and quasistates on cotangent bundles, and symplectic homogenization. J. Mod. Dyn., 6(2):205–249, 2012.
- [MZ11] Alexandra Monzner and Frol Zapolsky. A comparison of symplectic homogenization and Calabi quasi-states. J. Topol. Anal., 3(3):243–263, 2011.
- [Oh93] Yong-Geun Oh. Floer cohomology of Lagrangian intersections and pseudoholomorphic disks. I. Comm. Pure Appl. Math., 46(7):949–993, 1993.

[Oh97] Yong-Geun Oh. Symplectic topology as the geometry of action functional. I. Relative Floer theory on the cotangent bundle. J. Differential Geom., 46(3):499-577, 1997. [Oh99] Yong-Geun Oh. Symplectic topology as the geometry of action functional. II. Pants product and cohomological invariants. Comm. Anal. Geom., 7(1):1-54, 1999.[Oh05a] Yong-Geun Oh. Construction of spectral invariants of Hamiltonian paths on closed symplectic manifolds. In The breadth of symplectic and Poisson geometry, volume 232 of Progr. Math., pages 525–570. Birkhäuser Boston, Boston, MA, 2005. [Oh05b] Yong-Geun Oh. Spectral invariants, analysis of the Floer moduli space, and geometry of the Hamiltonian diffeomorphism group. Duke Math. J., 130(2):199-295, 2005.[Ost06] Yaron Ostrover. Calabi quasi-morphisms for some non-monotone symplectic manifolds. Algebr. Geom. Topol., 6:405–434 (electronic), 2006. Joel Oakley and Michael Usher. On certain Lagrangian submanifolds of $S^2 \times$ [OU16] S^2 and \mathbb{CP}^n . Algebr. Geom. Topol., 16(1):149–209, 2016. Leonid Polterovich. The geometry of the group of symplectic diffeomorphisms. [Pol01] Lectures in Mathematics ETH Zürich. Birkhäuser Verlag, Basel, 2001. [Pol12] Leonid Polterovich. Quantum unsharpness and symplectic rigidity. Lett. Math. Phys., 102(3):245–264, 2012. [Pol14] Leonid Polterovich. Symplectic geometry of quantum noise. Comm. Math. Phys., 327(2):481-519, 2014. Leonid Polterovich and Daniel Rosen. Function theory on symplectic mani-[PR14] folds, volume 34 of CRM Monograph Series. American Mathematical Society, Providence, RI, 2014. [PSS96] Sergey Piunikhin, Dietmar Salamon, and Matthias Schwarz. Symplectic Floer-Donaldson theory and quantum cohomology. In Contact and symplectic geometry (Cambridge, 1994), volume 8 of Publ. Newton Inst., pages 171–200. Cambridge Univ. Press, Cambridge, 1996. [San10] Sheila Sandon. An integer-valued bi-invariant metric on the group of contactomorphisms of $\mathbb{R}^{2n} \times S^1$. J. Topol. Anal., 2(3):327–339, 2010. Sheila Sandon. Contact homology, capacity and non-squeezing in $\mathbb{R}^{2n} \times S^1$ [San11] via generating functions. Ann. Inst. Fourier (Grenoble), 61(1):145-185, 2011. Matthias Schwarz. Cohomology operations from S^1 -cobordisms in Floer Ho-[Sch] mology. PhD thesis, Ruhr-Universität Bochum. [Sch93] Matthias Schwarz. Morse homology, volume 111 of Progress in Mathematics. Birkhäuser Verlag, Basel, 1993. [Sch00] Matthias Schwarz. On the action spectrum for closed symplectically aspherical manifolds. Pacific J. Math., 193(2):419-461, 2000. [Sei08] Paul Seidel. Fukaya categories and Picard-Lefschetz theory. Zurich Lectures in Advanced Mathematics. European Mathematical Society (EMS), Zürich, 2008.Sobhan Seyfaddini. C^0 -limits of Hamiltonian paths and the Oh–Schwarz spec-[Sey13a] tral invariants. Int. Math. Res. Not., (21):4920-4960, 2013. [Sey13b] Sobhan Seyfaddini. The displaced disks problem via symplectic topology. C. R. Math. Acad. Sci. Paris, 351(21-22):841-843, 2013. [Sey14] Sobhan Seyfaddini. Unboundedness of the Lagrangian Hofer distance in the Euclidean ball. Electron. Res. Announc. Math. Sci., 21:1-7, 2014. [Thé99] David Théret. A Lagrangian camel. Comment. Math. Helv., 74(4):591-614, 1999. [Ush11] Michael Usher. Deformed Hamiltonian Floer theory, capacity estimates and Calabi quasimorphisms. Geom. Topol., 15(3):1313-1417, 2011.
- [Vit92] Claude Viterbo. Symplectic topology as the geometry of generating functions. Math. Ann., 292(4):685–710, 1992.
- [Wel08] Jean-Yves Welschinger. Open strings, Lagrangian conductors and Floer functor. 2008. arXiv:0812.0276.
- [Wu12] Weiwei Wu. On an exotic Lagrangian torus in $\mathbb{C}P^2$. 2012. arXiv:1201.2446.
- [WW15] Katrin Wehrheim and Chris Woodward. Orientations for pseudoholomorphic quilts. 2015. arXiv:1503.07803.
- [Zap13a] Frol Zapolsky. Geometry of contactomorphism groups, contact rigidity, and contact dynamics in jet spaces. Int. Math. Res. Notices, 2013(20):4687–4711, 2013. doi:10.1093/imrn/rns177.
- [Zap13b] Frol Zapolsky. On the Hofer geometry for weakly exact Lagrangian submanifolds. J. Symplectic Geom., 11(3):475–488, 2013.
- [Zap15] Frol Zapolsky. The Lagrangian Floer-quantum-PSS package and canonical orientations in Floer theory. 2015. arXiv:1507:02253.
- [Zin13] Aleksey Zinger. The determinant line bundle for Fredholm operators: Construction, properties, and classification. 2013. arXiv:1304.6368.

RL: Université Paris-Sud, Département de Mathématiques, Bat. 425, 91400 Orsay, France

E-mail address: remi.leclercq@math.u-psud.fr

FZ: DEPARTMENT OF MATHEMATICS, UNIVERSITY OF HAIFA, MOUNT CARMEL, HAIFA 31905, ISRAEL

E-mail address: frol.zapolsky@gmail.com